Ambulatory Care Nurse-Sensitive Indicator Industry Report

Meaningful Measurement of Nursing in the Ambulatory Patient Care Environment

May 2016



Ambulatory Care Nurse-Sensitive Indicator Industry Report: Meaningful Measurement of Nursing in the Ambulatory Patient Care Environment

Editors

Rachel Start, MSN, RN, NE-BC Ann Marie Matlock, DNP, RN, NE-BC Peg Mastal, PhD, MSN, RN

Copyright © 2016

American Academy of Ambulatory Care Nursing (AAACN)
East Holly Avenue/Box 56, Pitman, New Jersey 08071-0056
800-AMB-NURS; FAX 856-589-7463; aaacn@ajj.com; www.aaacn.org

Publication Management by

Anthony J. Jannetti, Inc., East Holly Avenue/Box 56, Pitman, New Jersey 08071-0056

Phone: 856-256-2300; Fax: 856-589-7463

Website: www.ajj.com

All rights reserved. No part of this publication may be reproduced or transmitted in any form or by any means, electronic or mechanical, including photocopy, recording, or any information storage and retrieval system without the written permission of the American Academy of Ambulatory Care Nursing.

DISCLAIMER

The authors, editors, and publishers of this report have made serious efforts to ensure that practices, procedures, and treatments are accurate and conform to standards accepted at the time of publication. Due to constant changes in information resulting from continuous research and clinical experience, reasonable differences in opinions among authorities, unique aspects of individual clinical situations, and the possibility of human error in preparing such a publication, the reader should exercise individual judgment when making a clinical decision, and if necessary, consult and compare information from other authorities, professionals, or sources.

Suggested Citation

Start, R., Matlock, A.M., & Mastal, P. (2016). *Ambulatory care nurse-sensitive indicator industry report: Meaningful measurement of nursing in the ambulatory patient care environment.* Pitman, NJ: American Academy of Ambulatory Care Nursing.

What nursing brings to the future of health care is a steadfast commitment to patient care, improved safety and quality, and better outcomes. Most of the near-term challenges identified in the health care reform legislation speak to traditional and current strengths of the nursing profession in such areas as care coordination, health promotion, and quality improvement. How well nurses are trained and do their jobs is inextricably tied to most health care quality measures that have been targeted for improvement over the past few years. Thus for nursing, health care reform provides an opportunity for the profession to meet the demand for safe, high-quality, patient-centered, and equitable health care services. We believe nurses have key roles to play as team members and leaders for a reformed and better-integrated, patient-centered healthcare system (Institute of Medicine, 2010, pp. xi-xii).

TABLE OF CONTENTS

Forev	word		7	
Nurse	e-Sens	Academy of Ambulatory Care Nursing (AAACN) sitive Indicator Task Force Members and eport Contributors	9	
Amer	American Academy of Ambulatory Care Nursing			
Executive Summary			12	
Introduction and Background Information				
	view o	of Information Affecting Quality Measurement ory Care Nursing	18	
	A. B. C. D. E. F. G.	Definition of Ambulatory Care Nursing Defining Characteristics of Ambulatory Care Nursing History of Evaluating Nursing Quality Building a Culture of Safety Definition of Nurse-Sensitive Measure Quality Indicator Development and Pilot Testing Processes American Nurses Association Proposed Measure Topic Areas from ANA Ambulatory Summit AAACN Nurse-Sensitive Indicator Task Force Journey	19 19 21 22 22 23 24	
	w of E	Existent Endorsed Ambulatory Care Nurse-Sensitive leasures	26	
2.1	(NDN	s Ganey National Database of Nursing Quality Indicators IQI)/Press Ganey, Inc. Care Coordination: Medication Reconciliation Care Coordination: Pending Diagnostic Test Results	27 27 29	
2.2	Colla A. B. C. D.	Ambulatory Site Demographics Ambulatory Volume Measures (Denominators) Ambulatory Staffing, Skill Mix, and Patient Care Hours Ambulatory Surgery Adverse Outcomes of Care: Wrong Site, Wrong Side, Wrong Patient, Wrong Procedure, Wrong Implant Ambulatory Surgery Patient Burns Ambulatory Surgery Patient Falls	30 30 32 33 34 35 36	

G.	Ambulatory Surgery Patient Injury Falls	37
H.	Ambulatory Surgery Hospital Transfer/Admission	38
Developmen	Task Force Proposed Indicators for nt and Endorsement: Existent Health Care Measures d Recommended as Ambulatory Care Nurse-Sensitive easures	39
A.	Ambulatory Care Nurse Readmission Across the Lifespan	40
В.	Ambulatory Care Nurse Pain Assessment and Follow Up	45
C.	Ambulatory Care Nurse Screening for High Blood Pressure	48
	and Follow-up Care	
D.	Ambulatory Care Nurse Screening and Follow-Up	52
=	Documentation for Depression	
E.	Ambulatory Care Nurse Patient Falls in the Institution	57
F.	Ambulatory Care Nurse Screening for Future Falls Risk	59
G.	Ambulatory Care Nurse Screening for Body Mass Index (BMI)	61
H.	Ambulatory Care Nurse RN Demographics	63
	1. Diploma RN	63
	2. Associate's Degree in Nursing RN	63
	3. Bachelor's of Science in Nursing RN	64
	4. Master's of Science in Nursing RN	64
	5. Doctorate in Nursing RN	64
	6. Certification RN	64
	7. Certification RN in Field of Specialty	65
	8. Turnover by Full-Time Equivalent	65
_	9. Vacancy by Full-Time Equivalent	65
l.	Ambulatory Care Nurse Patient Satisfaction	66
Section 4		68
	Task Force Proposed Indicators for	
	nt and Endorsement: Original Measures Recommended as	
Ambulatory	Care Nurse-Sensitive Indicator Measures	
A.	Ambulatory Care Nurse Care Coordination: Appropriate Referral	69
В.	Ambulatory Care Nurse Patient Engagement:	73
	Measurement of Patient Activation	
C.	Ambulatory Care Nurse Administration of Vaccine	81
	Per Non-Patient Specific Protocol	
D.	Ambulatory Care Nurse Clean Urine Specimen	83

	nt Exe	mplars in Ambulatory Care Nurse-Sensitive easurement	85
	А. В.	Care Coordination: RN Care Manager Reduction of Readmissions Care Coordination: Population Health Management Through Advanced Care Planning of High Risk Populations	86 86
	C. D.	Care Coordination: Decreasing Pediatric Emergency Room Visits Wound Care Center Healing Rate Through RN Case Managed Cohorts of Patients	87 87
	E. F. G.	Opioid Use Monitoring and Safety at Pain Center Reduction of Readmissions Via a Follow-Up Phone Call Process Team BMI: Management of Pediatric Overweight and Obesity in Pediatric Primary Care	88 89 89
Section 6 Glossary of Definitions and Supportive Statements			90
	A. B. C. D. E.	AAACN Definition of the Role of the Nurse in Ambulatory Dimensions of the Ambulatory Care Nurse Role Definition of Care Coordination Definition of Transition Management Definition of the Role Dimensions of Nurse Care Coordination and Transition Management Nursing Alliance for Quality Care Definition of Patient Engagement	91 92 93 93 94
Section 7 References and Additional Readings			97
	A. B. C.	References Additional Readings Acronym List	98 109 115

FOREWORD

The current health care quality improvement infrastructure is the result of a century-long experience of cumulative efforts. It began with an acknowledgement of the role of quality in health care, and gradually evolved to encompass the prioritization of quality improvement and the development of organizational and technological systems to monitor, quantify, and incentivize quality improvement in health care.

Still evolving today, quality focuses on performance improvement and outcome measurement that documents improved health care processes and patient outcomes. Major national and organizational efforts have focused on quantifying, standardizing, routinely measuring, and reporting health care processes and outcomes (Marjoua & Bozic, 2012). Each of the health professions must endeavor to establish the important elements that will improve the way we deliver health care in the future.

It became evident to professional nurses in outpatient settings that identifying, defining, and measuring the elements of their practice related to patient care and outcomes was critical. A measureable value of the ambulatory care registered nurse (RN) role in improving health has not been documented to date. Many RNs in the ambulatory care setting have not been professionally socialized with an organizational emphasis on structural empowerment, autonomy, leadership, and practice management. They are, in many settings, unprepared to take up this mantle of health care delivery leadership. Furthermore, they have been under-supported as a professional group because they have yet to quantify their value with metrics that differentiate them as a specialty. However, as the outpatient setting increasingly connects patients and families to care coordination, transition management, health promotion, and community resources, the role of the RN as a leader in this setting is increasingly vital.

The American Academy of Ambulatory Care Nursing (AAACN) energized the quality activities of nurses in ambulatory care settings. In spring 2013, the AAACN Board of Directors commissioned a task force to identify and define nurse-sensitive indicators (NSIs) specific to the ambulatory setting.

Members of the task force represent all geographic regions of the United States and a broad range of practice settings. Members include development experts from the Collaborative Alliance for Nursing Outcomes (CALNOC) and the Press Ganey National Database for Nursing Quality Indicators (NDNQI)/Press Ganey Inc., as well as a past president of AAACN who was part of the original work done in the late 1990's via the American Nurses Association (ANA) to determine indicators sensitive to non-acute settings.

As experts in the specialty of ambulatory care nursing, we advocate for the measurement and benchmarking of RN-sensitive indicators that reflect the full contribution and value of the RN to the health care system and include the entire lifespan of our patients and families. We provide our expertise in the development of

indicators that best reflect the impact of the RN in this setting. We comprehend the many facets of health care dysfunction that continue to persist due to immature performance improvement infrastructures in the ambulatory environment. We strive to be the premier voice and endorser of any entity that identifies, develops, and benchmarks nurse-sensitive indicators in the ambulatory arena.

As this work continues to evolve, AAACN strives to continue to support the tenants of the Institute of Medicine's report *The Future of Nursing: Leading Change, Advancing Health* (IOM, 2010) and its follow-up report (IOM, 2015) to ensure that in all settings the transformation of health care is led by nurses. As recommended in the *Future of Nursing* follow-up report, nurses are encouraged to lead reforms to the health care system and improve data infrastructure through the collaboration of organizations and associations (IOM, 2015). In August 2015 as this report was being completed, AAACN and CALNOC entered into a collaborative partnership to explore the development and pilot testing of ambulatory care nurse-sensitive indicators. For more than 20 years, CALNOC has developed nurse-sensitive indicators and is the leading provider of actionable information and research on nurse-sensitive indicators. Its mission is to advance global patient care safety, outcomes, and performance measurement efforts. Through this collaborative partnership a steering committee was formed that is taking the proposed indicators in this report and further developing them for eventual piloting and benchmarking.

The work of this task force is critical as the transformation of health care is moving at a rapid pace and is increasingly focused on the outpatient realm. As inpatient admissions and overall length of stay are decreasing, ambulatory visits are increasing. As the focus of health care is transitioning from inpatient episodes of care to community and outpatient settings, ambulatory care nurses have been called upon to lead in multiple areas and practice to the full extent of their education and training. Their ability to utilize, develop, and measure role-specific data that is meaningful and participate as partners in every step of the transition of this paradigm is crucial. This work is an ongoing journey best led under the auspices of AAACN, the professional organization that represents and understands the value and role of the ambulatory care registered nurse.

Nancy May, MSN, RN-BC, NEA-BC President, 2015-2016 American Academy of Ambulatory Care Nursing

AMERICAN ACADEMY OF AMBULATORY CARE NURSING (AAACN) NURSE-SENSITIVE INDICATOR TASK FORCE MEMBERS AND INDUSTRY REPORT CONTRIBUTORS

AAACN Nurse-Sensitive Indicator Task Force Measure Development Leaders

Rachel Start, MSN, RN, NE-BC

Co-Chair and Industry Report Editor
Director, Ambulatory Nursing, Nursing Practice
and Magnet Performance
Rush Oak Park Hospital
Adjunct Faculty
Rush University College of Nursing

Ann Marie Matlock, DNP, RN, NE-BC

Co-Chair and Industry Report Editor
Service Chief, Medical Surgical Specialties
National Institutes of Health – Clinical Center
CAPT United States Public Health Service

Peg Mastal, PhD, MSN, RN

Industry Report Editor
Past President, AAACN

M. Elizabeth Greenberg, PhD, RN-BC, C-TNP

AAACN Board Liaison to the Nurse Sensitive-Indicator Task Force Assistant Clinical Professor Northern Arizona University

Rosemarie Battaglia, MSN, RN

Measure Development Subgroup Leader: Readmission Across the Lifespan Nurse Manager Children's Hospital at the Medical University of South Carolina

Diane Storer Brown, PhD, RN, CPHQ, FNAHQ, FAAN

Measure Development Subgroup Leader: Staffing, Demographics and Strategy Senior Scientist Collaborative Alliance for Nursing Outcomes (CALNOC) Executive Director, Coordination of Care Kaiser Permanente Northern California

Eileen Esposito, DNP, RN-BC, CPHQ

Measure Development Subgroup Leader: Patient Engagement Vice President, Ambulatory Clinical Practice Catholic Health Services of Long Island

Kathleen Martinez, BSN, RN, CPN

Measure Development Subgroup Leader: Appropriate Referral Clinical Policy Oversight Manager Children's Hospital Colorado

Mary Morin, RN, NEA-BC, RN-BC

Measure Development Subgroup Leader: Existent Endorsed Measures Adapted and Proposed as Ambulatory Care Nurse-Sensitive Indicator Measures Vice President and Nurse Executive Sentara Medical Group

Cynthia Murray, BN, RN-BC

Leader, Literature Review Ambulatory Care VA Health System

Catherine Rhodes, MSN, APRN, WHNP-BC, RNC-OB, SANE-A

Measure Development Subgroup Leader: Existent Endorsed Measures Adapted and Proposed as Ambulatory Care Nurse-Sensitive Indicator Measures Advanced Practice Registered Nurse The MetroHealth System

Leilani Siaki, LTC, PhD, FNP-BC

Measure Development Subgroup Leader: Existent Endorsed Measures Adapted and Proposed as Ambulatory Care Nurse-Sensitive Indicator Measures Deputy Chief Center for Nursing Science and Clinical Inquiry, Madigan Army Medical Center

AAACN Nurse-Sensitive Indicator Industry Report Contributors

Harriet Aronow, PhD

Director of Data Management Services Collaborative Alliance for Nursing Outcomes Research Scientist in Nursing Cedars-Sinai Medical Center

Cathy Besthoff, MHA, RN, CPHQ

MCIC Vermont, Risk Retention Group and Nursing Alliance Quality Care Member

Sharon Eck Birmingham, DNSc, MA, BSN, RN

Previous AAACN NSI Task Force Chair Principle Investigator, National Nurse Scheduling Study Adjunct Faculty, University of Iowa and Yale University Schools of Nursing

Nena Bonuel, PhD, RN, CCRN, CNS, ACNS-BC

Director, Nursing Education/Education Center for Patient Education Tenet Health – Park Plaza Hospital

Stefanie Coffey, DNP, MBA, FNP-BC, RN-BC, VHA-CM

Associate Chief Nurse Executive/Nurse Practitioner, Ambulatory Care Nursing Roseburg VA Healthcare System

Catherine Davis, BSN, MPS, RN, CRRN

Telehealth Program Coordinator Craig Hospital

Nancy Dunton, PhD, RN, FAAN

Research Professor and Director National Database of Nursing Quality Indicators (NDNQI) and University of Kansas Medical Center School of Nursing

Kristene Grayem, MSN, RN, CNP

Director of Clinical Systems and Performance Improvement Akron Children's Hospital

Ann Jacobson, PhD, CNS, ACNS-BC

Professor and Associate Dean Kent State University College of Nursing

Dana Nelson, MBA, MSN, RN-BC, CRRN

Director, Patient Safety and Accreditation/Outpatient Nursing Practice

Karen Seifert, MSN, RN, ACNS-BC, CDE

Clinical Nurse Specialist University of California Davis Medical Center

Beverly Sullivan, BSN, RN

Internal Medicine ACC University of California Davis Medical Center

AAACN Board of Directors

Debra Cox, MS, RN Rocquel Crawley, DHA, MBA, BSN, RNC-OB, NEA-BC Kristene Grayem, RN, MSN, CNP M. Elizabeth Greenberg, PhD, RN-BC, C-TNP Cynthia Nowicki Hnatiuk, EdD, RN, CAE, FAAN Nancy May, MSN, RN-BC, NEA-BC Kathy Mertens, MN, MPH, RN Wanda C. Richards, MPA, MSM, BSN, CAPT/NC/USN Marianne Sherman, RN-BC, MS

Project Staff

Linda Alexander, Managing Editor, AAACN Stephanie McDonald, Customer Service Coordinator, AAACN Patricia Reichart, Director of Association Services, AAACN

AMERICAN ACADEMY OF AMBULATORY CARE NURSING

The American Academy of Ambulatory Care Nursing (AAACN) is a welcoming, unifying community for registered nurses in all ambulatory care settings. The association offers:

- Connections with others in similar roles
- Help in advancing practice and leadership skills
- Advocacy that promotes greater appreciation for the specialty of ambulatory care nursing.

AAACN evolved from the American Academy of Ambulatory Nursing Administration, founded in 1978 as a not-for-profit educational forum. In 1993, the organization's name was changed to the American Academy of Ambulatory Care Nursing (AAACN). Membership was broadened to include nurses in direct practice, education, and research roles as well as those in management and administration. Today, membership is open to nurses and other professionals interested in ambulatory care, care coordination and transition management, and telehealth nursing. Corporations and individual corporate representatives are also welcomed as members. AAACN is the only specialty nursing association that focuses on excellence in ambulatory care.

AAACN serves as a voice for ambulatory care nurses across the continuum of health care delivery.

EXECUTIVE SUMMARY

The purpose of this report is to identify, define, and propose nurse-sensitive measures that reflect the work and value of registered nurses (RNs) in ambulatory care settings. The measures proposed in this report reflect the activities and outcomes of the AAACN Nurse-Sensitive Indicator (NSI) Task Force from spring 2013 through fall 2015. Section 2 details measures that are currently endorsed as ambulatory care nurse-sensitive indicators, while sections 3 and 4 describe measures proposed by the NSI Task Force for development. The report incorporates a review of current literature, health care quality measures, and best practices in ambulatory nursing care, as well as the opinions, expertise, and experience of AAACN members.

Section 1 provides a brief history of nurse quality evaluation and the development of nurse-sensitive indicators. This section reviews the role of the ambulatory care nurse and the urgent need for ambulatory care nursing leaders to demonstrate the value of the role of the RN in ambulatory care. Additional indicator topic areas for development discussed at the 2014 American Nurses Association Ambulatory Measurement Summit are also included.

Section 2 reviews existing and endorsed ambulatory care nurse-sensitive indicators from the Collaborative Alliance for Nursing Outcomes (CALNOC) and the Press Ganey National Database for Nursing Quality Indicators (NDNQI)/Press Ganey, Inc. These indicators are ready to be used by any organization that wishes to benchmark against other organizations within each respective database.

Section 3 identifies and describes nine indicators recommended for development, endorsement, and pilot testing. These measures are adapted from already existent, non-nursing specific, health care endorsed measures. These measures are well understood by the health care community at large, and with AAACN NSI Task Force recommendations for adaptation to the ambulatory care nurse role, have good potential for meaningful measurement of the RN in the ambulatory care setting.

Section 4 identifies and describes four new, original measures recommended for development, endorsement, and pilot testing. These four measures have the potential to uniquely reflect the role of the RN in the ambulatory care setting as well as promote enhanced nursing practice. Each measure represents a cumulative effort of evidence and best practice review.

Section 5 describes several current exemplars in ambulatory care nurse-sensitive indicator measurement at the microsystem level. These exemplars reflect the efforts of many organizations and highlight work being done to collect metrics that assist in quantifying and continuously improving the work of the ambulatory care RN. These exemplars and the lessons learned have informed the work presented in this report.

This report reflects the beginning of a journey to develop and endorse measures that meaningfully reflect the value of the RN in ambulatory care. The work should continue to be seen for what it is: highly complex, arduous, and constantly evolving. Members of this AAACN NSI Task Force are committed to a long journey of development, and we will continue to provide input and expertise to help shape measures that will best assign value to the role of the RN in ambulatory care.

INTRODUCTION AND BACKGROUND INFORMATION

Over the past 20 years, concerns have grown about health care delivery processes and the less-than-optimal outcomes many patients experience. There has been considerable work conducted to identify these issues and examine how to improve health care.

A recent report by the Commonwealth Fund stated, "Despite having the most expensive health care system, the United States ranks last overall among 11 industrialized countries on measures of health system quality, efficiency, access to care, equity, and healthy lives" (Davis, Stemkis, Squires, & Schoen, 2014, p. 6). The National Quality Strategy and the Patient Protection and Affordable Care Act (PPACA) aim to fix these dysfunctions by supporting innovative processes that provide patients with better care, develop structures that support health in both people and communities, and create systems that are more affordable and accessible to all (Berwick, Nolan, & Whittington, 2008; U.S. Department of Health and Human Services, 2011; Institute of Medicine [IOM], 2001).

In its landmark report, *The Future of Nursing: Leading Change, Advancing Health*, the IOM states:

The United States has the opportunity to transform its health care system to provide seamless, affordable, quality care that is accessible to all, patient centered, and evidence based and leads to improved health outcomes. Achieving this transformation will require remodeling many aspects of the health care system (2010, p. 1).

Impetus for Change: Paradigm Shift, Care Delivery, and Performance Improvement

Over the past two decades the delivery of health care has shifted from the inpatient to the ambulatory care setting. According to the American Hospital Association (AHA, 2015), inpatient admissions have decreased from 11.93% in 1993 to 10.63% in 2013, and the average length of stay has reduced from 7.0 days in 1993 to 5.4 days in 2013. Meanwhile, outpatient visits averaged 14.22% in 1993 and increased to 21.45% in 2013 (AHA, 2015), while the registered nurse (RN) workforce in the outpatient setting, particularly in primary care, has declined (IOM, 2010).

The National Association of Community Health Centers (NACHC) estimates 56 million people have inadequate access to a primary care medical provider and that with the rise in outpatient visits, health centers need to increase the number of patients served from 16 million to 30 million (IOM, 2010). With the dramatic increase of Americans with health insurance, there is an expectation that there will be a need for more RNs due to this policy change, requiring 16,000 to 20,000 more RNs in the workforce (IOM, 2010). Currently the percent of RNs employed in ambulatory care remained unchanged between 2004 and 2008 at 12% (IOM, 2010).

The PPACA of 2010 supports rebuilding the primary care workforce, increasing reimbursement for primary care, strengthening community health centers, and providing free preventive care for seniors. One such program is Community Care Transitions, established to help high-risk older adults who are hospitalized avoid unnecessary readmissions through care coordination and connection of patients to services in their communities (Patient Protection, 2010). Evidence supports RNs as critical to delivery of these preventive services, as well as care coordination activities to avoid rehospitalizations. The Accountable Care Organizations (ACOs) supported by the PPACA are centered on delivering coordinated, efficient, and effective care (Swan & Haas, 2011).

In the late 1990s, the American Nurses Association (ANA) Congress of Nursing Practice convened an advisory committee to "identify indicators sensitive to the impact of nursing practice in community-based non-acute settings" (ANA, 2000, p. viii). These settings were identified as long-term care, home health, school health, and ambulatory care (Swan, 2008). The framework used to organize these indicators were built on the following categories of care: utilization of services, patient satisfaction, risk reduction, increased protective factors, level of functioning, psychosocial functioning, changes in symptom severity, and strength of therapeutic alliance (Montalvo, 2007; Sawyer et al., 2002; Swan, 2008).

In its requirements for the Magnet Recognition Program, the American Nurses Credentialing Center (ANCC) states, "nurses at all levels analyze data and use national benchmarks to gain a comparative perspective about their performance and the care patients receive...Action plans are developed that lead to systematic improvements over time...Magnet organization data demonstrate outcome measures that generally outperform the benchmark statistic of the national database used in patient and nurse sensitive indicators" (ANCC, 2014, p.42). The key data elements required by application to this program are nurse-related patient satisfaction, nursing sensitive quality, nurse satisfaction, nurse staffing, nurse turnover and vacancy, among others, for every RN in every setting of the applicant organization (ANCC, 2014).

The Nursing Mandate and Importance of the Ambulatory Care RN Role in Health Care Transformation

The IOM stated:

By virtue of its numbers and adaptive capacity, the nursing profession has the potential to effect wide-reaching changes in the health care system. Nurses' regular, close proximity to patient and scientific understanding of care processes across the continuum of care give them a unique ability to act as partners with other health professionals and to lead in the improvement and redesign of the healthcare system and its many practice environments, including hospitals, schools, homes, retail health clinics, long term care facilities, battlefield and community and public health centers.

Nurses thus are poised to help bridge the gap between coverage and access, to coordinate increasingly complex care for a wide range of patients, to fulfill their potential as primary care providers to the full extent of their education and training and to enable the full economic value of their contributions across practice settings to be realized. In addition a promising field of evidence links nursing care to high quality of care for patient, including protecting their safety. Nurses are crucial in preventing medication errors, reducing rates of infection and even facilitating patients' transition from hospital to home (IOM, 2010, pp. 2-3).

These statements and the associated recommendations were not specific to any one environment in which nurses practice, rather they were all encompassing to every sector, including ambulatory care.

In order to bridge the gaps that exist in our health care system, the IOM recommends RNs in every setting:

- 1. Practice to the full extent of their education and training,
- 2. Achieve higher levels of education and training,
- 3. Be full partners with physicians and other health professionals in redesigning healthcare in the United States.
- 4. Seek more effective workforce planning and policy making that requires better data collection and improved information infrastructures (IOM, 2010).

In its evaluation of the progress made by its original *Future of Nursing* report, the IOM stated:

Baccalaureate prepared nurses are not fully utilized across all practice settings; in particular in ambulatory care settings where they are needed to provide population health, health promotion, disease prevention, and chronic disease management ... In some areas local credentialing requirements and practice policies can limit nursing practice even though state regulations permit a broader legal scope (Breslin, 2015. p. 2).

In its position statement on the role of the registered nurse in ambulatory care, AAACN states:

The transition of health care from the inpatient to ambulatory setting has created challenges related to accessing care and coordinating services, and escalated the complexity of care delivered outside the hospital. This shift has dramatically increased the need for professional nursing services, as patients and their families require increased depth and breadth of care. Ambulatory RNs facilitate patient care services by managing and individualizing care for patients and their families, who increasingly require assistance navigating the complex health care system. With provision of complex procedural care, ambulatory care RNs provide

support with decision making, patient education, and coordination of services (AAACN, 2010, p. 1).

Pay for performance initiatives are changing the quality environment (Swan, 2008). Nurse-sensitive indicators in acute care have been established and widely accepted nationally (Swan, 2008). Leaders in ambulatory care nursing must seek to describe the value of nurses' contributions, as has been accomplished through the development of indicators of quality patient care and nurse-sensitive outcomes in other environments where nurse care is delivered (Swan, 2008). According to Swan and Haas (2011), ambulatory care RNs are well positioned to fully participate in health care reform initiatives.

Ambulatory care RNs must be engaged in documenting and measuring the impact of nursing in care coordination and transition management related to patient outcomes as well as to cost effectiveness and improvements in the patient's and family's well-being (Swan & Haas, 2011). Many sectors are already working on interventions that improve the gaps in the health care system. Hamner (2005) summarized 16 studies that showed RN-led interventions in outpatient heart failure clinics (10 studies) and telephone- or technology-based interventions (six studies) that showed reduced hospital readmissions, reduced emergency room visits, decreased mortality, and improved self-care and quality of life, as well as improved patient satisfaction (Hamner, 2005; Swan, Conway-Phillips & Griffin, 2006).

With so many instances of dysfunction in the current health care system, a mandate exists for nurses, representative of all settings, to lead problem solving and transformational change. As preventive and population health continue to grow in their importance toward this transformational change, nurses in the community, outpatient setting, and ambulatory care arenas must assume these leadership roles to transform health throughout the continuum for patients and families.

SECTION 1 Overview of Information Affecting Quality Measurement in Ambulatory Care Nursing

- A. Definition of Ambulatory Care Nursing
- B. Defining Characteristics of Ambulatory Care Nursing
- C. History of Evaluating Nursing Quality
- D. Building a Culture of Safety
- E. Definition of Nurse-Sensitive Measure
- F. Quality Indicator Development and Pilot Testing Processes
- G. American Nurses Association Proposed Measure Topic Areas from ANA Ambulatory Summit
- H. AAACN Nurse-Sensitive Indicator Task Force Journey

SECTION 1 OVERVIEW OF INFORMATION AFFECTING QUALITY MEASUREMENT IN AMBULATORY CARE NURSING

AAACN Nurse-Sensitive Indicator (NSI) Task Force members reviewed past and present information about ambulatory care nursing and quality efforts. This review includes a definition of ambulatory care nursing, defining characteristics of the professional registered nurse (RN), a short overview of the history of nursing quality, and the task force's work to develop a set of nursing metrics.

A. Definition of Ambulatory Care Nursing

According to the American Academy of Ambulatory Care Nursing (AAACN, 2011):

Professional ambulatory care nursing is a complex, multi-faceted specialty that encompasses independent and collaborative practice. The comprehensive practice of ambulatory care nursing is built on a broad knowledge base of nursing and health sciences, and applies clinical expertise rooted in the nursing process. Nurses use evidence-based information across a variety of outpatient health care settings to achieve and ensure patient safety and quality of care while improving patient outcomes.

Ambulatory care includes those clinical, organizational, and professional activities engaged in by registered nurses with and for individuals, groups, and populations who seek assistance with improving health and/or seek care for health-related problems. Registered nurses promote optimal wellness, participate in the management of acute illness, assist the patient to manage the effects of chronic disease and disability, and provide support in end-of-life care. The ambulatory care registered nurse is accountable for the provision of nursing care in accordance with relevant federal requirements, state laws and nurse practice acts, regulatory standards, the standards of professional ambulatory care nursing practice, other relevant professional standards, and organizational policies (p. 3).

B. Defining Characteristics of Ambulatory Care Nursing

- Ambulatory nursing care requires critical reasoning and astute clinical judgment in order to expedite appropriate care and treatment, especially given that the patient may present with complex problems or potentially life threatening conditions.
- 2. Ambulatory care registered nurses provide care across the life span to individuals, families, caregivers, groups, populations, and communities.
- Ambulatory care nursing occurs across the continuum of care in a variety of settings, which include but are not limited to hospital-based clinic/centers, solo or group medical practices, ambulatory surgery & diagnostic procedure centers, telehealth service environments, university and community hospital clinics, military and veterans administration settings, nurse-managed clinics,

- managed care organizations, colleges and educational institutions, free-standing community facilities, care coordination organizations, and patient homes.
- 4. Ambulatory care registered nurses interact with patients during face-to-face encounters or through a variety of telecommunication strategies, often establishing long-term relationships.
- 5. Telehealth nursing is an integral component of professional ambulatory care nursing that utilizes a variety of telecommunications' technologies during encounters to assess, triage, provide nursing consultation, and perform follow up and surveillance of patients' status and outcomes.
- 6. During each encounter, the ambulatory care registered nurse focuses on patient safety and the quality of nursing care by applying appropriate nursing interventions, such as identifying and clarifying patient needs, performing procedures, conducting health education, promoting patient advocacy, coordinating nursing and other health services, assisting the patient to navigate the health care system, and evaluating patient outcomes.
- 7. Nurse-patient encounters can occur once or as a series of occurrences, are usually less than 24 hours in length at any one time, and occur singly or in group settings.
- 8. Ambulatory care registered nurses, acting as partners and advisors, assist and support patients and families to optimally manage their health care, respecting their culture and values, individual needs, health goals and treatment preferences.
- Ambulatory care registered nurses facilitate improved care delivery and outcomes using the precepts of care coordination and transition management (CCTM) when applying the nursing process, participating in multidisciplinary collaboration and utilizing appropriate health care services and resources across the care continuum.
- 10. Ambulatory care registered nurses are knowledgeable about and provide leadership in the clinical and managerial operations of the organization.
- 11. Ambulatory care registered nurses design, administer, and evaluate nursing services within the organization in accord with relevant federal requirements, state laws and nurse practice acts, regulatory standards, and institutional policies and procedures.
- 12. Ambulatory care registered nurses provide operational accountability for and coordination of nursing services, including the appropriate skill mix and delegation of roles and responsibilities for licensed and unlicensed nursing personnel.
- 13. Ambulatory care registered nurses apply the provisions of the American Nurses Association Code of Ethics for Nurses to their own professional obligations and for the patients entrusted to their care.
- 14. Ambulatory care registered nurses pursue lifelong learning that updates and expands their clinical, organizational, and professional roles and responsibilities (AAACN, 2011).

C. History of Evaluating Nursing Quality

Evaluating the quality of nursing practice started when Florence Nightingale identified nursing's role in health care quality and began to measure patient outcomes (Montalvo, 2007). Nightingale used statistical methods to follow trends correlating patient outcomes to environmental conditions (Dossey, Selanders, Beck, & Attewell, 2005; Nightingale, 1859).

Evolution of Modern Quality Measurement

Quality measurement in health care evolved in the latter part of the 20th century. The American Nurses Association (ANA) began work on nursing practice quality measurement in the 1970s. There was wide dissemination of the Quality Assurance (QA) model (Rantz, 1995), and Donabedian (1988) introduced the structure, process, and outcome model that has offered a comprehensive method for evaluating health care quality (Montalvo, 2007). In the early 1990s a need existed nationally to evaluate nurse staffing and identify linkages between nurse staffing and patient outcomes (Montalvo, 2007).

Evolution in the 21st Century

More recently, organizations such as the Collaborative Alliance for Nursing Outcomes (CALNOC) and the Press Ganey National Database of Nursing Quality Indicators (NDNQI)/Press Ganey Inc. conducted a series of pilot studies across the United States. These metrics were endorsed by the National Quality Forum (2004) in the initial set of nurse-sensitive measures that were reflective of the inpatient setting (Brown, Donaldson, Burnes Bolton, & Aydin, 2010; Brown & Wolosin, 2013; Montalvo, 2007). These indicators include the following:

- 1. Nursing Hours per Patient Day
- 2. Patient Falls
- 3. Patient Falls with Injury
- 4. Pediatric Pain Assessment, Intervention, Reassessment (AIR) Cycle
- 5. Pediatric Peripheral Intravenous Infiltration Rate
- 6. Pressure Ulcer Prevalence
- 7. Psychiatric Physical/Sexual Assault Rate
- 8. Restraint Prevalence
- RN Education/Certification
- 10. RN Satisfaction Survey Options
- 11. Staff Skill Mix
- 12. Voluntary Nurse Turnover
- 13. Nurse Vacancy Rate
- 14. Nosocomial Infections
 - Urinary Catheter-Associated Urinary Tract Infection
 - Central Line Catheter-Associated Blood Stream Infection
 - Ventilator-Associated Pneumonia (Montalvo, 2007)

D. Building a Culture of Safety

The landmark IOM report regarding the study of medical errors in hospitals further supported the need to assess and constantly improve metrics related to patient safety (Brown & Wolosin, 2013; IOM, 2001). This report, along with a subsequent IOM call to action (IOM, 2001) and the PPACA of 2010, in concert with the Centers for Medicaid & Medicare Services, have made it essential that organizations create cultures of safety (Brown & Wolosin, 2013).

Safety and quality are intrinsically linked to such factors as skill mix, turnover, and workload intensity (Aiken, Clarke, Sloane, Sochalski, & Silber, 2002; Brown & Wolosin, 2013). As such, the work of developing nurse-sensitive indicators in the ambulatory setting requires simultaneous measurement of RN staffing and demographics to effectively link to quality that is nursing sensitive. Additionally, the benchmarking of both quality and RN demographics across like size and type environments must be done to continue to advance and improve the quality of care RNs provide in these settings (Brown et al., 2010).

E. Definition of Nurse-Sensitive Measure

A Nurse-Sensitive Measure is an indicator that is sensitive to the input of nursing care, reflecting structure, process, and outcomes (NQF, 2004). ANA and CALNOC define "Nurse-Sensitive Indicators" as "those indictors that capture care or its outcomes most affected by nursing care" (ANA, 1996; CALNOC, 2015). NQF further defines a nursing-sensitive performance measure as processes and outcomes, and structural proxies such as skill mix and nurse staffing hours for these processes and outcomes, that are affected, provided, and/or influenced by nursing personnel but for which nursing is not exclusively responsible (NQF, 2004).

Nurse-sensitive indicators are often subdivided into three subcategories: structural terms, process terms, and outcome terms (Heslop & Lu, 2014). Structure refers to indicators that seek to measure the nursing impact on patients, nursing, and setting (Heslop & Lu, 2014). Process refers to indicators that seek to measure the nursing intervention and/or nursing practice (McCloskey & Bulechek, 2000). Outcome indicators relate to patient safety, perceptions, use of health care, functional status, and clinical management related to nursing care (Doran, 2011).

F. Quality Indicator Development and Pilot Testing Processes

Table 1. Indicator Development Process

- 1. Review scientific literature for: (a) evidence that some aspect of nursing care has an effect on a patient outcome; (b) specific definitions of the indicators; and (c) evidence that the indicators can be validly and reliably measured.
- 2. Collect information from researchers in the field on threats to reliability and validity.
- 3. Conduct expert review of draft indicator definitions, data collection guidelines, and data collection forms.
- 4. Distribute revised definitions, guidelines, and forms to clinical experts for comments on face validity and feasibility of reliable data collection.
- 5. Incorporate clinical expert feedback and develop revised versions of definitions, guidelines, and forms.
- 6. Conduct a pilot study (Table 2) using the draft data collection materials and review data; also interview hospital study coordinators to identify additional threats to reliability and validity.
- 7. Finalize definitions, data collection guidelines, and forms.
- 8. Train database participants in standardized data collection practices.

Source: Montalvo, 2007. Used with permission.

Table 2. Pilot Testing Process

- 1. Develop the indicator with draft guidelines and data collection instruments.
- 2. Recruit pilot testers via email and phone.
- 3. Select pilot sites from those interested. Sites are selected for hospital/unit diversity.
- 4. Guide pilot sites in collecting data according to the draft guidelines.
- 5. Analyze data submitted by pilot sites.
- 6. Collect written and telephone evaluations to assess for clarity, feasibility, and assessment of threats to validity and reliability.
- 7. Analyze pilot data for indicator refinement.
- 8. Finalize guidelines and instruments for dissemination.

Source: Montalvo, 2007. Used with permission.

G. American Nurses Association Proposed Measure Topic Areas from ANA Ambulatory Summit

In January 2014, ANA held the Ambulatory Measurement Summit. Members of the AAACN NSI Task Force and other ambulatory care leaders provided valuable input to this multi-organizational effort to identify important topic areas for ambulatory care nurse-sensitive indicators. The group reached consensus on five indicator topic areas:

- Pain Assessment and Follow Up
- Depression Screening
- Medication Reconciliation
- Readmissions
- Hypertension (Lewis, 2014)

The ANA Ambulatory Measurement Summit included 45 experts from across the nursing care continuum, including 10 from the AAACN NSI Task Force and AAACN leadership. Prior to the summit, preliminary work was done to evaluate existing ambulatory quality measures from the NQF and the National Committee for Quality Assurance (NCQA) that might serve as useful nursing quality measures. Special emphasis was put on measures in which nursing care or input is necessary or expected. As the American Nurses Credentialing Center's (ANCC) Magnet Recognition Program®, a subsidiary of ANA, requires the submission of eight quarters of data for all indicators in its 2014 Manual, the timeline was very tight to identify, test, and validate the selected measures for hospitals planning to apply for ANCC Magnet Recognition® in 2016 (Martinez, Battaglia, Start, Mastal & Matlock, 2015).

A definition of ambulatory care nursing was provided by AAACN members in attendance. AAACN NSI Task Force members as well as AAACN leaders advocated for indicators that were reflective of the lifespan as well as applicable to telephonic management of patients, with consideration of the variety of ambulatory care settings.

There has been no additional citation or communication related to the development of four of these indicators. Press Ganey NDNQI published *Care Coordination: Medication Reconciliation* in August 2014, reflective of the medication reconciliation measure topic area discussed at this summit.

H. AAACN Nurse-Sensitive Indicator Task Force Journey

The AAACN NSI Task Force has been working on many activities to ascertain both short- and long-term opportunities to meaningfully measure the role of the nurse in the ambulatory care setting. This was accomplished using a variety of methods:

- Reviewed historical efforts to measure nursing quality of care and outcomes.
- Conducted a literature review of quality metrics and best practices in ambulatory care.
- Elicited expert opinion.

- Reviewed current standards, scope, definitions, and dimensions of ambulatory care registered nurses.
- Reviewed all current endorsed measures in the ambulatory care and acute health care settings.
- Collaborated in identifying ambulatory care nursing measure topic areas at the ANA Summit in January 2015.
- Shared AAACN member opinions and feedback of quality activities at national meetings and through online surveys (Martinez et al., 2015; Mastal et al., 2012).
- Utilized an evidence-based practice approach to identify and develop multiple indicators for use in the ambulatory care environment. Those indicators are summarized in this report with suggestions for short-term and long-term development.
- Specified the process for developing appropriate indicators.
- Outlined the pilot testing process.
- Advocated for and supported the collaboration with CALNOC to assist in the development of proposed indicators.

These activities and strategies have led the task force to propose indicators that reflect the practice of ambulatory care RNs as well as measure key elements of the national quality strategy. Proposed measures may act as catalysts to assist in assigning value to key ambulatory care nursing role dimensions that have great value and potential for positive patient outcomes, yet have been inadequate and or inconsistently measured. The task force and AAACN are fully aware that nursing quality metrics will be an ongoing process as future changes occur in ambulatory care settings.

SECTION 2 Review of Existent Endorsed Ambulatory Care Nurse-Sensitive Indicator Measures

2.1 Press Ganey National Database of Nursing Quality Indicators (NDNQI)/ Press Ganey, Inc.

- A. Care Coordination: Medication Reconciliation
- B. Care Coordination: Pending Diagnostic Test Results

2.2 Collaborative Alliance for Nursing Outcomes (CALNOC)

- A. Ambulatory Site Demographics
- B. Ambulatory Volume Measures (Denominators)
- C. Ambulatory Staffing, Skill Mix, and Patient Care Hours
- D. Ambulatory Surgery Adverse Outcomes of Care: Wrong Site, Wrong Side, Wrong Patient, Wrong Procedure, Wrong Implant
- E. Ambulatory Surgery Patient Burns
- F. Ambulatory Surgery Patient Falls
- G. Ambulatory Surgery Patient Injury Falls
- H. Ambulatory Surgery Hospital Transfer/Admission

SECTION 2 REVIEW OF EXISTENT ENDORSED AMBULATORY CARE NURSE-SENSITIVE INDICATOR MEASURES

Currently, there are existing, endorsed ambulatory care nurse-sensitive indicators from the Press Ganey National Database of Nursing Quality Indicators (NDNQI)/Press Ganey, Inc. and the Collaborative Alliance for Nursing Outcomes (CALNOC). These indicators are reviewed in this section and are ready to be used by any organization that wishes to benchmark against other organizations within each respective database. The format for these indicators was replicated largely from the format displayed in each respective citation, and as such, differs from the format in the remainder of this report.

2.1 Press Ganey National Database of Nursing Quality Indicators (NDNQI)/ Press Ganey, Inc.

Measures displayed in this section are included with the permission of the Press Ganey National Database of Nursing Quality Indicators (NDNQI)/Press Ganey, Inc.

A. Care Coordination: Medication Reconciliation

Measure Source: Press Ganey NDNQI Guidelines for Data Collection and Submission on Care Coordination in Ambulatory Settings (NDNQI, 2014).

Measure Description: Explore relationship of nurse staffing to improving patient outcomes related to reducing potential for adverse events by providing patients with a Reconciled Medication List.

- Percent of patients who received a Reconciled Medication List.
- Percent of patients who received education about their Reconciled Medication List.

Numerator Statement:

- # of visits in which patient received a Reconciled Medication List for the reporting period.
- # of visits in which patient received education about their Reconciled Medication List for the reporting period.

Denominator Statement: # of Patient visits for the reporting period.

- **Inclusions:** Patient visits include any visit by an eligible patient during the reporting period. Patients are counted on the day of arrival.
- Exclusions: Patients with acute emergencies that warranted transfer to a
 higher level of care/setting (urgent care/emergency room/hospital
 admission). Patients who experience death. Patients who leave without
 being seen, leave before treatment is complete, or leave against medical
 advice.

Rationale: The purpose of the care coordination indicator is to explore the relationship of nurse staffing to improving patient outcomes related to:

Reducing potential for adverse events by providing patients with a Reconciled Medication List.

AAACN NSI Task Force Recommendations and Advocacy Statements

- 1. Population included must attempt to take into account all ages of patients across the lifespan.
- 2. AAACN NSI Task Force and AAACN members strongly believe that the term "Medication Reconciliation" is reflective of the acute care setting and that the more appropriate term and measure would be "Medication Review and/or Education."
- 3. Practice reality in the ambulatory care setting related to medication reconciliation incorporates many staff members, both licensed and unlicensed, and both provider and non-provider as participants in this activity, which lends to its weakness as an indicator of nursing specifically.
- 4. The reality in many ambulatory care settings is that the review may be completed by the registered nurse, but discontinuation of medications does not fall under the nurse role or license in most states. Therefore the list may be reviewed but not reconciled.
- 5. Organizations measure medication reconciliation with the printing of the After Visit Summary (AVS) at the end of the visit. The AVS is not an accurate measure of a correct, reviewed, and reconciled medication list.

B. Care Coordination: Pending Diagnostic Test Results

Measure Source: Press Ganey NDNQI Guidelines for Data Collection and Submission on Care Coordination in Ambulatory Settings (NDNQI, 2014).

Measure Description: Explore relationship of nurse staffing to improving patient outcomes related to reducing potential for adverse events by providing patients with a list of pending test results and follow-up instructions for receipt of results.

- Percent of patients who received a list of pending results and follow-up instructions for receipt of results.
- Percent of patients who received education about their list of pending results and follow-up instructions for receipt of results.

Numerator Statement:

- # of visits in which patient received a list of pending results and follow-up instructions for receipt of results.
- # of visits in which patient received education about list of pending results and follow-up instructions for receipt of results.

Denominator Statement: # of visits by patients for the reporting period.

- **Inclusions:** Patient visits include any visit by an eligible patient during the reporting period. Patients are counted on the day of arrival.
- **Exclusions:** Patients who did not have any testing performed or planned tests. Patients with acute emergencies that warranted transfer to a higher level of care/setting (urgent care/emergency room/hospital admission). Patients who experience death. Patients who leave without being seen, leave before treatment is complete, or leave against medical advice.

Rationale: The purpose of the care coordination indicator is to explore the relationship of nurse staffing to improving patient outcomes related to:

• Reducing potential for adverse events by providing patients with a list of pending test results and follow-up instructions for receipt of results.

AAACN NSI Task Force Recommendations and Advocacy Statements

- 1. Population included must attempt to take into account all ages of patients across the lifespan.
- Normal results, reviewed by a provider may be called to the patient or family by any clinical staff, so the indicator may not be nurse sensitive in the ambulatory care environment.

2.2 Collaborative Alliance for Nursing Outcomes (CALNOC)

Measures displayed in this section were contributed and included with the permission of the Collaborative Alliance for Nursing Outcomes (CALNOC) as part of the collaborative agreement between CALNOC and AAACN. As part of this collaborative relationship, members of the AAACN NSI Task Force were offered the opportunity to participate in the CALNOC ambulatory surgery measures pilot in summer 2015. Because of this opportunity and the feedback given to the AAACN NSI Task Force throughout the development of these indicators, advocacy for key concepts such as measurement of lifespan, outcome and meaningful connection to the ambulatory care environment were expressed throughout their development.

A. Ambulatory Site Demographics

Measure Source: CALNOC (2015)

The ultimate goal of quality measurement is the ability to benchmark performance against other organizations. Measuring performance of an individual unit or units within an organization or system allows nurses to understand performance trends. However, without external comparison groups, it is not possible to understand an individual unit's performance in reference to benchmarks to understand performance within the context of the industry. For example, performance may be improving gradually – a very good outcome of performance measurement. However, the performance may be an outlier and far off from where the rest of the industry performs. Performance may also be very good compared to the industry, and prioritization of improvement efforts could be shifted to other areas of practice. Without external benchmarking, an organization only has anecdotal information to put performance in context.

Measure Description: In order to benchmark with "like" organizations or unit types, data are required to stratify by type of unit. For the initial ambulatory care measures selected in ambulatory surgery centers or procedure units, the following demographics were chosen.

Measure Demographic Collected: Unit/Center Affiliation – Hospital-Based or Freestanding

Hospital-based clinics, ambulatory surgery, and procedure centers generally describe ambulatory care settings with direct access to emergency response teams (not a 911 call) and higher levels of care. Freestanding would not be under the license or billing of a hospital. Reimbursement may also be a driver for the distinction. Provider-based billing differentiates a center that is freestanding, versus hospital billing for a hospital-based center. A freestanding facility would not be on the hospital tax ID and would have billing associated with the practice that owns and operates the site. Freestanding would be Primary Care Network Clinics, Ambulatory Surgery Center (ASC), unlicensed clinics,

not billed as part of a licensed hospital's Centers for Medicare & Medicaid Services (CMS) billing number as an outpatient department of the hospital.

Measure Demographic Collected: Predominant Age Group Served – Pediatric, Adult, Mixed

- Pediatric centers/units are defined as having a predominate population of under 18 years.
- Adult units are defined as having a predominate population over the age of 18.
- Mixed units would be units that include general patient population not defined as pediatric or adult.

B. Ambulatory Volume Measures (Denominators)

Measure Source: CALNOC (2015)

Measure Description: Ambulatory care settings vary widely in size, staff required to provide care, and volume of patients served on a given day. Measures are reported for benchmarking as raw numbers – the counted number of X measures – which are then summarized over a standardized period of time such as a shift, a day, a week, a month, or a year. To be meaningful across settings, the data must also be standardized by creating rates using denominators. For the initial CALNOC measures for ambulatory surgery centers and procedure centers, the following volumes have been selected:

Total Patient Visits: Patient visits may be called by different names (e.g., registrations, admissions, encounters). Our general definition for one visit will be a bundled patient encounter: The patient crosses the threshold, registers, several things may happen to him/her while in the visit (including being sent to lab or x-ray, having one or more procedure/surgery), and then he/she is discharged and leaves the unit/center. This is considered one visit. Total patient visits will be used as a calculation for each participating ambulatory care unit and for each individual calendar month to calculate staffing rates and outcome measures such as fall or adverse events per 1,000 patient visits.

No Shows/Cancellations: Patients who cancel visits/appointments at the last minute or do not show up for their scheduled appointment, not permitting replacement, are included in this total count for the month. This total can be used as a benchmark by facilities interested in overall efficiency of center/unit operations.

Total Surgeries/Procedures: The total number of surgeries/procedures is an "unbundled" count of procedures/surgeries performed in the center/unit for the entire month. There may be more than one procedure in a patient visit. The method for obtaining the total count may vary facility to facility. Some may have standard reports that provide the monthly sum, some may have reports that provide the total by patient, and some may be calculating the total from billing records.

Total OR/Procedure Room Minutes for Surgeries/Procedures: This measure tracks the total amount of time patients spent in the OR/Procedure Room over the entire month. It is the summed amount of time in minutes the patients were actually in the OR or Procedure Suite – using the elapsed time between the time recorded for patient "in room" and "out of room." When a patient has more than one procedure in a visit, add up the cumulative time for all procedures during the visit. "Time in/time out" is a standard measurement in most ORs and procedure suites (e.g., CV lab, Interventional Radiology).

C. Ambulatory Staffing, Skill Mix, and Patient Care Hours

Measure Source: CALNOC (2015)

Measure Description: Direct hours are those related to employees providing face-to-face patient care (back office). Indirect hours are those related to front office employees (registration, IUR-surgery, scheduling, billing). Exclusions include managers, unit billing and registration clerks, medical records staff, monitor techs, and others with no direct patient care responsibilities. Other exclusions include front office staff from the counts of nursing care providers.

Registered Nurse (RN) Nursing Care Hours: Total number of productive hours worked by all registered nurses with direct patient care responsibilities.

Licensed Vocational Nurse (LVN) Nursing Care Hours: Total number of productive hours worked by all licensed vocational nurses (known in some states as Licensed Practical Nurses) with direct patient care responsibilities.

Non-RN/LVN Caregiver Care Hours: Total number of productive hours worked by other Unlicensed Assistants (e.g., Certified Medical Assistant, Patient Care Technician, Nurses Aide). Exclude unit clerks, monitor techs, and others with no direct patient care responsibilities.

Advanced Practice Registered Nurse (APRN) Care Hours: Total number of productive hours worked by advanced practice registered nurses employed by the unit/center. Include Certified Nurse Anesthetists, Clinical Nurse Specialists, Nurse Midwives, and Nurse Practitioners in this category. Exclude APRNs that work as providers for the medical staff as physician extenders.

Other Licensed Professional Hours: Total number of productive hours worked by other licensed professionals employed by the unit/center. Examples of other licensed professionals include physical/occupational therapists, neuropsychologists, physician assistants, licensed radiologic technologists, registered dieticians, medical social workers, and licensed clinical social workers.

D. Ambulatory Surgery Adverse Outcomes of Care: Wrong Site, Wrong Side, Wrong Patient, Wrong Procedure, Wrong Implant

Measure Source: NQF 0267 (NQF, 2015a) and CMS-ASC 3 (CMS, 2015a), included in CALNOC Ambulatory Surgery Measures Set (CALNOC, 2015).

Measure Description: Percentage of Ambulatory Surgery Center (ASC) admissions or procedures experiencing a wrong site, wrong side, wrong patient, wrong procedure, or wrong implant event. Wrong is defined as not in accordance with intended site, side, patient, procedure, or implant.

Numerator Statement: ASC admissions or procedure patients experiencing a wrong site, wrong side, wrong patient, wrong procedure, or wrong implant.

Denominator Statement: All ASC admissions or procedure patients.

Rationale/Supportive Measures:

- CMS for ASC: CY 2014, 2015, 2016 payment
- CMS Ambulatory Surgical Center Quality Reporting Program

E. Ambulatory Surgery Patient Burns

Measure Source: NQF 0263 (NQF, 2015b) and CMS-ASC 1 (CMS, 2015b), included in CALNOC Ambulatory Surgery Measures Set (CALNOC, 2015).

Measure Description: Percentage of Ambulatory Surgery Center (ASC) admissions or procedure patients experiencing a burn prior to discharge.

Numerator Statement: ASC admissions or procedure patients experiencing a burn prior to discharge. Burn is defined as an unintended tissue injury caused by any of the six recognized mechanisms: scalds, contact, fire, chemical, electrical, or radiation (e.g. warming devices, prep solutions, electrosurgical unit or laser).

Denominator Statement: All ASC admissions (procedures).

Rationale/Supportive Measures:

- CMS for ASC: CY 2014, 2015, 2016 payment
- CMS Ambulatory Surgical Center Quality Reporting Program

F. Ambulatory Surgery Patient Falls

Measure Source: NQF 0266 (NQF, 2014a) and CMS-ASC 2 (CMS, 2015c), included in CALNOC Ambulatory Surgery Measures Set (CALNOC, 2015).

Measure Description: Percentage of Ambulatory Surgery Center (ASC) admissions or procedure patients experiencing a fall. Fall is defined as a sudden, uncontrolled, unintentional, downward displacement of the body to the ground or other object, excluding falls resulting from violent blows or other purposeful actions.

Numerator: ASC admissions or procedure patients experiencing a fall within the confines of the ASC or procedure unit.

Denominator: All ASC admissions or procedure patients.

Numerator Inclusion: ASC admissions or procedure patients experiencing a fall within the confines of the ASC.

Exclusion: ASC admissions or procedure patients experiencing a fall outside the ASC.

Rationale/Supportive Measures:

• CMS for ASC: CY 2014, 2015, 2016 payment

G. Ambulatory Surgery Patient Injury Falls

Measure Source: CALNOC (CALNOC, 2015)

Measure Description: Percentage of Ambulatory Surgery Center (ASC) admissions or procedure unit visit patients experiencing a fall with injury. Fall is defined as a sudden, uncontrolled, unintentional, downward displacement of the body to the ground or other object, excluding falls resulting from violent blows or other purposeful actions. Injury is defined with the following Level of Injury Scale:

- 1=None. No injury as a result of fall.
- 2=Mild/Minor. Resulted in bruise or abrasion, and/or required application of a dressing, ice, cleaning of a wound, limb elevation, or topical medication (band aid pediatric patients).
- 3=Moderate. Resulted in muscle or joint strain, and/or required suturing, application of steristrips/skin/glue, or splinting.
- 4=Major. Resulted in surgery, casting, traction, fracture, or required consultation for neurological or internal injury.
- 5=Death. Fall determined to be cause of death.
- Note: Determine injury level at the time of discharge. X-ray, CT scan or other radiological evaluation resulting in a finding of no injury, with no treatment and no signs or symptoms of injury, select "None." Patients with coagulopathy who receive blood products as a result of a fall, select "4 Major."

Numerator Statement: Total number of patient falls of injury level minor or greater (whether or not assisted by a staff member) by eligible unit during the calendar month X 1,000.

Denominator Statement: Patient visits by Type of Unit during the calendar month.

H. Ambulatory Surgery Hospital Transfer/Admission

Measure Source: NQF 0265 (NQF, 2015c) and CMS-ASC 4 (CMS, 2015d), included in CALNOC Ambulatory Surgery Measures Set (CALNOC, 2015).

Measure Description: Rate of Ambulatory Surgery Center (ASC)/Procedure Unit admissions or visits requiring a hospital transfer or hospital admission upon discharge. Modified to include all procedure units.

Numerator Statement: ASC or Procedure Unit admissions requiring a hospital transfer or hospital admission upon discharge.

Denominator Statement: All ASC or Procedure Unit admissions.

Exclusions: None.

Rational/Supportive Measures:

CMS for ASC: CY 2014, 2015, 2016 payment

SECTION 3 AAACN NSI Task Force Proposed Indicators for Development and Endorsement: Existent Health Care Measures Adapted and Recommended as Ambulatory Care Nurse-Sensitive Indicator Measures

- A. Ambulatory Care Nurse Readmission Across the Lifespan
- B. Ambulatory Care Nurse Pain Assessment and Follow Up
- C. Ambulatory Care Nurse Screening for High Blood Pressure and Follow-up Care
- D. Ambulatory Care Nurse Screening and Follow-Up Documentation for Depression
- E. Ambulatory Care Nurse Patient Falls in the Institution
- F. Ambulatory Care Nurse Screening for Future Falls Risk
- G. Ambulatory Care Nurse Screening for Body Mass Index (BMI)
- H. Ambulatory Care Nurse RN Demographics
 - 1. Diploma RN
 - 2. Associate's Degree in Nursing RN
 - 3. Bachelor's of Science in Nursing RN
 - 4. Master's of Science in Nursing RN
 - 5. Doctorate in Nursing RN
 - 6. Certification RN
 - 7. Certification RN in Field of Specialty
 - 8. Turnover by Full-Time Equivalent
 - 9. Vacancy by Full-Time Equivalent
- I. Ambulatory Care Nurse Patient Satisfaction

SECTION 3

AAACN NSI TASK FORCE PROPOSED INDICATORS FOR DEVELOPMENT AND ENDORSEMENT: EXISTENT HEALTH CARE MEASURES ADAPTED AND RECOMMENDED AS AMBULATORY CARE NURSE-SENSITIVE INDICATOR MEASURES

Indicators in this section were found through a thorough analysis of existent, endorsed measures in health care. Most of them are non-specific to nursing in their original endorsed format. The AAACN NSI Task Force reviewed and proposed the following adaptations to utilize them as sensitive to the nurse in the ambulatory care setting.

A. AAACN Ambulatory Care Nurse-Sensitive Adapted Proposed Measure: Ambulatory Care Nurse Readmission Across the Lifespan

Introduction to the Measure

"Readmission following an acute care hospitalization is a costly and often preventable event" (Horwitz et al., 2011, p. 7). Between 2003 and 2004, approximately one-fifth of Medicare beneficiaries were readmitted within 30 days of discharge (Jencks, Williams, & Coleman, 2009). Hospital readmission is disruptive to patients and caregivers, putting patients at additional risk of hospital-acquired infections and complications (Horwitz et al., 2011). CMS (2015e) cites that the Commonwealth Fund estimates that Medicare could save \$1.9 billion annually if national readmission rates were lowered to the levels achieved by top-performing regions.

A lot of evidence suggests that readmissions are linked to decreased quality of care, lack of care coordination, or other factors within the control of health care clinicians (Horwitz et al., 2011). As such, entities such as CMS and the National Quality Forum (NQF) have implemented measures related to patient readmissions. Initially, readmissions were focused on certain high-risk populations, but CMS now focuses on all condition 30-day readmissions as a quality measure (CMS, 2015e; Horwitz et al., 2011). The ACO risk-standardized all condition readmission quality measure adapted in this report was originally developed as a hospital risk-standardized all condition readmission measure from Yale for CMS (Horwitz et al., 2011; RTI International & Telligen, 2013). The initial readmission measure was created by Yale University and CMS, and then endorsed by NQF. NQF measure #1789 is specific to patients aged 18 years and older and estimates the hospital-level, risk- standardized rate of unplanned, all cause readmissions for any eligible condition within 30 days of hospital discharge for these patients (NQF, 2015).

NQF has also endorsed a pediatric all condition readmission measure that was developed from the Center of Excellence for Pediatric Quality Measurement and is newly commissioned and developed as part of the Agency for Healthcare Research and Quality (AHRQ) and CMS Pediatric Quality Measures Program. NQF notes there are several

several challenges in measuring the pediatric population related to the fact that it is highly dependent on case volume and required risk adjustment for socio-demographic factors, however, because there is a shortage of measures related to the pediatric population, NQF voted to recommend the measure (NQF, 2014b). In 2009-2010, an analysis of data submitted to the National Association of Children's Hospitals and Related Institutions (NACHRI) was performed. The NACHRI Case Mix is the largest data set of children's hospitals. The 30-day readmission rate for children was 6.5% (n=36,734) where 39% (n=14,325) were readmitted within 7 days and 61.6% (n=22.628) were readmitted within 14 days (Berry et al., 2013).

Repeat admission for children felt to be amenable to high-quality outpatient care (asthma and seizure-related admissions, for example, or conditions related to the same medical problem, such as repeat admission for sickle cell crisis) may be considered potentially avoidable (Berry et al., 2011).

CMS ACO 8, NQF 1789, and NQF 2393, as cited below, are the respective names of currently endorsed measures related to readmission. These endorsed measures are described as the risk-adjusted percentage of ACO-assigned beneficiaries who were hospitalized who were readmitted to a hospital within 30 days following discharge from the hospital for the index admission. The numerator is reflective of non-Federal, shortstay, acute care or critical access hospitals, within 30 days of discharge from the index admission included in the denominator, and excluding planned readmissions. The denominator is related to all hospitalizations with the exception of the following: medical treatment of cancer, primary psychiatric disease, rehabilitation care, or fitting of prostheses and adjustment devices for ACO-assigned beneficiaries at non-Federal, short-stay acute care or critical access hospitals, where the beneficiary was aged 65 years or older, was continuously enrolled in fee-for-service Medicare Part A for at least one month after discharge, was not discharged to another acute care hospital, was not discharged against medical advice, and was alive upon discharge and for 30 days postdischarge. All patients not readmitted to a hospital within 30 days following discharge from the hospital are excluded from the measure as currently endorsed.

AAACN NSI Task Force Recommendations for Adaptation to Ambulatory Care RN

The AAACN NSI Task Force strongly believes that the role of the nurse in the ambulatory care setting is meaningful to patients throughout the lifespan, critical to care coordination, and can have an impact on the promotion of health, including, after discharge home from the inpatient setting, through various interventions enacted from the ambulatory care setting.

Ambulatory care nurses follow patient cohorts for many years of their lives and become uniquely acquainted with the specific socio-economic, behavioral, environmental, and patient-specific needs that their patients have. It is the belief of the AAACN NSI Task Force that ambulatory care nurses, with this knowledge, can have an impact on readmission rates.

It is anticipated that the nurse in the ambulatory care setting, when measured by these adaptive recommendations, will prove to have a significant impact in reducing use of the emergency department by specific diagnosis populations and in the achievement of several diagnosis-specific care measures per population such as quarterly pulmonary function tests in the asthma patient or quarterly Hemoglobin A1C measurement in the diabetic patient.

Measure Source: AAACN Adapted Proposed Measure: Ambulatory Care Nurse Readmission Across the Lifespan

Adapted by the AAACN NSI Task Force as an ambulatory care nurse-sensitive measure from the following entities: CMS ACO 8 (CMS, 2015e; RTI International & Telligen, 2012a), NQF 1789 (NQF, 2014b), NQF 2393 (NQF, 2015d).

Measure Description: Risk-adjusted percentage of Accountable Care Organization (ACO) assigned beneficiaries who were hospitalized who were readmitted to a hospital within 30 days following discharge from the hospital for the index admission.

- Percentage of patients in one of below identified populations that receive a
 post-discharge call within 48 hours of emergency department visit after an
 inpatient stay from an ambulatory care nurse responsible for care
 coordination of that patient.
- Percentage of patients in one of below identified populations from the ambulatory care nurse cohort that are readmitted within 30 days of discharge from inpatient setting.

Numerator Statement: Risk-adjusted readmissions at a non-Federal, short-stay, acute-care or critical access hospital, within 30 days of discharge from the index admission included in the denominator, and excluding planned readmissions.

Those patients, with the outlined specific diagnoses and delineated by age, seen in the Emergency Department (ED) more than one time per year, within 30 days of discharge from the inpatient setting.

Numerator Inclusions:

- Pediatric Populations
 - Respiratory
 - Asthma
 - Respiratory Syncytial Virus (RSV)
 - Pneumonia
 - Cardiac
 - Congenital Heart Disease
 - Congestive Heart Failure
 - Diabetes
 - Sickle Cell

- Seizure Disorder
- Adult Populations:
 - Respiratory:
 - Chronic Obstructive Pulmonary Disorder
 - Pneumonia
 - Cardiac:
 - Congestive Heart Failure
 - Diabetes
 - Sickle Cell
 - Stroke

Denominator Statement: Patient visits include any visit by an eligible patient, of any age, during the reporting period. Patients are counted on the day of arrival.

Denominator Inclusions: Patient visits include any visit by an eligible patient during the reporting period. Patients are counted on the day of arrival.

Denominator Exclusions: All hospitalizations not related to medical treatment of cancer, primary psychiatric disease, diagnoses not identified in numerator section of measure, rehabilitation care, or fitting of prostheses and adjustment devices for ACO assigned beneficiaries at non-Federal, short-stay acute care or critical access hospitals, where the beneficiary was aged 65 or older, was continuously enrolled in fee-for-service Medicare Part A for at least one month after discharge, was not discharged to another acute care hospital, was not discharged against medical advice, and was alive upon discharge and for 30 days post-discharge.

Methodology and Next Steps Recommendations

From the comprehensive literature review completed on this measure, several process recommendations for the development and piloting of this measure are outlined here. In Section 5 of this report, two exemplars are described related to the pediatric seizure disorder and asthma populations as well as several others related to the role of the care manager for any population. It is recommended that there be a nurse responsible for care coordination, care facilitation, and navigation of the health system. If this role is not possible, then the following activities are recommended to elicit low readmission rates of cohorts of patients managed by ambulatory care settings:

- 1. Registered nurse (RN) screening of patient with identification of barriers and services needed.
 - a. Follow-up appointment to specialist
 - b. Clinic visit
 - Use of afterhours care instead of ED
 - d. Provision of care instructions or medication follow up that may have prevented need for additional ED visit

- 2. Documented use of teach-back.
- 3. Specific care instructions or written action plan reviewed by RN.
- 4. Post-discharge call by RN lead to:
 - a. Follow-up appointment to specialist
 - b. Clinic visit
 - c. Use of afterhours care instead of ED
 - d. Provision of care instructions or medication follow up that may have prevented need for additional ED visit
- 5. Use of evidence-based practice (specific to diagnosis).
 - a. For example: Asthma patient seen in ED referred to pulmonologist within 48 hours.
 - b. For example: Newly diagnosed diabetic seen in ED or admitted seen in specialty clinic within one week.

B. AAACN Ambulatory Care Nurse-Sensitive Adapted Proposed Measure: Ambulatory Care Nurse Pain Assessment and Follow Up

Introduction to the Measure

The Physician Quality Reporting System (PQRS) 2015 Manual notes "several provisions from the National Pain Care Policy Act (H.R. 756/S. 660) have been included in the PPACA to improve pain care" (Centers for Medicare & Medicaid Services [CMS], 2015f, p. 209). This legislation mandated an IOM conference on pain to address key medical and policy issues affecting the delivery of quality pain care, the establishment of a training program to improve the skills of health care professionals to assess and treat pain, and the enhancement of pain research for the National Institute of Health (CMS, 2015f).

It is estimated that approximately 76.5 million Americans suffer from pain and that pain is the number one reason Americans pursue health care. Uncontrolled or undertreated pain is the leading cause of disability, diminishing quality of life and driving up health care costs (CMS, 2015f). Additionally, there are significant disparities related to pain perception, assessment, and treatment among racial and ethnic minorities (CMS, 2015f).

The Institute of Medicine (IOM) published the report, *Relieving Pain in America: A Blueprint for Transforming Prevention, Care, Education and Research* (2011) that suggested chronic pain rates will continue to increase due many factors including: an increase in chronic diseases that cause pain; an increase in obesity that is associated with chronic conditions causing pain; progress in lifesaving techniques for catastrophic injuries for people who should have previously died and now are at risk for lifelong pain; increased surgical related pain and an increasingly better understanding by the public of chronic pain syndromes and associated new treatments for those syndromes (IOM, 2011).

The Institute for Clinical Systems Improvement suggests chronic pain assessment should include determination of the mechanisms of pain through documentation of pain location, intensity, quality, onset, and duration, as well as functional ability and goals, as well as psychological and social factors such as depression or substance abuse (Hooten et al., 2013). The description of PQRS Measure #131 also states that the endorsed measures described here "may be reported by eligible professionals who perform the quality actions described in the measure based on the services provided" (CMS, 2015f). The AAACN NSI Task Force believes registered nurses (RNs) to be qualified for reporting of this measure.

National Quality Forum (NQF) 0420 (NQF, 2013a) and PQRS 131 (CMS, 2015f), as cited below, are the respective names for current existent endorsed measures related to pain assessment and follow up. They are described as the percentage of patients aged 18 years and older with documentation of a pain assessment through discussion with the patient including the use of a standardized tool(s) on each visit and documentation of a follow-up plan when pain is present. Their numerator is described by capturing the patient's pain assessment in documentation of a discussion with the patient using a standardized tool as well as a follow-up plan when pain is present. Their denominator captures patients aged 18 years and older on the date of the encounter. Exclusions that apply to these endorsed measures are for: severe mental and/or physical incapacity where the person in unable to express himself/herself in a manner understood by others and or when the patient is in an urgent or emergent situation where time is of the essence and to delay treatment would jeopardize the patient's health status.

AAACN NSI Task Force Recommendations for Adaptation to Ambulatory Care RN

Measure Source: AAACN Adapted Proposed Measure: Ambulatory Care Nurse Pain Assessment and Follow Up

Adapted by the AAACN NSI Task Force as an ambulatory care nurse-sensitive measure from the following entities: NQF 0420 (NQF, 2013a) and PQRS 131 (CMS, 2015f).

Measure Description: Percentage of patients of all ages with documentation of a pain assessment through discussion with the patient including the use of a standardized tool(s) on each visit AND documentation of a follow-up plan when pain is present.

Numerator Statement: Patient's pain assessment is documented through discussion with the patient including the use of a standardized tool(s) and a follow-up plan is documented when pain is present.

Denominator Statement: Patients of all ages on the date of the encounter.

Denominator Exclusions:

- Severe mental and/or physical incapacity where the person is unable to express himself/herself in a manner understood by others.
- Patient is in an urgent or emergent situation where time is of the essence and to delay treatment would jeopardize the patient's health status.

Methodology and Next Steps Recommendations

The AAACN NSI Task Force believes the assessment of pain and the resultant care planning associated with relief of pain to be highly linked to patient care from the ambulatory care nurse. As such, wherever possible, the AAACN NSI Task Force

recommends adoption of this measure and integration of its processes for data collection into electronic medical records.

C. AAACN Ambulatory Care Nurse-Sensitive Adapted Proposed Measure: Ambulatory Care Nurse Screening for High Blood Pressure and Follow-up Care

Introduction to the Measure

"Hypertension is a chronic condition that can lead to heart disease, stroke, and other diseases that can result in premature death" (Kung and Xu, 2015, p. 1). One of the objectives of Healthy People 2020 is to reduce the number of persons in the population with hypertension (Office of Disease Prevention and Health Promotion [ODPHP 2015). Approximately 1 in 3 U.S. adults have high blood pressure, and of those, only about half (52%) have their high blood pressure under control. This common condition increases the risk for heart disease and stroke, and both are leading causes of death for Americans (Farley, Dalal, Mostashari, & Frieden, 2010; Kung & Xu, 2015).

Despite considerable improvements in increasing awareness, treatment, and control of hypertension, in 2007-2008, approximately half of adults with hypertension did not have their blood pressure under control. Because of the fundamental role of hypertension in cardiovascular health, *Healthy People 2010* included national objectives to reduce the proportion of adults aged ≥20 years with hypertension to 14% from a baseline of 26%, and to increase the proportion of adults aged ≥18 years with hypertension whose blood pressure is under control to 68% from a baseline of 25% (ODPHP, 2015). Americans increase their risk for heart disease, stroke, or kidney disease at a potential cost of \$76.6 billion a year in health care services, medications, and missed days of work. The number of deaths from essential hypertension and hypertensive renal disease is 30,770 per 100,000 population (Nwankwo, Yoon, Burt, & Gu, 2013; NCQA, 2014).

The National Committee for Quality Assurance (NCQA, 2014) added age- and condition-specific treatment goals to its Healthcare Effectiveness and Data Information Set measures that align with the eighth Joint National Committee hypertension guidelines for controlling high blood pressure:

- 18–59 years (<140/90 mm Hg)
- 60–85 years with diabetes (<140/90 mm Hg)
- 60–85 years without diabetes (<150/90 mm Hg).

Although hypertension is very common in adults, approximately 1 to 5 out of every 100 children and adolescents also have hypertension. Unlike in adults, where blood pressure above 130/80 suggests further monitoring or evaluation, normal values for children's blood pressure are determined by age, gender, and height. The skill of the person obtaining the blood pressure affects accuracy (Battaglia, 2006). According to the National High Blood Pressure Education Program Working Group (NHBPEP, 2004),

children aged 3 years and older should have a BP evaluated in a medical setting at least once a year. Children younger than 3 years should have a BP measured under the following special circumstances: history of prematurity, neonatal complications requiring intensive care, congenital heart disease, and treatment with drugs known to raise blood pressure (NHBPEP, 2004). The strongest risk factor for hypertension in children and adolescents is being overweight. Finding and treating hypertension early in young people could lower their risk for complications during adulthood. To help clinicians decide whether to screen, it is necessary to know whether early detection actually improves health outcomes. It is also important to consider the potential harms of starting blood pressure medications and other treatments in young people (Moyer & U.S. Preventive Services Task Force [USPSTF], 2013).

With an estimated prevalence of between 1% and 5%, hypertension is a common chronic disease in children (Moyer & USPSTF, 2013). Pediatric hypertension may be secondary to another disease process or it may be essential hypertension. Secondary hypertension is more common in children than in adults, and common causes of hypertension in children include renal disease, coarctation of the aorta, and endocrine disease. However, as with adults, the majority of children and adolescents with mild to moderate hypertension have primary hypertension in which a cause is not identifiable (Moyer & USPSTF, 2013). Hypertension in children has been shown to correlate with family history of hypertension, low birth weight, and excess weight. With the increasing prevalence of childhood weight problems, increased attention to weight-related health conditions including hypertension is warranted. Several lines of evidence suggest that blood pressure in US children and adolescents is increasing in parallel with weight (Moyer & USPSTF, 2013).

In addition, the NHBHEP (2004) has published standards for addressing prehypertension and hypertension in children and adolescents. Blood pressure evaluation is within the scope of the RN. Many pre-hypertension interventions may be completed by an RN. Blood pressure monitoring and evaluations are frequently accomplished during a nurse-only visit in the outpatient setting.

The USPSTF concludes that current evidence is insufficient to assess the balance of benefits and harms of screening for primary hypertension in asymptomatic children and adolescents to prevent subsequent cardiovascular disease in childhood or adulthood (Moyer & USPSTF, 2013). There was no direct evidence demonstrating that screening children and adolescents for hypertension is effective in delaying onset of or reducing risk for adverse cardiovascular health outcomes related to hypertension, either in childhood or adulthood (Moyer & USPSTF, 2013; USPSTF, 2007).

ACO #21 (RTI International & Telligen, 2012b), NQF 0018 (NQF, 2013b), and GPRO PREV #11 (CMS, 2014a) are the respective names of currently endorsed measures related to screening and follow up documentation of high blood pressure.

AAACN NSI Task Force Recommendations for Adaptation to Ambulatory Care RN

The AAACN NSI Task Force believes the role of the nurse in the ambulatory care setting is meaningful throughout the lifespan of the patient. Hypertension is a prevalent condition that contributes to important adverse health outcomes, including premature death, heart attack, renal insufficiency, and stroke. The USPSTF (2007) found good evidence that blood pressure measurement can identify adults at increased risk for cardiovascular disease from high blood pressure (RTI International & Telligen, 2012b).

This measure takes into consideration the already endorsed measures from CMS ACO #21, NQF 0018, and GPRO PREV #11 as cited below. Pediatric patients between 1 and 18 years of age are included in the measure, as hypertension in children has been shown to correlate with family history of hypertension, low birth weight, and excess weight. With the increasing prevalence of childhood weight problems, increased attention to weight-related health conditions including hypertension is warranted (Moyer & USPSTF, 2013), even though the USPSTF concludes that the current evidence insufficient to link screening and preventative activities in children with prevention of subsequent cardiovascular disease in childhood or adulthood (Moyer & USPSTF, 2013). AAACN believes there is benefit to screen pediatric patients with elevated BMI and family history of hypertension.

Determination of standardized method of data extraction: Electronic Medical Record, (EMR), screening tool used, and documentation process for follow-up care.

Note: ACO 28 also talks about hypertension (HTN): Controlling high blood pressure (NQF 0018 as it relates to patients with a known diagnosis of HTN; measure 21 is a screening measure).

Measure Source: AAACN Adapted Proposed Measure: Ambulatory Care Nurse Screening for High Blood Pressure and Follow-up Care

Adapted by the AAACN NSI Task Force as an ambulatory care nurse-sensitive measure from the following entities: ACO #21 (RTI International & Telligen, 2012b), NQF 0018 (NQF, 2013b), and GPRO PREV #11 (CMS, 2014a).

Measure Description: Percentage of patients at all ages seen during the measurement period who were screened for high blood pressure (BP) and a recommended follow-up plan is documented based on the current blood pressure reading as indicated.

- Percentage of patients in one of the identified populations that received a blood pressure screening during a visit encounter.
- Percentage of patients in one of the identified populations who have a follow-up plan of care for hypertension.

Numerator Statement: Patients greater than 1 year of age who were screened for high blood pressure and a recommended follow-up plan is documented as indicated if the blood pressure is pre-hypertensive or hypertensive.

Numerator Inclusions: Number of patients screened for hypertension between 1 or greater years of age at time of encounter.

 Age, gender, and height are required to evaluate blood pressure in children. Blood pressure cuff size ranges may also be a factor in accurate measurement.

Denominator Statement: All patients aged 1 year and older at the beginning of the measurement period.

Denominator Inclusions: All patients aged 1 year and older at the beginning of the measurement period.

Denominator Exclusions:

- Individuals with an active diagnosis of HTN at the first blood pressure screening during the measurement year.
- All patients aged 1 year to less than 18 years with a diagnosis of renal disease, congenital heart disease, endocrine disease, prematurity, neonatal complications resulting in intensive care, or under treatment with a medication causing high blood pressure.

Methodology and Next Steps Recommendations

From the comprehensive literature review completed on this measure, several process recommendations for the development and piloting of this measure are outlined here. It is recommended that there be a nurse responsible for hypertension screening and documentation of a follow-up plan of care.

- RN screening of patient with hypertension for identification of barriers and services needed.
- Follow-up appointment with Primary Care Provider (PCP).
- Referral to weight-loss/nutrition specialist.
- Instruction on diet, activity, and medications (if applicable).
- Documented use of teach-back.
- Specific care instructions or written action plan reviewed by RN.
- Use of evidence-based practice.

D. AAACN Ambulatory Care Nurse-Sensitive Adapted Proposed Measure: Ambulatory Care Nurse Screening and Follow-Up Documentation for Depression

Introduction to the Measure

In the United States, depression affects approximately 9% of patients and accounts for more than \$43 billion in medical care costs and \$17 billion in lost productivity annually. Depression is projected to become the second largest cause of disability by 2020 (Maurer, 2012).

Depression is under treated. Even when treated appropriately, more than 75% of patients with depression have recurrent episodes and up to 30% have residual symptoms (Maurer, 2012). Depression has been associated with worsened outcomes in patients with a variety of medical conditions, such as coronary artery disease, diabetes mellitus, and stroke (Ciechanowski, Katon, & Russo, 2000; Ford et al., 1998; Robinson, Bolduc, & Price 1987). The Centers for Disease Control and Prevention (CDC) recognizes the mental health of workers is an area of increasing concern to organizations, as depression causes disability, absenteeism, and loss of productivity among working-age adults (CDC 2013). In a three-month period, patients with depression miss an average of 4.8 workdays and suffer 11.5 days of reduced productivity. Being able to identify major depression in the workplace is complicated as there are a number of issues which may cause workers to avoid screening such as their concerns about confidentiality or the impact it may have on their job (CDC, 2013; Maurer, 2012; NQF, 2015e).

Treatment of depression may reduce mortality from these conditions, as well as help prevent suicide. Therefore, accurately identifying patients who have depression is important so that appropriate treatment can be initiated (CDC, 2013; Gibbons, Hur, Bhaumik, & Mann, 2005; Jorge, Robinson, Arndt, & Starkstein, 2003; Maurer, 2012; Taylor et al., 2005).

The U.S. Preventive Services Task Force (USPSTF) recommends screening adolescents and adults in clinical practices that have systems in place to ensure accurate diagnosis, effective treatment, and follow-up care. It does not recommend for or against screening for depression in children seven to 11 years of age or screening for suicide risk in the general population (Maurer, 2012; USPSTF, 2009).

The USPSTF found evidence that treatment with antidepressants, psychotherapy, or both decreases clinical morbidity and improves outcomes in adults with depression, as identified through screening in primary care settings. Screening adults for depression is recommended for practices that have systems in place to ensure accurate diagnosis, effective treatment, and follow-up care. Additionally, the USPSTF found no evidence of harms of screening for depression in adults (Maurer, 2012; USPSTF, 2009).

The USPSTF (2007) found insufficient evidence to recommend for or against screening for suicide risk in the general population, compared with screening only those with depression.

The USPSTF recommends screening adolescents 12 to 18 years of age for depression in clinical practices that have systems in place to ensure accurate diagnosis, psychotherapy, and follow-up care. There is insufficient evidence to balance the benefits and harms of depression screening in children seven to 11 years of age (Richardson et al., 2010; USPSTF, 2007).

Many instruments have been developed for depression screening and yet none have been supported as superior (Maurer, 2012; USPSTF, 2007). Positive results on a depression screening test should trigger a follow-up process that incorporates full diagnostic interviews that use standard diagnostic criteria from the *Diagnostic and Statistical Manual of Mental Disorders*, 4th ed. (DSM-IV) (Kroenke, Spitzer, Williams, & Löwe: 2010; USPSTF, 2007, 2009).

Succinct screening instruments, such as the Patient Health Questionnaire-2 (PHQ-2) may rule out, but not definitively diagnose, depression (Mitchell & Coyne, 2007). The PHQ-2 is as effective as such longer screening instruments as the Beck Depression Inventory or Zung Depression Scale (Arroll, Khin, & Kerse, 2003; Gilbody, House, & Sheldon, 2005; Whooley, Avins, Miranda, & Browner, 1997). The PHQ-2 and PHQ-9 are commonly used and validated screening tools. The PHQ-2, which asks two simple questions about mood and anhedonia, has a 97% sensitivity and 67% specificity in adults, whereas the PHQ-9 has a 61% sensitivity and 94% specificity in adults. The PHQ-2 is reported to have a 74% sensitivity and 75% specificity in adolescents (PHQ-A) (Arroll et al., 2010; Löwe, Kroenke, & Gräfe, 2005; Richardson et al., 2010).

The PHQ-2 performed reliably to a standard diagnostic interview as well as established depression scales and proved sensitive to change (Lowe et al., 2005) Thus, the PHQ-2 can be used as a brief multipurpose measure for detecting depression, grading its severity, and monitoring outcomes over time. If the PHQ-2 is positive for depression, the PHQ-9 should be administered. If these screening tests are positive for depression, further evaluation is needed to confirm that the patient's symptoms meet the *Diagnostic* and Statistical Manual of Mental Disorders' criteria for diagnosis. The PHQ offer clinicians concise, self-administered screening and diagnostic tools for mental health disorders that have been field-tested in the office practice setting. The screenings are quick and user-friendly, improving the recognition rate of depression and anxiety and facilitating diagnosis and treatment (Arroll et al., 2010).

The PHQ-9 is one of the most common instruments used for depression screening (Maurer, 2012). Although it can be used on its own as a screening test or to monitor treatment, it is often administered for confirmation of a positive PHQ-2 result. The PHQ-9 has been validated, takes under five minutes to complete, and has demonstrated 61% sensitivity and 94% specificity for mood disorders in adults, and 89.5% sensitivity and 77.5% specificity in adolescents (Arroll et al., 2010; Maurer, 2012; Richardson et al., 2010).

CMS 2v3 (CMI, 2014), ACO #18 (RTI International & Telligen, 2012c), GPRO PREV-12 (CMS, 2014b), and NQF #0418 (NQF, 2014c) are the respective names of currently endorsed measures related to screening and follow-up documentation of depression.

AAACN NSI Task Force Recommendations for Adaptation to Ambulatory Care RN

The AAACN NSI Task Force believes the role of the nurse in the ambulatory care setting is meaningful throughout the lifespan of the patient. The World Health Organization, as seen in Pratt & Brody (2008), found that major depression was the leading cause of disability worldwide (Lopez & Murray, 1998). Depression causes suffering, decreased quality of life, and impairment with social and/or occupational functioning (Wells et al., 1989). Depression is associated with increased health care costs as well as higher rates of many chronic medical conditions (Katon, 2003). The negative outcomes associated with early onset depression make it crucial to identify and treat depression in its early stages. Major depressive disorder (MDD) is a debilitating condition that has been increasingly recognized among youth, particularly in adolescents. The economic burden of depression is substantial for individuals as well as society. Costs to an individual may include suffering, possible side effects from treatment, fees for mental health and medical visits and medications, time away from work and lost wages, transportation, and reduced quality of personal relationships (CMS 2v3, ACO # 18).

The nurse specificity is not certain as the screening can be done by the RN using a standardized tool and place for documentation, though the follow-up plan must be made by the patient's provider. The measure is a process and not an outcomes measure, and could be used for patients of all ages as the literature doesn't recommend for or against screening for patients < 12 years of age. The AAACN NSI Task Force advises the use of PHQ-2 and PHQ-9, and PHQ-A (for adolescents).

Measure Source: AAACN Adapted Proposed Measure: Ambulatory Care Nurse Screening and Follow-Up Documentation for Depression

Adapted by the AAACN NSI Task Force as an ambulatory care nurse-sensitive measure from the following entities: CMS 2v3 (CMI, 2014), ACO #18 (RTI International & Telligen, 2012c), GPRO PREV-12 (CMS, 2014b), NQF #0418 (NQF, 2014c).

Measure Description: Ambulatory Care Nurse Screening and Follow-Up Documentation for Depression

- Percentage of patients in one of the identified populations that received a depression screening during a visit encounter.
- Percentage of patients in one of the identified populations who screened positive and have a follow-up plan of care for depression,

Numerator Statement: Percentage of patients aged 12 years and older screened for clinical depression during the measurement period using an age appropriate standardized depression screening tool and if positive, a follow-up plan is documented on the date of the positive screen.

Numerator Inclusions:

- Patients aged 12 years to less than 18 years who screened positive for depression using PHQ-A AND have a follow-up plan documented on date of positive screen.
- Patients 18 years of age or older who screened positive for depression using PHQ-2 and PHQ-9 AND have a follow-up plan documented on date of positive screen.

Denominator Statement: All patients aged 12 years and older before the beginning of the measurement period with at least one eligible encounter during the measurement period.

Denominator Inclusions: All patients aged 12 years and older at the beginning of an eligible encounter.

Denominator Exclusions: All patients aged 12 years and older at the beginning of a measurement period who have already screened positive for depression and/or are being treated for depression prior to eligible encounter.

 Any patient unable to be screened due to physical or cognitive issues during an eligible encounter.

Methodology and Next Steps Recommendations

From the comprehensive literature review completed on this measure, several process recommendations for the development and piloting of this measure are outlined here:

- Registered Nurse (RN) screening of patient, using PHQ-2 and PHQ-9, or PHQ-A, with identification of barriers and services needed.
- Follow-up appointment with primary care provider (PCP).
- Referral to specialist or appropriate community services.
- Instruction on diet, activity, and medications (if applicable).
- Documented use of teach-back.
- Specific care instructions or written action plan reviewed by RN.
- Use of evidence-based practices.

 There is a need to determine a standardized method of data extraction. The Electronic Medical Record (EMR), PHQ-2 and PHQ-9, documentation process for follow-up plan of care for positive depression screening may be underdeveloped.

E. AAACN Ambulatory Care Nurse-Sensitive Adapted Proposed Measure: Ambulatory Care Nurse Patient Falls in the Institution

Introduction to the Measure

Falls are defined as a sudden, unintentional change in position causing an individual to land at a lower level, on an object, the floor, or the ground, other than as a consequence of a sudden onset of paralysis, epileptic seizure, or overwhelming external force. Unintentional falls are significant sources of morbidity and mortality, especially in people over 65 years of age for whom falls are the leading cause of accidental death (Rubenstein, 2006). Up to one third of falls are preventable. The Centers for Medicare & Medicaid Services (CMS) does not routinely reimburse hospitals for fall-related injuries. With a cost of up to \$36 billion and consequences such as fractures, internal bleeding, and death, patient falls are a significant safety concern across settings. A robust body of evidence targeting prevention and toolkits is available from such organizations as the Agency for Healthcare Research and Quality (AHRQ) and the American Nurses Association (ANA). Fall prevention requires targeted multidisciplinary vigilance, individualized for the patient and is a 2015 National Patient Safety Goal (The Joint Commission, 2015).

CALNOC Falls Measure (2014), Press Ganey NDNQI Falls Measure (Press Ganey, 2015a), and NQF 0141 (NQF, 2015f) are the respective names of currently endorsed measures related to falls in the institution.

AAACN NSI Task Force Recommendations for Adaptation to Ambulatory Care RN

The AAACN NSI Task Force and AAACN membership at large believe this measure is reflective of the institution as a whole and the related safety plan for entry from the outside environment to the point of care. When falls occur in the ambulatory care area (clinic, Emergency Department, surgery center, etc.) they should be considered a unit/clinic event, however, when falls occur outside of the clinic environment, they should be considered an environmental event. The registered nurse plays a role in surveying the clinic environment and identifying possible falls risk areas. Because of its widespread use in inpatient as well as in some outpatient settings, however, the following measure descriptions are provided as recommendations.

Measure Source: AAACN Adapted Proposed Measure: Ambulatory Care Nurse Patient Falls in the Institution

Adapted by the AAACN NSI Task Force as an ambulatory care nurse-sensitive measure from the following entities: CALNOC Falls Measure (2015), Press Ganey/NDNQI Falls Measure (Press Ganey, 2015a), and NQF 0141 (NQF, 2015f).

Measure Description: All documented falls, with or without injury, experienced by patients on eligible unit types in a calendar quarter. Reported as Total Falls per 1,000 Patient visits and Unassisted Falls per 1,000 Patient visits.

Numerator Statement: Total number of patient falls (with or without injury to the patient and whether or not assisted by a staff member) within the ambulatory care site of service per calendar month X 1,000.

Denominator Statement: Total number of patient visits in ambulatory care service site during the calendar month.

Methodology and Next Steps Recommendations

It is recommended that further validation of this measure and its applicability to the ambulatory care setting be conducted. Evidence that falls occur from a lack of nursing care in the ambulatory care setting needs to be collected and documented so that an emphasis on measuring this patient outcome in a setting that is not similar to the inpatient setting where falls are concerned may be disseminated.

F. AAACN Ambulatory Care Nurse-Sensitive Adapted Proposed Measure: Ambulatory Care Nurse Screening for Future Falls Risk

Introduction to the Measure

Falls are defined as a sudden, unintentional change in position causing an individual to land at a lower level, on an object, the floor, or the ground, other than as a consequence of a sudden onset of paralysis, epileptic seizure, or overwhelming external force. The standard for future falls risk screening is that all older persons who are under the care of a heath professional (or their caregivers) should be asked at least once a year about falls (American Geriatrics Society [AGS], British Geriatrics Society [BGS], & American Academy of Orthopaedic Surgeons Panel on Falls Prevention [AAOSPFP], 2001), Any older persons who present for medical attention because of a fall, report recurrent falls in the past year, or demonstrate abnormalities of gait and/or balance should receive a fall evaluation performed by a clinician with appropriate skills and experience. This screening may necessitate referral to a specialist (e.g., geriatrician) (AGS, BGS, & AAOSPFP, 2001). Older patients in contact with health care professionals should be asked routinely whether they have fallen in the past year and about the frequency, context, and characteristics of the falls (RTI International & Telligen, 2011a). Older people reporting a fall or considered at risk of falling should be observed for balance and gait deficits and considered for their ability to benefit from interventions to improve strength and balance (RTI International & Telligen, 2011a).

ACO 13 (RTI International & Telligen, 2011a), GPRO CARE 2 (CMS, 2014c), NQF #0101 (NQF, 2015g), and PQRS 154 (CMS, 2014d) are the respective names of currently endorsed measures related to screening for future falls risk.

AAACN NSI Task Force Recommendations for Adaptation to Ambulatory Care RN

Measure Source: AAACN Adapted Proposed Measure: Ambulatory Care Nurse Screening for Future Falls Risk

Adapted by the AAACN NSI Task Force as an ambulatory care nurse-sensitive measure from the following entities: ACO 13 (RTI International & Telligen, 2011a), GPRO CARE 2 (CMS, 2014c), NQF #0101 (NQF, 2015g), and PQRS 154 (CMS, 2014d).

Measure Description: Percentage of patients who were screened for future fall risk at least once within 12 months.

Numerator Statement: Patients who were screened for future fall risk at least once within 12 months.

- **Fall** Is defined as a sudden, unintentional change in position causing an individual to land at a lower level, on an object, the floor, or the ground, other than as a consequence of a sudden onset of paralysis, epileptic seizure, or overwhelming external force.
- **NOTE:** Patients are considered at risk for future falls if they have had two or more falls in the past year or any fall with injury in the past year.

Denominator Statement: All patients seen in a defined period of time in the ambulatory care setting.

Denominator Exclusions: (Exclusion only applied if patient was not screened for future fall risk) Documentation of medical reason(s) for not screening for future fall risk (e.g., patient is not ambulatory).

Methodology and Next Steps Recommendation

- Referral to home health nurse for evaluation of living conditions.
- Teaching of medication side effects.
- Referral for evaluation of muscle and body strength and endurance.
- Educate about falls risks and prevention.
- Review medication and make recommendations to provider for alteration of current regimens to promote enhanced safety where falls are concerned.
- Assess vision and refer to optimize vision.
- Assess for foot problems and refer as needed.

G. AAACN Ambulatory Care Nurse-Sensitive Adapted Proposed Measure: Ambulatory Care Nurse Screening for Body Mass Index (BMI)

Introduction to the Measure

Obesity is a public health concern in the United States and throughout the world. In the United States, obesity prevalence doubled among adults between 1980 and 2004 (Flegal, Carroll, Ogden, & Johnson, 2002; Ogden et al., 2006). Obesity is linked with an increased risk of a number of conditions, including diabetes mellitus, cardiovascular disease, hypertension, and certain cancers, as well as increased risk of disability and a modestly elevated risk of all-cause mortality (Adams et al., 2006; RTI International & Telligen, 2011b). Body Mass Index (BMI) is also related to an increased risk of death, particularly in adults younger than 65 years of age. Obesity has been shown to reduce life expectancy by 6 to 20 years depending on age and race. Ischemic heart disease, diabetes, cancer (especially liver, kidney, breast, endometrial, prostate and colon), and respiratory diseases are the leading causes of death in persons who are obese (Adams et al., 2006; RTI International & Telligen, 2011b).

BMI is expressed as weight/height and BMI parameters for adults are expressed as (RTI International & Telligen, 2011b):

- Overweight: BMI 25.0-29.9
- Obesity: BMI greater than or equal to 30.0
- Extreme Obesity: BMI Greater than or equal to 40

Thirty-one percent of children 2-19 years of age are overweight, defined as above the 85th percentile for BMI, and 16.9% are obese, defined as BMI above the 95th percentile (Ogden, Carroll, Curtin, Lamb, & Flegal, 2006). Diseases previously seen primarily in adults such as diabetes, hypertension, heart disease, hyperlipidemia, reproductive problems, asthma, and sleep disorders are now being observed in overweight children (Lobstein, Baur, & Uauy, 2004).

ACO #16 (RTI International & Telligen, 2011b), GPRO PREV-9 (CMS, 2014e) and NQF #0421 (NQF, 2014d) are the respective names of currently endorsed measures related to screening for future falls risk.

AAACN NSI Task Force Recommendations for Adaptation to Ambulatory Care RN

Measure Source: AAACN Adapted Proposed Measure: Ambulatory Care Nurse Screening for Body Mass Index (BMI)

Adapted by the AAACN NSI Task Force as an ambulatory care nurse-sensitive measure from the following entities: ACO #16 (RTI International & Telligen, 2011b), GPRO PREV-9 (CMS, 2014e), and NQF #0421 (NQF, 2014d).

Measure Description: Percentage of patients with a calculated BMI in the past six months or during the current visit documented in the medical record and if the most recent BMI is outside of normal parameters, a follow-up plan is documented within the past six months or during the current visit.

Numerator Statement: Patients, aged 2 years and older with BMI calculated within the past six months or during the current visit and a follow-up plan is documented within the last six months or during the current visit if the BMI is outside of normal parameters.

Denominator Statement: All patient visits during the defined measurement period.

Denominator Exclusions:

- Documentation of medical reason(s) for not having a BMI measurement performed during the measurement period (e.g., patient is receiving palliative care, patient is pregnant or patient is in an urgent or emergent medical situation where time is of the essence and to delay treatment would jeopardize the patient's health status).
- Documentation of patient reason(s) for not having a BMI measurement performed during the measurement period (e.g., patient refuses BMI measurement or if there is any other reason documented in the medical record by the provider explaining why BMI measurement was not appropriate).

Methodology and Next Steps Recommendations

- Provide education to patients and parents of pediatric patients regarding BMI measurement.
- Provide education/referral to a nutritionist to patients and parents of pediatric patients with a high BMI.
- Provide education to patients and parents of pediatric patients regarding need for weight control.
- Review diet and eating habits, along with exercise, during visits with patients.
- Assess adults for signs of diseases associated with elevated BMI.
- Assess need for laboratory studies (blood and urine) based on history of patient and when previous testing was performed.

H. AAACN Ambulatory Care Nurse-Sensitive Adapted Proposed Measure: Ambulatory Care Nurse RN Demographics

Introduction to the Measure

The following acute care nurse-sensitive indicators are thought to be conceptually important but require further development to apply to the ambulatory care setting. To encourage advancement of education and certification and improvement of nurse practice environment through initiatives that promote retention and reduced RN vacancy, close measurement of these metrics and associated implementation of performance improvement initiatives are required in the ambulatory care environment.

AAACN NSI Task Force Recommendations for Adaptation to Ambulatory Care RN

Measure Source: AAACN Adapted Proposed Measure: Ambulatory Care Nurse RN Demographics

Adapted from Collaborative Alliance for Nursing Outcomes (CALNOC) inpatient demographic measures (CALNOC, 2014).

1. <u>Diploma RN</u>

Measure Source: CALNOC

Measure Description: % RN Full-Time Equivalent (FTE) with a Diploma in

Nursing in specified ambulatory care setting

Numerator Statement: # RN FTE with a Diploma in Nursing specified

ambulatory care setting

Denominator Statement: Total # RN FTE in specified ambulatory care setting

2. Associate's Degree in Nursing RN

Measure Source: CALNOC

Measure Description: % RN FTE with ADN in specified ambulatory care setting **Numerator Statement:** # RN FTE with ADN in specified ambulatory care setting

Denominator Statement: Total # FTE in specified ambulatory care setting

3. Bachelor's of Science in Nursing RN

Measure Source: CALNOC

Measure Description: % RN FTE with BSN in specified ambulatory care setting **Numerator Statement:** # RN FTE with BSN in specified ambulatory care setting **Denominator Statement:** Total # FTE in specified ambulatory care setting

4. Master's of Science in Nursing RN

Measure Source: CALNOC

Measure Description: % RN FTE with MSN in specified ambulatory care setting **Numerator Statement:** # RN FTE with MSN in specified ambulatory care setting **Denominator Statement:** Total # FTE in specified ambulatory care setting

5. Doctorate in Nursing RN

Measure Source: CALNOC

Measure Description: % RN FTE with Doctorate in nursing in specified

ambulatory care setting

Numerator Statement: # RN FTE with Doctorate in nursing in specified

ambulatory care setting

Denominator Statement: Total # RN FTE in specified ambulatory care setting

6. Certification RN

Measure Source: CALNOC

Measure Description: % RN FTE with any approved nursing certification working in specified ambulatory care setting (approved certifications list per CALNOC code book)

Numerator Statement: # RN FTE with any approved nursing certification working in specified ambulatory care setting (approved certifications list per CALNOC code book)

Denominator Statement: Total # RN FTE in specified ambulatory care setting

7. Certification RN in Field of Specialty

Measure Source: CALNOC

Measure Description: % RN FTE with nursing certification in field of specialty working in specified ambulatory care setting (approved certifications list per CALNOC code book)

Numerator Statement: # RN FTE with nursing certification in field of specialty working in specified ambulatory care setting (approved certifications list per CALNOC code book)

Denominator Statement: Total # RN FTE in specified ambulatory care setting

8. <u>Turnover by Full-Time Equivalent</u>

Measure Source: CALNOC

Measure Description: % RN FTE turnover per CALNOC specified definition in

specified ambulatory care setting

Numerator Statement: # RN FTE that turned over within specified time period in

specified ambulatory care setting

Denominator Statement: Total # RN FTE in specified ambulatory care setting

9. <u>Vacancy by Full-Time Equivalent</u>

Measure Source: CALNOC

Measure Description: % RN FTE Vacancy per CALNOC specified definition in

specified ambulatory care setting

Numerator Statement: # RN FTE that are vacant per CALNOC specified

definition in specified ambulatory care setting

Denominator Statement: Total # RN FTE in specified ambulatory care setting

I. AAACN Ambulatory Care Nurse-Sensitive Adapted Proposed Measure: Ambulatory Care Nurse Patient Satisfaction

Introduction to the Measure

Satisfaction with the health care experience is linked to better outcomes for both patients and organizations and is a reportable metric required or recommended by such agencies as The Joint Commission, Agency for Healthcare Research and Quality, and Health Resources and Services Administration. In addition, the Patient Portability and Accountable Care Act (PPACA) linked Medicare payment to satisfaction rates. Generally, patient "satisfaction" is somewhat more subjective, focused on care received, while "patient experience" measures objective elements related to the overall interaction. In patient-centered medical home models of care, the CG-CAHPS is utilized to evaluate patient experience with accessible, coordinated, and patient-centered care (Burnet et al., 2014).

Very few patient satisfaction surveys for ambulatory care nursing were identified in the literature. The majority of articles focused on the advanced practice nursing role or target a specific area such as oncology, long-term care, adult day care, palliative care, or surgical centers. For example, the HCAHPS ambulatory survey directions state nurse practitioner can be substituted for "provider." Communication, medications, and technical/professional skills of nurses were mentioned in some articles that included survey questions about nurses. However, questions are not specific enough to evaluate the registered nurse (RN) professional role in facilitating patient satisfaction/experience in ambulatory care. An outpatient survey available from the Human Resources and Services Administration asks whether or not nurses and medical assistants were friendly and helpful (Michael, Schaffer, Egan, Little, & Pritchard, 2013). The National Research Corporation's Picker Patient Satisfaction tool offers nurse-specific questions, but in the ambulatory setting does not yet have benchmarking capability.

The AAACN NSI Task Force would like to pursue at least 1-2 questions that are applicable if an RN cared for the patient. We are aware that many settings do not have an RN and that if an RN is employed in that facility, patients often do not know the difference between RNs, medical assistants (MAs), and other staff.

Measure Source: AAACN Adapted Proposed Measure: Ambulatory Care Nurse Patient Satisfaction

Adapted by AAACN NSI Task Force from the following entities: Press Ganey (Press Ganey, 2015b), CMS (CMS, 2015g), CG-CAHPS (AHRQ, 2015), PSQ-18 (RAND, 2015), and Risser Patient Satisfaction Survey (Risser, 1975), PPE-15 (Picker Survey) (Jenkinson, Coulter, & Bruster, 2002).

Measure Description: The percentage of surveys that were returned within a defined period and of questions that are nursing specific and answered by the patient as "ALWAYS".

Numerator Statement: The number of patient surveys that were returned within a defined period that answered "ALWAYS" to defined nurse-sensitive questions.

Denominator Statement: The total number of surveys received back within a defined period of time.

AAACN NSI Task Force Recommendations for Adaptation to Ambulatory Care RN

Methodology and Next Steps Recommendations

Overall outpatient satisfaction involves many factors outside the nurse's control. Some ambulatory care settings across the country focus patient satisfaction performance improvement initiatives related to the RN on the following:

- Access to care: Family is able to get an appointment in a time they feel is reasonable, usually in 3-14 days.
- Moving through the visit: How the patient perceives the speed in which they are seen in the clinic and whether or not that meets their expectation.
- Response time to phone calls: The patient usually wants their call returned before the end of the day.
- Nurse assessment of pain and follow through.
- Nurse assessment of safety and preparation for home care.

SECTION 4 AAACN NSI Task Force Proposed Indicators for Development and Endorsement: Original Measures Recommended as Ambulatory Care Nurse-Sensitive Indicator Measures

- A. Ambulatory Care Nurse Care Coordination: Appropriate Referral
- B. Ambulatory Care Nurse Patient Engagement: Measurement of Patient Activation
- C. Ambulatory Care Nurse Administration of Vaccine Per Non-Patient Specific Protocol
- D. Ambulatory Care Nurse Clean Urine Specimen

SECTION 4

AAACN NSI TASK FORCE PROPOSED INDICATORS FOR DEVELOPMENT AND ENDORSEMENT: ORIGINAL MEASURES RECOMMENDED AS AMBULATORY CARE NURSE-SENSITIVE INDICATOR MEASURES

The AAACN NSI Task Force identifies and describes four new measures recommended for development, endorsement, and pilot testing. These four measures have the potential to uniquely reflect the role of the registered nurse (RN) in the ambulatory care setting as well as promote enhanced nursing practice. Each measure represents a cumulative effort of evidence and best practice review.

A. AAACN Ambulatory Care Nurse-Sensitive Proposed Original Measure: Ambulatory Care Nurse Care Coordination: Appropriate Referral

Introduction of New Proposed Indicator

The American Nurses Association (ANA) along with the American Academy of Ambulatory Care Nursing (AAACN) and other professional organizations recognize telehealth nursing, including telephone triage, as a sub-specialty within the larger specialty of ambulatory care nursing. It is supported by a unique body of knowledge outlined in the AAACN *Scope and Standards for Professional Telehealth Nursing* (2011). In recent years there has been an increase in the use of telephone triage. This growth has been international and includes systems in Australia, Netherlands, Canada, New Zealand, Denmark, Sweden, Dominican Republic, United Kingdom, Ecuador, Venezuela, and Greece.

Callers access the telephone triage system for advice and direction related to health care concerns. This is different than the Emergency Management System (EMS), which is accessed by callers who are in life-threatening or emergency situations. Although a caller to a telephone triage program may be directed to call EMS (911), industry standards report this occurs in approximately 1% or less of calls.

Many telephone triage calls are managed after hours (also referred to as out of hours) through centralized call centers using evidence-based guidelines, or protocols, embedded into clinical decision support software. Although the exact number of nurse-led telephone triage calls done annually in the US is not known, Sabin (1998) states that as of 2001, 100 million people had access to telephone triage advice and that growth of 25% per year is expected.

Although telephone triage is generally accepted to be a safe method of providing care (Bunn, Byrne, & Kendall, 2004), studies continue to surface that question the safety of telephone triage. A systematic review by Huibers, Smits, Renaud, Giesen, & Wensing

(2011) concluded that on average 10% of telephone triage contacts were unsafe, although these findings were not associated with any serious safety events. This is a worrisome estimate, as the number of telephone triage encounters annually is estimated in the tens of millions. The authors found many of the studies evaluating safety were old (greater than 10 years). They suggest new studies on the safety of telephone triage are needed to inform practice and health policy decisions.

Studies of telephone triage often focus on under referral, failing to recognize a medical emergency or signs and symptoms of significant illness, or over referral, determining the patient requires a higher level of service than necessary. Under referral may jeopardize patient safety, whereas over referral places increased stress on an overburdened emergency care system and results in unnecessary costs. Few, if any, studies have attempted a comprehensive evaluation of nurse triage dispositions to evaluate the rate of "appropriate disposition." Studies on telephone triage lack consensus on what level of intervention constitutes an appropriate referral. There is even disagreement on a definition for "appropriate."

Appropriate referral (AR) is defined as directing the right patient to the right level of service at the right time. This decision is made by evaluating the patient at the time of the call and assessing the severity of symptoms. The nurse assigns a level of urgency or an acuity level based on assessment of symptoms and assigns a triage disposition. Although there is no universal guide for triage dispositions, in general they fall into the following categories:

Table 1. Triage Disposition Categories

911	Urgent
See in Emergency Department/Urgent Care /Office Immediately	Urgent
See in Emergency Department/Urgent Care /Office in 4 hours	Non-urgent
See within 24 hours	Non-urgent
See in greater than 24 hours	Non-urgent

Source: Agency for Healthcare Research and Quality, 2014.

Many call centers and triage programs perform rigorous quality assurance to ensure nurses are complying with established evidence-based guidelines and sending patients to the appropriate level of care. There are also industry standards specific to certain software and patient populations that benchmark referral rates for the triage dispositions listed above. The problem is that these are internal measures and as a result subject to bias.

In contrast, nurse-sensitive indicators (NSI) are objective, consistent, and easily benchmarked across all settings. The goal is to determine a measure that is objective, can be extracted from the patient electronic medical record (EMR) and can be compared across multiple settings and patient populations. Ideally, an NSI in telephone triage would compare the patient disposition assigned by the triage nurse to the findings

of the provider who first examines the patient during the telephone encounter. Since patient status is rarely static, we hypothesize that the sooner the patient is seen following telephone triage, the more likely that the two assessments will align. In order to limit bias, it is preferable that the patient's status upon arrival at the appropriate facility be determined without prior knowledge of the assigned triage disposition level.

A universal patient assessment tool used to assign acuity in emergency departments (ED) is the emergency severity index (ESI) (Gilboy, Tanabe, Travers, & Rosenau, 2011). The ESI is a five-level system (See Table 2) that can be used with adults and children to accurately assign a level of acuity and urgency to patients in the ED by predicting which patients require admission, have an increased length of stay (LOS) in the ED, or require a higher number of resources to arrive at a final disposition (Green et al., 2012).

Table 2. Emergency Severity Index

Level 1: Patient requires immediate life saving interventions	High acuity
Level 2: Patient is unstable or in severe distress	High acuity
Level 3: Resources required: many	High acuity
Level 4: Resources required: one	Low acuity
Level 5: Resources required: none	Low acuity

Source: Gilboy et al., 2011.

Consequently, the question is posed: for patients who contact a nurse-led telephone triage line and are referred to be seen within 24 hours, is it possible to compare the telephone triage disposition with the ESI to determine an appropriate referral rate (ARR)?

Since it has already been established that the intent of telephone triage programs differ from EMS programs such as 911 (accidents, sudden cardiac arrest, SIDS), Level 1 of ESI ("is the patient dying?") would seldom be expected to map to a telephone triage disposition. Additionally, since ESI is not influenced by patients' mode of arrival, the presence or absence of emergency transport should not impact ESI score. Therefore, we would anticipate that the following telephone triage dispositions and ESI would align.

Table 3. Proposed Appropriate Referral Measure

Telephone Triage Disposition	ESI	Acuity
Does not map/	1	High
Do not routinely receive this level of call		
911	2	High
See in ED immediately	3	High
See in 4 hours	4	Low
See in 24 hours	5	Low

Measure Source: AAACN New Proposed Measure: Ambulatory Care Nurse Care Coordination: Appropriate Referral Rate

Measure Description: The percentage of patients whose acuity measurement on triage disposition is equal or within 1 point of the ESI acuity measurement given of that same patient in the Emergency Department on arrival.

• Triage disposition determined by nurse during telephone encounter. ESI assigned by triage nurse in Emergency Department (ED).

Numerator Statement: Number of calls where the triage disposition is equal to or within 1 (higher or lower) of the ESI.

Denominator Statement: All patients who receive a triage disposition from a call center nurse and are then seen in ED.

Exclusion: Patients who receive a triage disposition but are not seen in a facility where the EMR can be accessed.

Inclusion: Patients who call a telephone nurse triage center after hours and are directed to an ED with a disposition of 911, See in ED/urgent care Immediately, See in ED in 4 hours, See in 24 hours.

- We propose the measure is limited to After Hour Call Centers with access to:
 - Triage disposition data
 - Patient EMR information

Methodology and Next Steps Recommendations

- Expand trial to include a statistically significant *n* and publish data.
- An initial pilot was completed that compared triage disposition with ESI for 25 pediatric after hours calls, and 24 of the 25 reviewed were equal to or within one of the ESI. Considering that patient conditions can vary considerably in a 24-hour period, this seemed like a reasonable result.
- This measure has limitations. Currently, there is little in place in EMR that captures nurse actions during daytime triage in an office or clinic or care coordination efforts in a clinic. Therefore, this measure would be limited to areas that have access to disposition data from telephone triage phone calls and access to EMR of patients who present in the ED within 24 hours. Most large Call Centers and After Hours Programs utilize software that provides this information. Many are associated with large hospital networks and can access patient visit information.

B. AAACN Ambulatory Care Nurse-Sensitive Proposed Original Measure:
Ambulatory Care Nurse Patient Engagement: Measurement of Patient
Activation

Introduction of New Proposed Indicator

Background

Patient engagement is an increasingly important part of a national strategy to improve health outcomes and enhance health care quality. Recognized as a component of "accountable" health care, patient engagement was conceptualized from a recognized need for reform in the United States health care delivery system to include a more patient-centered approach capable of recognizing and responding to patient and family needs and preferences (Carman et al., 2013; IOM, 2001).

Patient engagement encompasses active participation by individuals (i.e., consumers, patients, and family members) in their health and health care. Examples of patient engagement behaviors include using electronic health portals for two-way communication with health care providers, setting wellness goals for weight management and physical activity with health care team members, or discussing advanced directives such as health care proxy with family members and actively communicating end-of-life care decisions to a primary care provider. Despite a multitude of patient engagement definitions, approaches, and evolving strategies, evidence supports that when patients are engaged and actively involved in their health care, it can bring about measurable improvements in health outcomes, safety, and quality of care (Hibbard & Greene, 2013; Maurer, Dardess, Carman, Frazier, & Smeeding, 2012).

Current Patient Engagement Strategies

Numerous national organizations such as The Joint Commission (TJC), Agency for Healthcare Research and Quality (AHRQ), Patient-Centered Outcomes Research Institute (PCORI), American Institutes for Research (AIR), American Hospital Association (AHA), and the Health Research and Educational Trust (HRET) have embraced and advocated for patient engagement. Consequently, several strategies have emerged to address how health care providers, hospitals, and health systems can engage patients and consumers at the individual, health care team, organizational, and community levels. Most recently, AIR (2014) identified eight strategies for change around patient engagement. These included: (1) patient and family preparation; (2) clinician and leadership preparation; (3) care and system redesign; (4) organization partnership; (5) measurement and research; (6) transparency and accountability; (7) legislation and regulation; and (8) partnership in public policy. Patient and family engagement resources and tools have also been developed to facilitate national

widespread adoption (AHA, 2013; HRET, 2013; Scholle, Torda, Peikes, Han, & Genevro, 2010).

These efforts are augmented by strategic, planned individual/organizational partnerships and research endeavors to increase patient engagement resources and to provide evidence to health care providers, hospitals, and health systems aimed toward improved health outcomes and reduced harm in health care (AHA, 2013; AIR, 2014; HRET, 2013).

The Nursing Alliance for Quality Care (NAQC), currently under the auspices of the American Nurses Association (ANA), defines patient engagement as "the involvement in their care by individuals (and others they designate on their own behalf), with the goal that they make competent, well-informed decisions about their health and health care and take action to support those decisions" (Sofaer & Schumann, 2013, p. 5). Their framework describes the development and outcomes of patient engagement as being closely related to patient activation.

Patient engagement was more recently defined by Maurer et al. (2012, p.10) as "a set of behaviors by patients, family members, and health professionals and a set of organizational policies and procedures that foster both the inclusion of patients and family members as active members of the health care team and collaborative partnerships with providers and provider organizations."

The multidimensional framework described by Carman et al. (2013) advances patient engagement concepts, recognizes its structural intricacies, and illustrates the various levels at which it is realized. This conceptual model serves as a roadmap for further patient engagement program development across every level and establishes a foundation for a broad-based patient engagement research agenda.

Patient Engagement and Activation

Patient activation is closely related to patient engagement. Activation encompasses the degree to which patients are motivated and possess the knowledge, skills and confidence to effectively manage their health (Bandura, 2004; Hibbard et al., 2008; Hibbard, Mahoney, Stockard & Tusler, 2005; Hibbard, Stockard, Mahoney, & Tusler, 2004). Activation includes the underlying construct of self-efficacy and is closely associated with the degree of individual engagement in health management.

Patient activation can be measured using the quantitative tool, PAM (Patient Activation Measure). The tool, developed by Judith Hibbard and currently licensed by Insignia Health, is a well-researched, reliable, validated instrument that has demonstrated applicability across different languages, cultures, demographic groups, and populations of people with varying health status (Hibbard & Mahoney, 2010; Hibbard et al., 2005; Hibbard, Mahoney, Stock, & Tusler, 2006; Hibbard, et al., 2004).

The PAM is a one-dimensional, probabilistic Guttman-like scale that reflects a developmental model of activation that appears to involve four levels: (1) believing the patient role is important; (2) having the confidence and knowledge necessary to take action; (3) actually taking action to maintain and improve one's health; and (4) staying the course even under stress. An individual's PAM score and respective activation level is determined through administration of a standardized PAM tool. The 13-question PAM is highly predictive of varied health behaviors and can used to tailor health care interventions and assess changes (Hibbard et al., 2004).

Patients who score high on the PAM and are determined to have a higher level of activation can be managed with supporting education and counseling. Patients who score low on the PAM and have a high disease burden, such as with chronic diseases, would be matched with a registered nurse and provider in a team-based care model (Hibbard & Greene, 2013). Evidence consistently demonstrates patients with higher levels of activation are more likely to engage in positive self-management behaviors and have improved outcomes (Hibbard, 2009; Hibbard & Greene, 2013; Hibbard, Greene, & Overton, 2013; Hibbard et al., 2008; Hibbard et al., 2006; Marshall et al., 2013; Sacks, Greene, Hibbard, & Overton, 2014; Skolasky et al., 2011).

John Wasson, professor at Dartmouth, and colleagues created a tool for patients to self-assess their confidence in managing and understanding their overall health and disease (Wasson & Coleman, 2014). The instrument is a simplistic visual scale, does not require answering multiple questions, and is presented in a familiar red, yellow, and green scale complemented by smiling and frowning faces at the end points. A single question prompts the reader to answer what it would take for them to become more confident in their self-care abilities. This tool is not presented as a robust, predictive tool like the PAM, but as a patient-reported outcome. It is intended to open the conversation to patient engagement strategies. Patients are invited to complete a more comprehensive health profile on a free web site (www.howsyourhealth.org). The self-assessment can be completed in advance of a healthcare appointment, shared with providers, and used to guide patients (Chase, 2011).

The National Quality Forum's (NQF) guidance in *Patient Reported Outcomes (PROs) in Performance Measurement* offers principles and a defined approach to the relationship among structure, process, and outcome in health care. PROs are defined as "any report of the status of the patient's (or person's) health condition, health behavior, or experience with health care that comes directly from the patient, without interpretation of the patient's response by a clinician or anyone else" (NQF, 2013c, abstract). Key domains include health-related quality of life (including quality of life), symptoms and symptom burden, experience with care, and health behaviors (NQF, 2013c).

The Role of Nursing in Patient Engagement

Nursing's role is fundamental in advancing evidence-based practice. Ambulatory care nurses and advanced practice primary care nurses have already begun to incorporate engagement strategies into care systems and will be instrumental in furthering the effort (Haas & Swan, 2014; Rutherford, 2014). Such strategies include use of motivational interviewing and creation of patient self-determined goals. The ambulatory care registered professional nurse and advanced practice nurses are uniquely positioned to further patient engagement interventions and program development.

Patient engagement underpinnings are detailed by Sofaer and Schumann (2013), authors of the NAQC white paper *Fostering Successful Patient and Family Engagement: Nursing's Critical Role.* This includes the development of patient engagement strategies, associated behaviors, outcomes, and the underlying orientation to patient activation. Engagement strategies include chronic disease self-management, motivational interviewing, family rounding, and health coaching, and are designed to promote and foster self-efficacy and confidence over time (See Figure 1). These strategies should result in engagement behaviors and ultimately effect changes in health behaviors and improved outcomes such as functional status and quality of life (Sofaer & Schumann, 2013). The NAQC logic model (See Figure 2) for maximizing the contributions of nursing to patient engagement includes strategy domains, changes in awareness, and behaviors of nurses, as well as proximal, intermediate, and longer-term outcomes (Sofaer & Schuman, 2013) (See Figure 3).

Figure 1. Guiding Principles for Patient Engagement

Patient engagement is a critical cornerstone of patient safety and quality. NAQC has grounded its approach to this topic by recognizing the primary importance of *relationships* between engaged patients and families and their clinicians, including but not limited to nurses. The following are principal assumptions that guide NAQC in addressing care that is patient centered.

- 1 There must be an active partnership among patients, their families, and the providers of their health care.
- 2. Patients are the best and ultimate source of information about their health status and retain the right to make their own decisions about care.
- 3. In this relationship, there are shared responsibilities and accountabilities among the patient, the family, and clinicians that make it effective.
- 4. While embracing partnerships, clinicians must nevertheless respect the boundaries of privacy, competent decision-making, and ethical behavior in all their encounters and transactions with patients and families. These boundaries protect recipients as well as providers of care. This relationship is grounded in confidentiality, where the patient defines the scope of the confidentiality.
- 5. This relationship is grounded in an appreciation of patient's rights and expands on the rights to include mutuality. Mutuality includes sharing of information, creation of consensus, and shared decision-making.
- 6. Clinicians must recognize that the extent to which patients and family members are able to engage or choose to engage may vary greatly based on individual circumstances, cultural beliefs and other factors.
- Advocacy for patients who are unable to participate fully is a fundamental nursing role. Patient advocacy is the demonstration of how all of the components of the relationship fit together.
- 8. Acknowledgment and appreciation of culturally, racially or ethnically diverse backgrounds is an essential part of the engagement process.
- 9. Health care literacy and linguistically appropriate interactions are essential for patient, family, and clinicians to understand the components of patient engagement. Providers must maintain awareness of the language needs and health care literacy level of the patient and family and respond accordingly.

Source: Sofaer & Schumann, 2013. Used with permission.

Figure 2.

Maximizing the Contributions of Nurses to Patient Engagement: A Logic Model

Domains of Strategy	Changes in Awareness of Nurses	Changes in Behaviors Among Nurses	Outcomes
Ensuring that all nursing education emphasizes patient engagement. Amplifying the professional standing of nurses as champions of patient engagement. Strengthening support for nurses as advocates in the care environment of patients. Aligning incentives to encourage patient engagement. Elevating regulatory expectations and standards that support patient engagement principles in practice. Intensifying efforts to conduct and disseminate research on patient engagement.	All nurses respect patient/ consumer place at the center of health care decision-making and in accordance with principles of patient centered care. All nurses embrace the belief that patients and families are or can become competent to engage fully in making informed health care decisions. All nurses recognize their own important role in supporting patients who are encountering obstacles in the health care system. Nurses at front lines: Recognize that patient goals, values, preferences, cultural context, and particular circumstances must be incorporated into the care plan. Recognize that when the patient perspective is not being honored, they will need to be active as an advocate, preferably with rather than for the patient/ family. Nurses in managerial roles: Recognize the impact of organizational structure and culture on the likelihood that engagement will be easy and well supported. Recognize the role of rewards and incentives in shaping an organization moving in the direction of fuller engagement. Nurses in executive/policy roles: Recognize the relationship of patient engagement to achieving Goals of National Quality Strategy. Recognize that they may need to lead transformation to achieve full engagement.	 At the front lines: Listening. Speaking in the language of the patient/family (plain English, Spanish, etc.). Using motivational interviewing and other methods of eliciting patient goals/values. Seeing the world from the "shoes" of the patient. Using key engagement strategies such as bedside and family rounds, bedside change of shift, shared decision making. Incorporating patients into advisory committees and QI teams. Ensuring patient friendliness of information and technology; facilitating their use by patients. Ensuring authenticity of informed consent. Advocating effectively whenever necessary for the patient and family. In managerial, executive, and policy roles: Becoming a "champion" for patient engagement and shared decision making. Building small "p" political support for engagement, including use of "business cases." Analyzing barriers and facilitators of engagement from the nursing and patient perspectives. Advocating for needed reforms and research. 	 Proximate Increases in scores of nurses on the Clinical Support for Patient Activation Measure. Higher nurse work satisfaction, better assessments of work environment, nurse retention. Higher scores on Patient Activation Measure. Higher prevalence of CFAH "engagement behaviors" among patients/family. More self-management of chronic conditions. Greater use of evidence in treatment decision making. Intermediate Improved patient experiences of care scores. Increased professional standing of nursing. Supportive care environments where nurses advocate for patients. Longer Term Improved quality and safety. Decreased/Eliminated disparities in care and health across age, gender, race, ethnicity, etc. Reduced overuse of unneeded tests and treatments. Increased use of evidence-based services. Lower costs. Improved population health.

Source: Sofaer & Schumann, 2013. Used with permission.

Figure 3.

Development and Outcomes of Patient Engagement

Engagement Strategies	Underlying Orientation to Engagement	Engagement Behaviors	Outcomes
Tailoring treatment plans to patient's level of activation	Measured by the Patient Activation Measure (PAM)	Frameworks for observing behaviors	Absence of harm as a result of care received
		preventive health strategies Self-management of	
		episodic illness	

Source: Sofaer & Schumann, 2013. Used with permission.

Patient Engagement Measurement and Next Steps

Patient engagement measurement is in early stages of development and exploration of engagement measures should be considered at multiple levels (i.e., policy, organizational, systems, individual). Sufficient attention must be given to quantitative, qualitative, and mixed-method measurement approaches.

Patient engagement measures in ambulatory care nursing may reflect team-based approaches, particularly in primary care, and care and attention must be given to teasing out nursing's influence. As the scope of ambulatory care nursing continues to evolve amidst emerging team-based care models and electronic health record interoperability, opportunities may exist to define "nursing-centric" patient engagement measures. More studies are needed to determine the influence of the nurse on patient engagement and patient activation.

Measure Source: AAACN New Proposed Measure: Ambulatory Care Nurse Patient Engagement: Measurement of Patient Activation

Measure Description: Change in patient activation measurement in patients with chronic disease following education and coaching by a registered nurse.

• Use of the PAM tool to assess and document the patient's level or stage of engagement and readiness to manage their health.

Numerator Description: Number of patients whose PAM score moved one or more stages higher from baseline to repeat score at 6 months.

Denominator Description: Number of patients who received chronic disease management education and coaching from an RN at least once during the measurement period.

- Stage 1: Does not yet understand that an active role is important (score <=47).
- Stage 2: Lacks knowledge and confidence to take action (>=47.1 to <=55.1).
- Stage 3: Beginning to take action (>=55.2 to <=67.0).
- Stage 4: Maintains positive health behaviors over time (>=67.1) (Mosen et al., 2007).

Measure Exclusions:

- Patients under 18 years of age.
- Patients with a baseline PAM of Stage 4.

Methodology and Next Steps Recommendations

- Determine baseline by administering PAM to patients with one or more chronic diseases.
- The RN provides education and coaching using motivational interviewing techniques and supports the patient in creating self-determined goals.
- The PAM is re-administered to the patient at 6 months.

References: See Section 7.

C. AAACN Ambulatory Care Nurse-Sensitive Proposed Original Measure:
Ambulatory Care Nurse Administration of Vaccine Per Non-Patient
Specific Protocol

Introduction of New Proposed Indicator

In order to prevent infectious diseases and their sequelae, immunizations are recommended throughout life (Centers for Disease Control and Prevention [CDC], 2012). For most routinely recommended vaccines, adult coverage remains low (CDC, 2012). Vaccination against influenza and pneumococcus should occur annually for specific age groups (McLaughlin, McGinnis, Tan, Mercatante & Fortuna, 2015). The shingles vaccine should be provided to adults aged 60 years and older, and the Tdap booster should be administered to all adults once (McLaughlin et al., 2015). Some vaccines can be administered via standing health care personnel (where permitted by state) to assess the immunization status of a patient and administer the vaccine according to a protocol (Office of Disease Prevention and Health Promotion, 2015b). A systematic review performed in 1997 and completed in 2009 found strong evidence for effectiveness for standing orders in adults and children (Guide to Community Preventative Services, 2015).

The law and regulations on the administration of non-patient specific orders for certain immunizations, anti-anaphylactic agents, and PPD mantoux skin tests standardize practice across all service delivery systems. Registered nurses may administer certain vaccines to patients without the use of a patient-specific order to improve the health of the population as identified in the applicable state's Nurse Practice Act. Examples of this include influenza and pneumococcal vaccination.

Measure Source: AAACN New Proposed Measure: Ambulatory Care Nurse Administration of Vaccine Per Non-Patient Specific Protocol

Measure Description: Percentage of patients eligible for influenza or pneumococcal vaccination under non-patient specific protocols for RNs who receive the vaccination by the RN in the specified measure period.

Numerator Statement: # of patients who receive the influenza or pneumococcal vaccination by the RN under non-patient specific protocol in the specified measure period.

Denominator Statement: # of patients seen by the RN who are eligible to receive the influenza or pneumococcal vaccine under non-patient specific protocol in the specified measurement period.

Inclusions: Only patients who are seen by the RN.

- Influenza vaccine: patients aged 2 years and older.
- Pneumococcal vaccine: Patients aged 65 years and older.

Exclusions: Patients not seen, screened, or assessed by a RN.

Methodology and Next Steps Recommendations

As this measure is developed, seek to replicate with immunizations given throughout the lifespan such as older adult immunizations and childhood immunizations.

References: See Section 7.

D. AAACN Ambulatory Care Nurse-Sensitive Proposed Original Measure: Ambulatory Care Nurse Clean Urine Specimen

Introduction of New Proposed Indicator

A urinary tract infection (UTI) is one of the most common patient infections (Shrestha, Gyawali, Gurung, Amatya, & Kumar-Bhattacharaya, 2013). Urine cultures are the gold standard for diagnosing UTIs (Gibly, 1998). Improper collection, however, may lead to contaminated specimens (Shrestha et al., 2013). Patients who are provided with instructions regarding the proper cleaning procedures prior to obtaining a clean urine specimen have a lower contamination rate, thus reducing the need to repeat the sample (Bekeris, Jones, Walsh, & Wagar, 2008). Urinalysis is necessary for outpatients who have recurrent UTIs, experience treatment failures, or have more complicated UTIs (Wilson & Gaido, 2004).

It is estimated that UTIs represent approximately 7 million visits to outpatient clinics (Wilson & Gaido, 2004). Correct processing and handling of urine specimens is dependent on the method of collection and as such represents an important practice improvement for clinicians (Wilson & Gaido, 2004). Urine cultures are one of the most common point of care tests performed by nurses in the ambulatory care setting and as such, this is an activity with potential to reflect quality of nursing care in that setting.

Measure Source: AAACN New Proposed Measure: Ambulatory Care Nurse Clean Urine Specimen

Measure Description: % of urine specimens that are contaminated.

Numerator Statement: # of contaminated samples per specified time period.

Exclusions: Non-toilet trained children, patients with Foley catheters or other urinary diversion devices.

Denominator Statement: # of total urine samples sent per specified time period.

Methodology and Next Steps Recommendations

This is a measure that can be pilot tested in the ambulatory care setting that has a high incidence of contaminated specimens. Patients, when provided with the proper education regarding the collection of a specimen, will be better able to collect the specimen without contamination. Additional considerations are:

No standard for cleaning the genitourinary area exists.

 Variability will exist with this dependent on patient self-collection of specimen. The quality of patient education performed by the RN may be reflected as a process component of this measure, and seen in comparison groups based on the quality of specimens.

References: See Section 7.

SECTION 5 Current Exemplars in Ambulatory Care Nurse-Sensitive Indicator Measurement

- A. Care Coordination: RN Care Manager Reduction of Readmissions
- B. Care Coordination: Population Health Management Through Advanced Care Planning of High Risk Populations
- C. Care Coordination: Decreasing Pediatric Emergency Room Visits
- D. Wound Care Healing Rate Through RN Case Managed Cohorts of Patients
- E. Opioid Use Monitoring and Safety at Pain Center
- F. Reduction of Readmissions Via a Follow-Up Phone Call Process
- G. Team BMI: Management of Pediatric Overweight and Obesity in Pediatric Primary Care

SECTION 5 CURRENT EXEMPLARS IN AMBULATORY CARE NURSE-SENSITIVE INDICATOR MEASUREMENT

Many innovative processes are being developed to elicit positive outcomes for patients in the ambulatory care setting. In many cases, the only comparative benchmarks are either current literature or institutional goals. However, as listed in a few of the below examples, some organizations are finding unique ways to benchmark quality that are nurse sensitive and create better outcomes for patients. Many of the members of the AAACN NSI Task Force joined the task force because they were starting to pilot various nurse-sensitive processes in their home organizations.

A. Care Coordination: RN Care Manager Reduction of Readmissions

Measure Source: Sentara Medical Group, Sentara Health System

Measure Description: RN Care Managers in the Medical Group follow every medical discharge for at least 30 days. Percentage of patient medical readmissions in every RN Care Manager's cohort of patients.

Numerator Statement: # Medical Group patient medical readmissions (YTD).

Denominator Statement: Total number of medical group discharges (YTD).

B. Care Coordination: Population Health Management Through Advanced Care Planning of High Risk Populations

Measure Source: Sentara Medical Group, Sentara Health System

Measure Description: Advanced Care Planning in High Risk Patient Populations: % of patients in a high risk patient population with a documented facilitated Advanced Care Plan. Advanced Care Plans are assigned, facilitated, and documented by specialized RN Care Managers trained in Advanced Care Planning of complex populations.

Numerator Statement: # patients with a documented, facilitated Advanced Care Plan assigned to an Advanced Care Planning (ACP) RN Care Manager in Sentara Medical Group.

Denominator Statement: # patients in a given high risk population assigned to an RN Care Manager.

C. Care Coordination: Decreasing Pediatric Emergency Room Visits

Measure Source: Medical University of South Carolina. Hospital billing data, Medicare/Medicaid billing data. This data will capture patients admitted to ERs around the state. Data can be tracked by MRN/DX and for children can be associated with point of care.

Measure Description: # of ambulatory care patients with chronic health condition (asthma, epilepsy, diabetes, etc.) seen in clinic or doctor's office with documentation of specific education/care coordination activity provided by registered nurse.

Numerator Statement: # of patients seen in clinic/office with specified diagnosis and subsequently seen in the emergency department.

Denominator Statement: Total # of patients seen with specified diagnosis matching that of the numerator group, with documentation of specific education or care coordination activity provided by registered nurse.

Exclusions: Patients who received an after-visit summary AVS or teaching sheet, but not specific RN education. Patients who received visit instruction by the provider.

Comments:

- Seizure patient intervention/RN instruction: Asthma action plan and magnet for first aid for seizure.
- Baclofen pump care coordination/RN intervention: Instructions on pump refill timeline, giving family the medication dose and rate along with date pump will be empty. Tracking by RN staff for on-time visits and intense follow up for no-show appointments.
- Asthma patient intervention/RN instruction: Asthma action plan.
- Synagsis patient intervention/RN instruction and care management: Synagsis education by video, magnet given with appointment dates and dose due. EMR flagged so if patient is seen in another clinic there are no missed opportunities and dose can be given with an extra visit. Tracking by RN staff for on-time visits and intense follow up for no-show appointments.

D. Wound Care Center Healing Rate Through RN Case Managed Cohorts of Patients

Measure Source: Rush Oak Park Hospital Wound Care Clinic

Measure Description: % of patients with complete wound healing after 14 weeks, once weekly visits to RN case managed wound care clinic.

Numerator Statement: # patients with complete wound healing after 14 weeks, once weekly visits to RN case managed wound care clinic.

Denominator Statement: # all patients seen in RN case managed wound care clinic.

Exclusions: Patients that choose to abort treatment before 14 weeks.

Rationale: Nurse-sensitive related to role of nurse as case manager and care coordinator.

E. Opioid Use Monitoring and Safety at Pain Center

Overview of Exemplar:

- The organizational exemplar is from the number of calls for opioid prescription refills and the number of clinic appointments for opioid prescription refills.
- A significant number of patients seen at the clinic are on long-term opioids. Based on the July 2013 FDA approval of a risk evaluation and mitigation strategy (REMS) for extended-release (ER) and long-acting (LA) opioid medications, the clinic changed prescribing practices. Prior to implementation, the mean number of calls to nurses was 16.85 calls per day. Post implementation the mean volume of calls decreased to 2.23, an 86 percent reduction. As the number of phone calls decreased, the number of clinic visits increased by 26 percent. As a result of counseling and education from nursing, many patients were able to reduce their opioid usage.
- The organization utilized a data collection tool to track the number of phone calls and the electronic health record for clinical and appointment related information.
- The organization presented this event to all staff as a quality improvement initiative as well as presented the information to the Midwest Pain Society and the American Academy of Physical Medicine and Rehabilitation General Assembly.

Measure Source: Rehabilitation Institution of Chicago

Measure Description: Number of phone calls per day was monitored from December 2012 through May 2013. Number of office visits was monitored during the same period.

Numerator Statement: Number of phone calls and number of clinic appointments.

Denominator Statement: Total number of patient visits for a defined period of time.

Exclusions: Non-opioid patients.

F. Reduction of Readmissions Via a Follow-Up Phone Call Process

Measure Source: University of California, Davis Health System

Measure Description: Percentage of patients that received a follow-up phone call within 3 days and the percentage of patients in that same cohort who were readmitted to the hospital within 30 days.

Numerator Statement: Number of patients that received a follow-up phone call within 3 days and the number of patients in that same cohort who were readmitted to the hospital within 30 days.

Denominator Statement: Total number of patients discharged from the hospital, as cared for by an ambulatory care setting within a defined period of time.

G. Team BMI: Management of Pediatric Overweight and Obesity in Pediatric Primary Care (Allen et al., 2014)

Measure Source: Medical University of South Carolina, Center for Evidence Based Practice

Measure Description: Percentage of patients, ages 2-17 years with overweight or obese Body Mass Index (BMI).

Numerator Statement: Number of patients, ages 2-17 years with overweight (over 85th percentile) BMI and obese (over 95th percentile) BMI, as seen in defined period of time.

Denominator Statement: Total number of patient visits for a defined period of time.

Special Comments or Considerations

Application of 5-2-1-0 patient and family education. Obesity counseling has been a long-standing component of well-child anticipatory guidance; however, more clear, consistent, and consolidated recommendations are needed to make clinical counseling more useful. In response, the Main Youth Overweight Collaborative initiated a 5-2-1-0 campaign applying this pneumonic: 5 or more servings of fruits and vegetables, <2 hours of screen time (TV, computer, video games), at least 1 hour of physical activity and no (0) sugar-sweetened beverages (Foltz et al., 2011).

SECTION 6 Glossary of Definitions and Supportive Statements

- A. AAACN Definition of the Role of the Nurse in Ambulatory
- B. Dimensions of the Ambulatory Care Nurse Role
- C. Definition of Care Coordination
- D. Definition of Transition Management
- E. Definition of the Role Dimensions of Nurse Care Coordination and Transition Management
- F. Nursing Alliance for Quality Care Definition of Patient Engagement

SECTION 6 GLOSSARY OF DEFINITIONS AND SUPPORTIVE STATEMENTS

A. AAACN Definition of the Role of the Nurse in Ambulatory

Background

Ambulatory care nursing is a unique realm of specialized nursing practice. Ambulatory nurses are leaders in their practice settings and across the continuum of care. They are uniquely qualified to influence organizational standards related to patient safety and care delivery in the outpatient setting. Ambulatory care nurses are knowledge workers who function in a multidisciplinary, collaborative practice environment, where they utilize critical thinking skills to interpret complex information and guide patients and families to health and well being (Swan, Conway-Phillips, & Griffin, 2006).

"Historically, the outpatient setting was the 'professional home' of physicians. They saw the majority of their patients in their offices and referred them for other services or levels of care, as needed. Registered nurses were few, as the system was physician driven. However, fiscal caps for hospital care and technological advances moved patients from inpatient venues into the ambulatory care setting. Patients required higher levels of care than in the traditional outpatient settings, and the ambulatory venue saw a growth in the number of professional nurses" (Mastal, 2010, p. 267).

The transition of health care from the inpatient to the outpatient setting has led to challenges with access to care and coordination of services, and has increased the complexity of care delivered outside the hospital walls. This shift has dramatically increased the need for professional nursing services, as patients and their families require increased depth and breadth of care. Ambulatory RNs facilitate patient care services by managing and individualizing care for patients and their families, who increasingly require assistance navigating the complex health care system. In addition to the provision of complex procedural care, professional nursing services provide support with decision-making, patient education and coordination of services.

"Many characteristics differentiate ambulatory care nursing from other specialty practices, including the settings, the characteristics of the patient encounters and focus on groups, communities and populations, as well as individual patients and their families" (Mastal, 2010, p. 267). The current ambulatory care setting is diverse and multifaceted, requiring nurses highly skilled in patient assessment and with the ability to implement a broad range of nursing interventions in a variety of settings. RNs in ambulatory care must possess strong clinical, education and advocacy skills and demonstrate the ability to manage care in complex organizational systems. Registered nurses are uniquely qualified, autonomous providers of patient/family-centered care that is ethical, evidence-based, safe, expert, innovative, healing, compassionate and universally accessible.

Efforts to conserve financial and nursing resources, along with a lack of understanding of differing roles, has led many organizations to under-utilize RNs in ambulatory settings. The economic benefit of care delivered by RNs has been demonstrated by their impact on patient satisfaction, quality patient outcomes, patient safety, reduced adverse events, and reductions in hospital/emergency department admissions (Haas, 2008; Institute of Medicine, 2011; O'Connell, Johnson, Stallmeyer, & Cokingtin, 2001). The future of the American health care system depends upon our ability to utilize registered nurses to the maximum of their expertise, licensure and certification.

Position Statement

It is the position of the American Academy of Ambulatory Care Nursing that:

- RNs enhance patient safety and the quality and effectiveness of care delivery and are thus essential and irreplaceable in the provision of patient care service in the ambulatory setting.
- RNs are responsible for the design, administration and evaluation of professional nursing services within the organization in accordance with the framework established by state nurse practice acts, nursing scope of practice and organizational standards of care.
- RNs provide the leadership necessary for collaboration and coordination of services, which includes defining the appropriate skill mix and delegation of tasks among licensed and unlicensed health care workers.
- RNs are fully accountable in all ambulatory care settings for all nursing services and associated patient outcomes provided under their direction.

Source: AAACN, 2010. Used with permission.

B. Dimensions of the Ambulatory Care Nurse Role

AAACN defined nine core dimensions of the staff nurse clinical practice role and three core dimensions of the quality improvement/research role in ambulatory care (Haas & Hackbarth, 1995).

Clinical Practice Role

Factor 1: Enabling Operations

Factor 2: Technical Procedures

Factor 3: Nursing Process

Factor 4: Telephone Communication

Factor 5: Advocacy

Factor 6: Client Teaching

Factor 7: High-Tech Procedures

Factor 8: Care Coordination

Factor 9: Expert Practice/Community Outreach

Quality Improvement/ Research Role

Factor 1: Quality Improvement

Factor 2: Research

Factor 3: Continuing Education

C. Definition of Care Coordination

McDonald et al. (2007 & 2014) state:

Care Coordination is the deliberate organization of patient care activities between two or more participants (including the patient) involved in a patient's care to facilitate the appropriate delivery of health care services. Organizing care involves the marshalling of personnel and other resources needed to carry out all required patient care activities and is often managed by the exchange of information among participants responsible for different aspects of care (p. 4).

The care coordination process requires:

- The professional assessment of risk stratified patient/population needs and preferences,
- The identification, planning and organizing of appropriate organizational and community resources through the use of the nursing process,
- Communication and knowledge sharing among a multidisciplinary team and the patient/family/population resulting in full patient activation in his or her care,
- Evaluation of the outcomes of health care received.

This approach operates through an integrated multidisciplinary health care team inclusive of a professional care coordinator, who optimally is a registered nurse."

Source: AAACN, 2016. Used with permission.

D. Definition of Transition Management

Haas, Swan, and Haynes (2014) state:

A critical element inherent in care coordination is Transition Management which is the ongoing support of patients and their families over time as they navigate care and relationships among more than one provider and/or more than one health care setting and/or more than one health service. The need for transition management is not determined by age, time, place, or health care condition, but rather by patients' and/or families' needs for support for ongoing, longitudinal individualized plans of care and follow-up plans of care within the context of health care delivery (p. 3).

This process necessitates professional assessment, risk identification, stratification of group and individual patient needs and preferences that require:

- Multidisciplinary collaboration
- Evidence-based care delivery
- Patient and/ or caregiver activation and empowerment
- Utilization of quality and safety standards
- Ability to work independently in the domain of nursing to identify and access community resources which meet individual or group needs."

Source: AAACN 2016. Used with permission.

E. Definition of the Role Dimensions of Nurse Care Coordination and Transition Management

- Support for Self-Management
- Advocacy
- Education and Engagement of Patient and Family
- Cross Setting Communication and Transition
- Coaching and Counseling of Patients and Families
- Nursing Process
- Population Health Management
- Teamwork and Collaboration
- Patient Centered Care Planning

Source: Haas & Swan, 2014.

F. Nursing Alliance for Quality Care Definition of Patient Engagement

The Nursing Alliance for Quality Care (NAQC) is a partnership among the nation's leading nursing organizations, consumers, and other key stakeholders to advance the highest quality, safety, and value of consumer-centered health care for all individuals, their families, and their communities (NAQC, 2016). AAACN provides a representative to sit on this committee and various members from NAQC served as content experts in the development of the Patient Engagement: Measurement of Patient Activation Measure proposed Nurse-Sensitive Indicator described in Section 4B of this report. NAQC has permitted the use of the documents and statements included here as well as in the aforementioned text.

"Patient engagement is the involvement in their own care by individuals (and others they designate to engage on their behalf), with the goal that they make competent, well-informed decisions about their health and health care and take action to support those decisions" (Sofaer & Schumann, 2013, p. 5).

Introduction & Definition

According to Sofaer and Schumann (2013):

Patient engagement is a key element and even a necessary condition for the achievement of patient-centered care (Gerteis, Edgman-Levitan, Daley, & Delbanco, 1993; Institute of Medicine, 2001). In 2008, the National Priorities Partnership (NPP) of the National Quality Forum (NQF) identified patientcentered care that encompasses full engagement of patients and their families in shared decision-making processes as one of its six key priorities (NPP, 2008). NPP recommended its inclusion as a priority of the National Quality Strategy (NQS) (U.S. Department of Health and Human Services, 2011). The Center for Medicare & Medicaid Innovation (CMMI) has created a national Partnership for Patients (P4P) that places a strong emphasis on patient engagement and along with the NPP drives implementation of the priority. The American Academy of Nursing (AAN), as part of its efforts to identify the need to measure the impact of nursing, has published an Action Brief on measurement of patient engagement (Pelletier and Stichler, 2013). RN daily experiences include working with patients and families who have often fared poorly because patients' concerns, preferences and knowledge have not been valued (p. 5).

NAQC Guiding Principles for Patient Engagement

Figure 1. Guiding Principles for Patient Engagement

Patient engagement is a critical cornerstone of patient safety and quality. NAQC has grounded its approach to this topic by recognizing the primary importance of *relationships* between engaged patients and families and their clinicians, including but not limited to nurses. The following are principal assumptions that guide NAQC in addressing care that is patient centered.

- 1 There must be an active partnership among patients, their families, and the providers of their health care.
- 2. Patients are the best and ultimate source of information about their health status and retain the right to make their own decisions about care.
- 3. In this relationship, there are shared responsibilities and accountabilities among the patient, the family, and clinicians that make it effective.
- 4. While embracing partnerships, clinicians must nevertheless respect the boundaries of privacy, competent decision-making, and ethical behavior in all their encounters and transactions with patients and families. These boundaries protect recipients as well as providers of care. This relationship is grounded in confidentiality, where the patient defines the scope of the confidentiality.
- 5. This relationship is grounded in an appreciation of patient's rights and expands on the rights to include mutuality. Mutuality includes sharing of information, creation of consensus, and shared decision-making.
- Clinicians must recognize that the extent to which patients and family members are able to engage or choose to engage may vary greatly based on individual circumstances, cultural beliefs and other factors.
- 7. Advocacy for patients who are unable to participate fully is a fundamental nursing role. Patient advocacy is the demonstration of how all of the components of the relationship fit together.
- 8. Acknowledgment and appreciation of culturally, racially or ethnically diverse backgrounds is an essential part of the engagement process.
- 9. Health care literacy and linguistically appropriate interactions are essential for patient, family, and clinicians to understand the components of patient engagement. Providers must maintain awareness of the language needs and health care literacy level of the patient and family and respond accordingly.

Source: Sofaer & Schumann, 2013. Used with permission.

SECTION 7 References and Additional Readings

SECTION 7 REFERENCES AND ADDITIONAL READINGS

A. References

- Adams, K.F., Schatzkin, A., Harris, T.B., Kipnis, V., Mouw, T., Ballard-Barbash, R., ... Leitzmann, M.F. (2006). Overweight, obesity, and mortality in a large prospective cohort of persons 50 to 71 years old. *New England Journal of Medicine*, *355*(8), 763–778.
- Agency for Healthcare Research and Quality. (2014). Chapter 1. Introduction to the emergency severity index: A research-based triage tool. Rockville, MD: Author. Retrieved from http://www.ahrq.gov/professionals/systems/hospital/esi/esi1.html
- Agency for Healthcare Research and Quality. (2015). *Clinician and group CAHPS survey*. Retrieved from https://cahps.ahrq.gov/Surveys-Guidance/CG/index.html
- Aiken, L.H., Clarke, S.P., Sloane, D.M., Sochalski, J., & Silber, J.H. (2002). Hospital nurse staffing and patient mortality, nurse burnout and job dissatisfaction. *Journal of the American Medical Association, 288*(16), 1987-1993.
- Allen, A., Brown, M., Gustafson, K., Kamal, Z., Litsey, J., Odulana, D., ... Crabtree, E. (2014). *Team BMI: Management of pediatric overweight and obesity in peds primary care clinic.* Medical University of South Carolina Center for Evidence Based Practice. Quality Management Library, Medical University of South Carolina.
- American Academy of Ambulatory Care Nursing. (2010). The role of the registered nurse in ambulatory care position statement. Pitman, NJ: Author. Retrieved from https://www.aaacn.org/sites/default/files/documents/PositionStatementRN.pdf
- American Academy of Ambulatory Care Nursing. (2011). Scope and standards of practice for professional telehealth nursing. Pitman, NJ: Author.
- American Academy of Ambulatory Care Nursing. (2011). What is ambulatory care nursing? Pitman, NJ: Author. Retrieved from https://www.aaacn.org/what-ambulatory-care-nursing
- American Academy of Ambulatory Care Nursing. (2016). Scope and standards of practice for registered nurses in care coordination and transition management. Pitman, NJ: Author.
- American Geriatrics Society, British Geriatrics Society, & American Academy of Orthopaedic Surgeons Panel on Falls Prevention. (2001). Guideline for the prevention of falls in older persons. *Journal of the American Geriatrics Society*, 49(5), 664-672.
- American Hospital Association. (2013). Engaging health care users: A framework for healthy individuals and communities. Chicago, IL: American Hospital Association 2012 Committee on Research, Benjamin K. Chu and John G. O'Brien, co-chairs. Retrieved from http://www.aha.org/research/cor/content/engaging_health_care_users.pdf
- American Hospital Association. (2015). Utilization and volume. In *Trend watch chart book 2015: Trends affecting hospitals and health systems*. Washington, DC: Author. Retrieved from http://aha.org/research/reports/tw/chartbook/2014/chapter3.pdf
- American Institutes for Research. (2014). A roadmap for patient and family engagement in healthcare practice and research: Practical strategies for advancing engagement in healthcare—Starting today. Palo Alto, CA: Gordon and Betty Moore Foundation. Retrieved from http://patientfamilyengagement.org
- American Nurses Association. (1996). *Nursing quality indicators: Definitions and implications*. Washington, DC: American Nurses Publishing.
- American Nurses Association. (2000). *Nursing quality indicators beyond acute care: Measurement instruments*. Washington, D.C.: American Nurses Publishing.

- American Nurses Credentialing Center. (2014). 2014 Magnet® application manual. Silver Spring, MD: American Nurses Credentialing Center.
- Arroll, B., Goodyear-Smith, F., Crengle, S., Gunn, J., Kerse, N., Fishman, T., ... Hatcher, S. (2010). Validation of PHQ-2 and PHQ-9 to screen for major depression in the primary care population. *The Annals of Family Medicine*, *8*(4), 348-353.
- Arroll, B., Khin, N., & Kerse, N. (2003). Screening for depression in primary care with two verbally asked questions: Cross sectional study. *BMJ (Clinical research ed), 327*(7424), 1144–1146.
- Bandura, A. (2004). Health promotion by social cognitive means. Health Education & Behavior, 31(2),143-164.
- Battaglia, R. (2006). Pediatric blood pressure: Nurses must be alert for primary hypertension in children. *Advance for Nurses Southeastern States*, 6(23), 29.
- Bekeris, L.G., Jones, B.A., Walsh, M.K., & Wagar, E.A. (2008). Urine culture contamination: A College of American Pathologists Q-Probes study of 127 laboratories. *Archives of Pathology and Laboratory Medicine, 132*(6), 913-917.
- Berry, J.G., Hall, D.E., Kuo, D.Z., Cohen, E., Agrawal, R., Feudtner, C., ... Neff, J. (2011). Hospital utilization and characteristics of patients experiencing recurrent readmissions within children's hospitals. *JAMA*, 305(7), 682-690.
- Berry, J.G., Toomey, S.L., Zaslavsky, A.M., Jha, A.K., Nakamura, M.M., Klein, D. J., ... Schuster, M.A. (2013). Pediatric readmission prevalence and variability across hospitals. *JAMA*, *309*(4), 372-380.
- Berwick, D.M., Nolan, T.W., & Whittington, J. (2008). The triple aim: Care, health, and cost. *Health Affairs*, 27(3), 759-769
- Breslin, E. (2015). AACN remarks: IOM Committee on the Evaluation of the Future of Nursing Report. Retrieved from http://www.aacn.nche.edu/media-relations/IOM-Evaluation-Committee-Talking-Points.pdf
- Brown, D.S., Donaldson, N., Burnes Bolton, L., & Aydin, C.E. (2010). Nursing-sensitive benchmarks for hospitals to gauge high-reliability performance. *Journal of Healthcare Quality*, 32(6), 9-17. doi: 10.1111/j.1945-1474.2010.00083.x.
- Brown, D.S., & Wolosin, R. (2013). Safety culture relationships with hospital nursing sensitive metrics. *Journal of Healthcare Quality*, *35*(4), 61-74. doi: 10.1111/jhq.12016.
- Bunn, F., Byrne, G., & Kendall, S. (2004). Telephone consultation and triage: effects on health care use and patient satisfaction. *The Cochrane Database of Systematic Reviews, 4.*
- Burnet, D., Gunter, K.E., Nocon, R.S., Gao, Y., Jin, J., Fairchild, P., & Chin, M.H. (2014). Medical home characteristics and the pediatric patient experience. *Medical Care*, *52*(11 Suppl 4), S56-S63. doi: 10.1097/MLR.000000000000238
- Carman, K.L., Dardess, P., Maurer, M., Sofaer, S., Adams, K., Betchel, C., & Sweeney, J. (2013). Patient and family engagement: A framework for understanding the elements and developing interventions and policies. *Health Affairs*, 32(2), 223-231.
- Center for Medicare and Medicaid Innovation. (2014). Comprehensive primary care initiative: eCQM user manual. Version 3.0. Retrieved from https://innovation.cms.gov/files/x/cpci-ecqm-manual2014.pdf
- Centers for Disease Control and Prevention (CDC). (2012). Adult vaccination coverage--United States, 2010. *Morbidity and Mortality Weekly Report*, 61(4), 66-72.
- (CDC). promotion: Centers for Disease Control and Prevention (2013)Workplace health http://www.cdc.gov/workplacehealthpromotion/ Depression. Atlanta. GA: Author. Retrieved from implementation/topics/depression.html

- Centers for Medicare & Medicaid Services (CMS). (2014a). 2015 PQRS GPRO web interface measure specifications release notes. Measure # GPRO PREV 11: Preventive care and screening: Screening for high blood pressure and follow up documented. Retrieved from https://www.cms.gov/medicare/quality-initiatives-patient-assessment-instruments/pqrs/gpro web interface.html
- Centers for Medicare & Medicaid Services (CMS). (2014b). 2015 PQRS GPRO web interface measure specifications release notes. Measure # GPRO PREV 12: Preventive care and screening: Screening for clinical depression and follow up plan. Retrieved from https://www.cms.gov/medicare/quality-initiatives-patient-assessment-instruments/pqrs/gpro_web_interface.html
- Centers for Medicare & Medicaid Services (CMS). (2014c). 2015 PQRS GPRO web interface measure specifications release notes. Measure # GPRO CARE 2: Falls: Screening for future falls risk. Retrieved from https://www.cms.gov/medicare/quality-initiatives-patient-assessment-instruments/pgrs/gpro web interface.html
- Centers for Medicare & Medicaid Services (CMS). (2014d). 2015 PQRS #154: Falls: Risk assessment. In *Physician quality reporting system measure*. Retrieved from https://pqrs.cms.gov/#/home
- Centers for Medicare & Medicaid Services (CMS). (2014e). 2015 PQRS GPRO web interface measure specifications release notes. Measure # GPRO PREV 9: Preventive care and screening: Body mass index screening and follow-up. Retrieved from https://www.cms.gov/medicare/quality-initiatives-patient-assessment-instruments/pqrs/gpro_web_interface.html
- Centers for Medicare & Medicaid Services (CMS). (2015a). ASC 3: Wrong site, wrong side, wrong procedure, wrong implant. In *Ambulatory surgical center quality reporting program: Quality measures specifications manual.*Version 4.0a. Retrieved from http://www.ascaconnect.org/HigherLogic/System/Download DocumentFile.ashx?DocumentFileKey=b3148327-3606-fea9-b17a-e47bffad6931&forceDialog=0
- Centers for Medicare & Medicaid Services (CMS). (2015b). ASC 1: Patient burn. In Ambulatory surgical center quality reporting program: Quality measures specifications manual. Version 4.0a. Retrieved from http://www.ascaconnect.org/HigherLogic/System/DownloadDocumentFile.ashx?DocumentFileKey=b3148327-3606-fea9-b17a-e47bffad6931&forceDialog=0
- Centers for Medicare & Medicaid Services (CMS). (2015c). ASC 2: Patient fall. In Ambulatory surgical center quality reporting program: Quality measures specifications manual. Version 4.0a. Retrieved from http://www.ascaconnect.org/HigherLogic/System/DownloadDocumentFile.ashx?DocumentFileKey=b3148327-3606-fea9-b17a-e47bffad6931&forceDialog=0
- Centers for Medicare & Medicaid Services (CMS). (2015d). ASC 4: Hospital transfer/admission. In *Ambulatory surgical center quality reporting program: Quality measures specifications manual.* Version 4.0a. Retrieved from http://www.ascaconnect.org/HigherLogic/System/DownloadDocumentFile.ashx?DocumentFileKey=b3148327-3606-fea9-b17a-e47bffad6931&forceDialog=0
- Centers for Medicare & Medicaid Services (CMS). (2015e). Accountable Care Organization Measure #8: Risk Standardized All Condition Readmissions. Version 9. CMS. Retrieved from https://www.cms.gov/medicare/medicare-fee-for-service-payment/sharedsavingsprogram/downloads/aco-8.pdf
- Centers for Medicare & Medicaid Services (CMS). (2015f). PQRS #131. Pain assessment and follow up. In *Physician quality reporting system measure specifications manual for claims and registry reporting of individual measures*. Retrieved from https://pqrs.cms.gov/#/measure/Pain%2520Assessment%2520and%2520 FollowUp?originalQuery=%7B%22measure number%22:%7B%22pqrs%22:131%7D%7D&originalPage=1
- Centers for Medicare & Medicaid Services (CMS). (2015g). Consumer assessment of healthcare providers and systems (CAHPS). Retrieved from https://www.cms.gov/Research-Statistics-Data-and-Systems/Research/CAHPS/index.html?redirect=/CAHPS/
- Chase, D. (2011). Patients gain information and skills to improve self-management through innovative tools. *Quality Matters*, December 2010/January 2011. Retrieved from http://www.commonwealthfund.org/~/media/files/newsletters/quality-matters/qm 2010 dec jan 2011.pdf

- Ciechanowski, P.S., Katon, W.J., & Russo, J.E. (2000). Depression and diabetes: Impact of depressive symptoms on adherence, function, and costs. *Archives of Internal Medicine*, *160*(21), 3278-3285.
- Collaborative Alliance for Nursing Outcomes (CALNOC). (2014). *Measures codebook supplement*. San Ramon, CA: Author.
- Collaborative Alliance for Nursing Outcomes (CALNOC). (2015). *Measures codebook supplement*. San Ramon, CA: Author.
- Davis, K., Stemikis, K., Squires, D., & Schoen, C. (2014). 2014 Update: Mirror, mirror on the wall: How the performance of the U.S. health care system compares internationally. Commonwealth Fund pub. .no. 1755.

 New York, NY: Commonwealth Fund. Retrieved from http://www.commonwealthfund.org/~/media/files/publications/fund-report/2014/jun/1755_davis_mirror_mirror_2014.pdf
- Donabedian, A. (1988). The quality of care: How can it be assessed? *Journal of the American Medical Association*, 260(12), 1743-1748.
- Doran, D.M. (2011). Nursing outcomes: State of the science. 2nd Ed. Sudbury, MA: Jones & Bartlett Learning.
- Dossey, B.M., Selanders, L.C., Beck D.M., & Attewell, A. (2005). Florence Nightingale today: Healing, leadership, global action. Silver Spring, MD: American Nurses Association.
- Farley, T.A., Dalal, M.A., Mostashari, F., & Frieden, T.R. (2010). Deaths preventable in the U.S. by improvements in the use of clinical preventive services. *American Journal of Preventive Medicine*, 38(6), 600-609.
- Flegal, K.M., Carroll, M.D., Ogden, C.L., & Johnson, C.L. (2002). Prevalence and trends in obesity among U.S. adults, 1999-2000. *Journal of the American Medical Association*, 288(14), 1723-1727.
- Foltz, J.L., Cook, S.R., Szilagyi, P.G., Auinger, P., Stewart, P.A., Bucher, S., & Baldwin, C.D. (2011). U.S. adolescent nutrition, exercise, and screen time baseline levels prior to national recommendations. *Clinical Pediatrics*, 50(5), 424-33.
- Ford, D.E., Mead, L.A., Chang, P.P., Cooper-Patrick, L., Wang, N.Y., & Klag, M.J. (1998). Depression is a risk factor for coronary artery disease in men: The precursors study. *Archives of Internal Medicine*, *158*(13), 1422-1426.
- Gerteis, M., Edgman-Levitan, S., Daley, J., & Delblanco, T.L. (1993). Through the patient's eyes: Understanding and promoting patient-centered care. San Francisco, CA: Jossey-Bass.
- Gibbons, R.D., Hur, K., Bhaumik, D.K., & Mann, J.J. (2005). The relationship between antidepressant medication use and rate of suicide. *Archives of General Psychiatry*, *62*(2), 165-172.
- Gibly, R.L. (1998). Infections of the urinary tract and male genitalia. In Brillman, J.C. & Quenzer, R.W. (Eds.), *Infectious disease in emergency medicine, 2nd ed.* Philadelphia, PA: Lippincott-Raven.
- Gilbody, S., House, A.O., & Sheldon, T.A. (2005). Screening and case finding instruments for depression. *The Cochrane Database of Systematic Reviews*, (4), CD002792.
- Gilboy, N., Tanabe, T., Travers, D., & Rosenau, A.M. (2011). *Emergency Severity Index (ESI): A triage tool for emergency department care, Implementation handbook, 4th ed.* AHRQ Publication no. 12-0014. Rockville, MD: Agency for Healthcare Research and Quality.
- Green, N.A., Durani, Y., Brecher, D., DePiero, A., Loiselle, J., & Attia, M. (2012). Emergency severity index version 4: A valid and reliable tool in pediatric emergency department triage. *Pediatric Emergency Care*, *28*(8), 753-757. doi:10.1097/PEC.0b013e3182621813
- Guide to Community Preventive Services. (2015). *Increasing appropriate vaccination: Standing orders (abbreviated)*. Retrieved from http://www.thecommunityguide.org/vaccines/standingorders.html
- Haas, S.A. (2008). Resourcing evidence-based practice in ambulatory care nursing. *Nursing Economic*\$, 26(5), 319-322.

- Haas, S.A., & Hackbarth, D.P. (1995). Dimensions of the staff nurse role in ambulatory care: Part III--Using research data to design new models of nursing care delivery. *Nursing Economic*\$, 13(4), 230-241.
- Haas, S.A., & Swan, B.A. (2014). Developing the value proposition for the role of the registered nurse in care coordination and transition management in ambulatory care settings. *Nursing Economic\$*, *32*(2), 70-79.
- Haas, S.A., Swan, B.A., & Haynes, T.S. (Eds.) (2014). *Care coordination and transition management core curriculum*. Pitman, NJ: American Academy of Ambulatory Care Nursing.
- Hamner, J.B. (2005). State of the science: Posthospitalization nursing interventions in congestive heart failure. *Advances in Nursing Science*, *28*(2), 175-190.
- Health Research & Educational Trust (HRET). (2013). *A leadership resource for patient and family engagement strategies*. Chicago, IL: Author. Retrieved from www.hpoe.org/Patient-family-engagement
- Heslop, L., & Lu, S. (2014). Nursing sensitive indicators: A concept analysis. *Journal of Advanced Nursing*, 70(11), 2469-2482. doi: 10.1111/jan.12503.Epub 2014 Aug 12.
- Hibbard, J.H. (2009). Using systematic measurement to target consumer activation strategies. *Medical Care Research and Review, 66*(1 Suppl), 9S-27S.
- Hibbard, J.H., & Greene, J. (2013). What the evidence shows about patient activation: Better health outcomes and care experiences; Fewer data on costs. *Health Affairs*, *32*(2), 207-214. doi:10.1377/hlthaff.2012.1061.
- Hibbard, J.H., Greene, J., Becker, E.R., Roblin, D., Painter, M.W., Perez, D.J., ... Tusler, M. (2008). Racial/ethnic disparities and consumer activation in health. *Health Affairs*, 27(5), 1442-1453. doi:10.1377/hlthaff.27.5.1442.
- Hibbard, J.H., Greene, J., & Overton, V. (2013). Patient with lower activation associated with higher costs; Delivery systems should know their patients' scores. *Health Affairs*, *32*(2), 216-222. doi:10.1377/hlthaff.2012.1064
- Hibbard, J.H., & Mahoney, E. (2010). Toward a theory of patient and consumer activation. *Patient Education and Counseling*, 78(3), 377-381. doi: 10.1016/j.pec.2009.12.015.Epub 2010 Feb 25.
- Hibbard, J.H., Mahoney, E.R., Stock, R., & Tusler, M. (2006). Do increases in patient activation result in improved self-management behaviors? *Health Services Research*, 42(4), 1443-1463.
- Hibbard, J.H., Mahoney, E.R., Stockard, J., & Tusler, M. (2005). Development and testing of a short form of the patient activation measure. *Health Services Research*, 40(6 Pt 1), 1918-30.
- Hibbard, J.H., Stockard, J., Mahoney, E.R., & Tusler, M. (2004). Development of the Patient Activation Measure (PAM): Conceptualizing and measuring activation in patients and consumers. *Health Services Research*, 39(4 Pt 1), 1005-1026.
- Hooten, W.M., Timming, R., Belgrade, M., Gaul, J., Goertz, M., Haake, B., ... Walker, N. (2013). *Health care guideline: Assessment and management of chronic pain.* Bloomington, MN: Institute for Clinical Systems Improvement. Retrieved from https://www.icsi.org/ asset/bw798b/ChronicPain.pdf
- Horwitz, L., Partovian, C., Lin, Z., Herrin, J., Grady, J., Conover, M., ... Bernheim, S. (2011). *Hospital-wide (all-condition) 30-day risk-standardized readmission measure. Draft measure methodology report.* Yale New Haven Health Services Corporation Center for Outcomes Research and Evaluation (YNHHSC/CORE). Contract number: HHSM-500-2008-0025I/HHSM-500-T0001, Modification No. 000005
- Huibers, L., Smits, M., Renaud, V., Giesen, P., & Wensing, M. (2011). Safety of telephone triage in out-of-hours care: A systematic review. *Scandinavian Journal of Primary Health Care*, *29*(4), 198-209.
- Institute of Medicine. (2001). Crossing the quality chasm: A new heath system for the 21st Century. Washington, DC: National Academies Press.
- Institute of Medicine. (2010). The Future of nursing: Leading change, advancing health. Washington, DC: National Academies Press.

- Institute of Medicine. (2011). Relieving pain in America: A blueprint for transforming prevention, care, education and research. Washington, DC: National Academies Press.
- Institute of Medicine. (2015). Assessing progress on the IOM report the future of nursing. Washington, DC: The National Academies Press.
- Jencks, S.F., Williams, M.V., & Coleman, E.A. (2009). Rehospitalization among patients in the Medicare fee-for-service program. *New England Journal of Medicine*, *360*(14), 1418-1428. doi:10.1056/NEJMsa0903563.
- Jenkinson, C., Coulter, A., & Bruster, S. (2002). The Picker Patient Experience Questionnaire: Development and validation using data from in-patient surveys in five countries. International *Journal for Quality in Health Care,* 14(5), 353-358.
- Jorge, R.E., Robinson, R.G., Arndt, S., & Starkstein, S. (2003). Mortality and poststroke depression: A placebocontrolled trial of antidepressants. *American Journal of Psychiatry*, *160*(10), 1823-1829.
- Katon, W.J. (2003). Clinical and health services relationships between major depression, depressive symptoms, and general medical illness. *Biological Psychiatry*, *54*(3), 216-226.
- Kroenke, K., Spitzer, R.L., Williams, J.B., & Löwe, B. (2010). The patient health questionnaire somatic, anxiety, and depressive symptom scales: A systematic review. *General Hospital Psychiatry*, *32*(4), 345-359.
- Kung, H.C., & Xu, J. (2015). *Hypertension-related mortality in the United States, 2000–2013.* NCHS Data Brief, no. 193. Hyattsville, MD: National Center for Health Statistics.
- Lewis, L.C. (2014). Charting a new course: Advancing the next generation of nursing-sensitive indicators. *The Journal of Nursing Administration*, 44(5), 247-249. doi: 10.1097/NNA.000000000000001.
- Lobstein, T., Baur, L., & Uauy, R. (2004). Obesity in children and young people: A crisis in public health. *Obesity Reviews*. *5*(Suppl 1), 4–104.
- Lopez, A.D., & Murray, C.C. (1998). The global burden of disease, 1990-2020. Nature Medicine, 4(11), 1241-1243.
- Löwe, B., Kroenke, K., & Gräfe, K. (2005). Detecting and monitoring depression with a two-item questionnaire (PHQ-2). *Journal of Psychosomatic Research*, *58*(2), 163-171.
- Marjoua, Y., & Bozic, K.J. (2012). Brief history of quality movement in US healthcare. *Current Reviews in Musculoskeletal Medicine*, *5*(4), 265-273.
- Marshall, R., Beach, M.C., Saha, S., Mori, T., Loveless, M.O., Hibbard, J.H., ... Korthuis, P.T. (2013). Patient activation and improved outcomes in HIV-infected patients. *Journal of General Internal Medicine*, *28*(5), 668-674. doi:10.1007/s11606-012-2307-y
- Martinez, K., Battaglia, R., Start, R., Mastal, M.F., & Matlock, A.M. (2015). Nursing-sensitive indicators in ambulatory care. *Nursing Economic\$*, 33(1), 59-63, 66.
- Mastal, M.F. (2010). Ambulatory care nursing: Growth as a professional specialty. *Nursing Economic*\$, 28(4), 267-269, 275.
- Mastal, M.F., Vinson, M., Bord-Hoffman, M.A., Futch, C., Harden, L., Jessie, A., ... Rutenberg, C. (2012). American Academy of Ambulatory Care Nursing position statement: The role of the registered nurse in ambulatory care. *Nursing Economic*\$, 30(4), 233-239.
- Maurer, D.M. (2012). Screening for depression. American Family Physician, 85(2), 139-144.
- Maurer, M., Dardess, P., Carman, K.L., Frazier, K., & Smeeding, L. (2012). Guide to patient and family engagement: Environmental scan report. (Prepared by American Institutes for Research, AHRQ Publication No. 12-0042-EF). Rockville, MD: Agency for Healthcare Research and Quality. Retrieved from http://www.ahrq.gov/research/findings/final-reports/ptfamilyscan/ptfamilyscan.pdf

- McCloskey, J.C., & Bulechek, G.M. (Eds). (2000). Nursing interventions classification (NIC), 3rd ed. St. Louis, MO: Mosby-Year Book.
- McDonald, K., Schultz, E.S., Albin, L., Pineda, N., Lonhart, J., Sundaram, V., & Malcolm, E. (2011). *Care coordination measures atlas* (AHRQ Publication No. 11-0023-EF). Rockville, MD: Agency for Healthcare Research and Quality.
- McDonald, K., Sundaram, V., Bravada, D., Lewis, R., Lin, N., Kraft, S.A., & Owens, D.K. (2007). Care coordination. In K. Shojania, K. McDonald, R. Wachter, & D. Owens (Eds.), *Closing the quality gap: A critical analysis of quality improvement strategies* (AHRQ Publication No. 4 (07-0051-7). Technical Review 9 (Prepared by Stanford UCSF Evidence-Based Practice Center under contract No. 290-02-0017). Vol 7. Rockville, MD: Agency for Healthcare Research and Quality.
- McLaughlin, J.M., McGinnis, J.J., Tan, L., Mercatante, A., & Fortuna, J. (2015). Estimated human and economic burden of four major adult vaccine-preventable diseases in the United States, 2013. *The Journal of Primary Prevention*, 36(4), 259–273. doi:10.1007/s10935-015-0394-3
- Michael, M., Schaffer, S.D., Egan, P.L., Little, B.B., & Pritchard, P.S. (2013). Improving wait times and patient satisfaction in primary care. *Journal for Healthcare Quality*, *35*(2), 50-60.
- Mitchell, A.J., & Coyne, J.C. (2007). Do ultra-short screening instruments accurately detect depression in primary care? A pooled analysis and meta-analysis of 22 studies. *British Journal of General Practice*, *57*(535), 144-151.
- Montalvo, I. (2007). The National Database of Nursing Quality Indicators (NDNQI). *OJIN: The Online Journal of Issues in Nursing*, 12(3), Manuscript 2. doi:10.3912/OJIN.Vol12No03Man02
- Mosen, D.M., Schmittdiel, J., Hibbard, J., Sobel, D., Remmers, C., & Bellows, J. (2007). Is patient activation associated with outcomes of care for adults with chronic conditions? *Journal of Ambulatory Care Management,* 30(1), 21-29.
- Moyer, V.A., & U.S. Preventive Services Task Force. (2013). Screening for primary hypertension in children and adolescents: U.S. Preventive Services Task Force recommendation statement. *Annals of Internal Medicine*, 159(9), 613-619.
- National Alliance for Quality Care (NAQC). (2016). Nursing Alliance for Quality Care: About NAQC. Retrieved from http://www.naqc.org/Functional/About-NAQC
- National Committee for Quality Assurance (NCQA). (2014). Summary table of measures, product lines and changes. In HEDIS 2015, Volume 2. Retrieved from http://www.ncqa.org/Portals/0/HEDISQM/Hedis2015/List_of_HEDIS_2015_Measures.pdf
- National Database of Nursing Quality Indicators (NDNQI). (2014). Guidelines for data collection and submission on care coordination in ambulatory settings. Elkridge, MD: Press Ganey.
- National High Blood Pressure Education Program Working Group on High Blood Pressure in Children and Adolescents. (2004). The fourth report on the diagnosis, evaluation, and treatment of high blood pressure in children and adolescents. *Pediatrics*, 114(2 Suppl 4th Report), 555-576.
- National Priorities Partnership (NPP). (2008). *National priorities and goals: Aligning our efforts to transform America's healthcare*. Washington, DC: National Quality Forum.
- National Quality Forum (NQF). (2004). National voluntary consensus standards for nursing sensitive care: An initial performance measure set. Washington, DC: Author. Retrieved from http://www.qualityforum.org/Publications/2004/10/National_Voluntary_Consensus_Standards_for_Nursing-Sensitive_Care__An_Initial_Performance_Measure_Set.aspx
- National Quality Forum (NQF). (2013a). *Measure #0420: Pain assessment and follow-up* (Steward: Centers for Medicare & Medicaid Services). Washington, DC: Author. Retrieved from http://www.qualityforum.org/QPS/QPSTool.aspx

- National Quality Forum (NQF). (2013b). *Measure #0018: Controlling high blood pressure.* (Steward: National Committee for Quality Assurance). Washington, DC: Author. Retrieved from http://www.qualityforum.org/QPS/QPSTool.aspx
- National (NQF). (2013c) Patient (PROs) Quality Forum reported outcomes in performance measurement. Washington, DC: Author. Retrieved from https://www.gualityforum.org/Publications/2012/12/Patient-Reported Outcomes in Performance Measurement.aspx
- National Quality Forum (NQF). (2014a). *Measure #0266: Patient fall* (Steward: Ambulatory Surgical Centers Quality Collaborative). Washington, DC: Author. Retrieved from http://www.qualityforum.org/QPS/QPSTool.aspx
- National Quality Forum (NQF). (2014b). *Measure #1789: Hospital-wide all-cause unplanned readmission measure (HWR)*. (Steward: Centers for Medicare & Medicaid Services). Washington, DC: Author. Retrieved from http://www.qualityforum.org/QPS/QPSTool.aspx
- National Quality Forum (NQF). (2014c). Measure #0418: Preventive care and screening: Screening for clinical depression and follow-up plan. (Steward: Centers for Medicare and Medicaid Services). Washington, DC: Author. Retrieved from http://www.qualityforum.org/QPS/QPSTool.aspx
- National Quality Forum (NQF). (2014d). Measure #0421: Preventive care and screening: Body mass index (BMI) screening and follow-up (Steward: Centers for Medicare and Medicaid Services). Washington, DC: Author. Retrieved from http://www.qualityforum.org/QPS/QPSTool.aspx
- National Quality Forum. (2015a). Measure #0267: Wrong site, wrong side, wrong patient, wrong procedure, wrong implant (Steward: Ambulatory Surgical Centers Quality Collaborative). Washington, DC: Author. Retrieved from http://www.qualityforum.org/QPS/QPSTool.aspx
- National Quality Forum (NQF). (2015b). *Measure #0263: Patient burn* (Steward: Ambulatory Surgical Centers Quality Collaborative). Washington, DC: Author. Retrieved from http://www.qualityforum.org/QPS/QPSTool.aspx
- National Quality Forum (NQF). (2015c). *Measure #0265: All-cause hospital transfer/admission* (Steward: Ambulatory Surgical Centers Quality Collaborative). Washington, DC: Author. Retrieved from http://www.qualityforum.org/QPS/QPSTool.aspx
- National Quality Forum (NQF). (2015d). *Measure #2393: Pediatric all-condition readmission measure*. (Steward: Center of Excellence for Pediatric Quality Measurement). Washington, DC: Author. Retrieved from http://www.qualityforum.org/QPS/QPSTool.aspx
- National Quality Forum (NQF). (2015e). Behavioral health endorsement maintenance 2014 final report Phase 3. Washington, DC: Author. Retrieved from http://www.qualityforum.org/Publications/2015/05/Behavioral_Health_ Endorsement_Maintenance_2014_Final_Report_-_Phase_3.aspx
- National Quality Forum (NQF). (2015f). *Measure #0141: Patient fall rate* (Steward: American Nurses Association). Washington, DC: Author. Retrieved from http://www.qualityforum.org/QPS/QPSTool.aspx
- National Quality Forum (NQF). (2015g). Measure #0101: Falls: Screening, risk-assessment, and plan of care to prevent future falls (Steward: National Committee for Quality Assurance). Washington, DC: Author. Retrieved from http://www.qualityforum.org/QPS/QPSTool.aspx
- Nightingale, F. (1859). Notes on nursing: What it is, and what it is not. Philadelphia, PA; Edward Stern & Company.
- Nwankwo, T., Yoon, S.S., Burt, V., & Gu, Q. (2013). Hypertension among adults in the United States: National Health and Nutrition Examination Survey, 2011-2012. *NCHS Data Brief*, (133), 1-8.
- O'Connell, J.M., Johnson, D.A., Stallmeyer, J., & Cokingtin, D. (2001). A satisfaction and return-on-investment study of a nurse triage service. *American Journal of Managed Care*, 7(2), 159-169.

- Office of Disease Prevention and Health Promotion (ODPHP). (2015). Healthy people clinical preventive services. Retrieved from https://www.healthypeople.gov/2020/leading-health-indicators/2020-lhi-topics/Clinical-Preventive-Services/determinants
- Ogden, C.L., Carroll, M.D., Curtin, L.R., Lamb, M.M., & Flegal, K.M. (2010). Prevalence of high body mass index in US children and adolescents, 2007–2008. *JAMA*, 303(3), 242–249.
- Ogden, C.L., Carroll, M.D., Curtin, L.R., McDowell, M.A., Tabak, C.J., & Flegal, K.M. (2006). Prevalence of overweight and obesity in the United States, 1999-2004. *JAMA*, 295(13), 1549-1555.
- Patient Protection and Affordable Care Act, 111-1, 2nd Session Cong (ACA) (2010). Retrieved from http://www.gpo.gov/fdsys/pkg/BILLS-111hr3590enr/pdf/BILLS-111hr3590enr.pdf
- Pelletier, L.R., & Stichler, J.F. (2013). Action Brief: Patient engagement and activation: A health reform imperative and improvement opportunity for nursing. *Nursing Outlook*, *61*(1) 51-54.
- Pratt, L.A., & Brody, D.J. (2008). Depression in the United States household population, 2005-2006. *NCHS Data Brief*, (7), 1-8.
- Press Ganey. (2015a). *Improve care quality, prevent adverse events with deep nursing quality insights*. Retrieved from http://www.nursingquality.org/About-NDNQI/Quality-Data-Solutions#intro
- Press Ganey. (2015b). Patient experience: Understand patient needs to reduce suffering and improve performance. Retrieved from http://www.pressganey.com/solutions/patient-experience
- Rand Corporation. (2015). Patient satisfaction questionnaire from RAND Health: PSQ 18. Retrieved from http://www.rand.org/health/surveys_tools/psq.html
- Rantz, M. (1995). *Nursing quality measurement: A review of nursing studies*. Washington, DC: American Nurses Publishing. Retrieved from http://www.worldcat.org/title/nursing-quality-measurement-a-review-of-nursing-studies/oclc/32384295
- Richardson, L.P., McCauley, E., Grossman, D.C., McCarty, C.A., Richards, J., Russo, J.E., ... Katon, W. (2010). Evaluation of the Patient Health Questionnaire-9 Item for detecting major depression among adolescents. *Pediatrics*, *126*(6), 1117-1123.
- Risser, N.L. (1975). Development of an instrument to measure patient satisfaction with nurses and nursing care in primary care settings. *Nursing Research*, 24(1), 45-51.
- Robinson R.G., Bolduc, P.L., & Price, T.R. (1987). Two-year longitudinal study of poststroke mood disorders: Diagnosis and outcome at one and two years. *Stroke*, *8*(5), 837-843.
- RTI International & Telligen. (2011a). Measure #13: Falls: Screening for future falls risk. In *Accountable care organization* 2012 program analysis: Quality performance standards, narrative measure specifications (RTI Project Number 0213195.000.004). Baltimore, MD: Quality Measurement and Health Assessment Group. Retrieved from https://www.cms.gov/medicare/medicare-fee-for-service-payment/sharedsavingsprogram/downloads/aco_qualitymeasures.pdf
- RTI International & Telligen. (2011b). Measure #16: Preventive care and screening: Body mass index (BMI) screening and follow-up. In *Accountable care organization 2012 program analysis: Quality performance standards, narrative measure specifications* (RTI Project Number 0213195.000.004). Baltimore, MD: Quality Measurement and Health Assessment Group. Retrieved from https://www.cms.gov/medicare/medicare-fee-for-service-payment/sharedsavingsprogram/downloads/aco_qualitymeasures.pdf
- RTI International & Telligen. (2012a). Measure #8: Risk standardized all condition readmission. In Accountable care organization 2013 program analysis: Quality performance standards, narrative measure specifications (RTI Project Number 0213195.000.004). Baltimore, MD: Quality Measurement and Health Assessment Group. Retrieved from https://www.cms.gov/Medicare/Medicare-Fee-for-Service-Payment/sharedsavingsprogram/Quality-Measures-Standards.html

- RTI International & Telligen. (2012b). Measure #21: Preventive care and screening: Screening for high blood pressure and follow-up documented. In *Accountable care organization 2013 program analysis: Quality performance standards, narrative measure specifications* (RTI Project Number 0213195.000.004). Baltimore, MD: Quality Measurement and Health Assessment Group. Retrieved from https://www.cms.gov/Medicare/Medicare-Fee-for-Service-Payment/sharedsavingsprogram/Quality-Measures-Standards.html
- RTI International & Telligen. (2012c). Measure #18: Preventive care and screening: Screening for clinical depression and follow-up plan. In *Accountable care organization 2013 program analysis: Quality performance standards, narrative measure specifications* (RTI Project Number 0213195.000.004). Baltimore, MD: Quality Measurement and Health Assessment Group. Retrieved from https://www.cms.gov/Medicare/Medicare-Feefor-Service-Payment/sharedsavingsprogram/Quality-Measures-Standards.html
- Rubenstein, L.Z. (2006). Falls in older people: Epidemiology, risk factors and strategies for prevention. *Age and Ageing*, 35(Suppl 2), ii37-ii41.
- Rutherford, M.M. (2014). The value of trust to nursing. Nursing Economic\$, 32(6) 283-288; 327.
- Sabin, M. (1998). Telephone triage improves demand management effectiveness. *Healthcare Financial Management*, 52(8), 49-51.
- Sacks, R.M., Greene, J., Hibbard, J.H., & Overton, V. (2014). How well do patient activation scores predict depression outcomes one year later? *Journal of Affective Disorders, 169*, 1-6. doi:10.1016/j.jad2014.07.030
- Sawyer, L.M., Berkowitz, B., Haber, J.E., Larrabee, J.H., Marino, B.L., Martin, K.S., ... Walker, M.K. (2002). Expanding American Nurses Association nursing quality indicators to community-based practices. *Outcomes Management*, 6(2), 53-61.
- Scholle, S.H., Torda, P., Peikes, D., Han, E., & Genevro, J. (2010). *Engaging patients and families in the medical home*. (Prepared by Mathematica Policy Research under Contract No. HHSA290200900019I TO2.) AHRQ Publication No. 10-0083-EF. Rockville, MD: Agency for Healthcare Research and Quality. Retrieved from https://pcmh.ahrq.gov/page/engaging-patients-and-families-medical-home
- Shrestha, R., Gyawali, N., Gurung, R., Amatya, R., & Kumar-Bhattacharaya, S. (2013). Effect of urogenital cleaning with paper soap on bacterial contamination rate while collecting midstream urine specimens. *Journal of Laboratory Physicians*. *5*(1), 17-20. doi:10.4103/0974-2727.115910
- Skolasky, R.L., Green, A.F., Scharfstein, D., Boult, C., Reider, L., & Wegener, S.T. (2011). Psychometric properties of patient activation measure among multimorbid older adults. *Health Services Research*, *46*(2), 457-478. doi:10.1111/j.1475-6773.2010.01210.x.
- Sofaer, S., & Schumann, M.J. (2013). Fostering successful patient and family engagement: Nursing's critical role. Silver Spring, MD: Nursing Alliance for Quality Care. Retrieved from http://www.nagc.org/WhitePaper-PatientEngagement
- Swan, B.A. (2008). Making nursing-sensitive quality indicators real in ambulatory care. *Nursing Economic*\$, 26(3) 195-201, 205.
- Swan, B.A., Conway-Phillips, R., Griffin, K.F. (2006). Demonstrating the value of the RN in ambulatory care. *Nursing Economic*\$, 24(6), 315-322.
- Swan, B.A., & Haas, S.A. (2011). Health care reform: Current updates and future initiatives for ambulatory care nursing. *Nursing Economic\$*, 29(6), 331-334.
- Taylor, C.B., Youngblood, M.E., Catellier, D., Veith, R.C., Carney, R.M., Burg, M.M., ... ENRICHD Investigators. (2005). Effects of antidepressant medication on morbidity and mortality in depressed patients after myocardial infarction. *Archives of General Psychiatry*, 62(7), 792-798.
- The Joint Commission. (2015). Sentinel event alert: Preventing falls and fall-related injuries in health care facilities.

 Oakbrook Terrace: IL: Author. Retrieved from http://www.jointcommission.org/assets/1/18/SEA_55.pdf

- U.S. Department of Health and Human Services. (2011). 2011 report to Congress: National strategy for quality improvement in health care. Washington, DC: Author. Retrieved from http://www.ahrq.gov/workingforquality/nqs/nqs2011annlrpt.htm
- U.S. Preventive Services Task Force. (2007). Screening for high blood pressure: U.S. Preventive Services Task Force reaffirmation recommendation statement. *Annals of Internal Medicine*, *147*(11), 783-786.
- U.S. Preventive Services Task Force. (2009). Screening for depression in adults: U.S. Preventive Services Task Force recommendation statement. *Annals of Internal Medicine*, *151*(11), 784-792.
- Wasson, J. & Coleman, E.A. (2014). Health confidence: An essential measure for patient engagement and better practice. *Family Practice Management*, 21(5), 8-12.
- Wells, K.B., Stewart, A., Hays, R.D., Burnam, M.A., Rogers, W., Daniels, M., ... Ware, J. (1989). The functioning and well-being of depressed patients. Results from the Medical Outcomes Study. *JAMA*, *262*(7), 914-919.
- Whooley, M.A., Avins, A.L., Miranda, J., & Browner, W.S. (1997). Case-finding instruments for depression. Two questions are as good as many. *Journal of General Internal Medicine*, 12(7), 439-445.
- Wilson, M.L., & Gaido, L. (2004). Laboratory diagnosis of urinary tract infections in adult patients. *Clinical Infectious Diseases*, 38(8), 1150-1158.

B. Additional Readings

- Aiken, L.H., Clarke, S.P., & Sloane, D.M. (2000). Hospital restructuring: Does it adversely affect care and outcomes? *Journal of Nursing Administration*, 30(10), 457-465.
- Aiken, L.H., & Patrician, P.A. (2000). Measuring organizational traits of hospitals: The revised nursing work index. *Nursing Research*, 49(3), 146-153.
- Aiken, L.H., Sloane, D.M., & Sochlaski, J. (1998). Hospital organisation and outcomes. *Quality in Health Care*, 7(4), 222-226.
- Aiken, L.H., Sochlaski, J., & Lake, E.T. (1997). Studying outcomes of organizational change in health services. *Medical Care*, *35*(11 Suppl. 1), NS6-NS18.
- American Academy of Ambulatory Care Nursing (AAACN). (2010). Scope and standards of practice for professional ambulatory care nursing (8th ed). Pitman, NJ: AAACN.
- American Academy of Ambulatory Care Nursing (AAACN). (2010). Scope and standards of practice for professional telehealth nursing (5th ed). Pitman, NJ: AAACN.
- American Academy of Nursing. (2011). *Performance measures for care coordination: Strategic actions for nursing.*Preconference session presented at American Academy of Nursing 38th Annual Meeting and Conference, Washington, DC.
- American Association of Critical-Care Nurses (AACN). (2016). *AACN standards for establishing and sustaining healthy work environments: A journey to excellence, 2nd edition.* Aliso Viejo, CA: Author.
- American College of Physicians–American Society of Internal Medicine. (2000). *Telephone triage: White paper of the American College of Physicians–American Society of Internal Medicine*. Philadelphia, PA: Author.
- Anderson, R.A., Issel, L.M., & McDaniel, Jr., R.R. (2003). Nursing homes as complex adaptive systems: Relationship between management practice and resident outcomes. *Nursing Research*, *52*(1), 12-21.
- Anthony, A., & Milone-Nuzzo, P. (2005). Factors attracting and keeping nurses in home care. *Home Healthcare Nurse*, 23(6), 372-377.
- Belman, S., Chandramouli, V., Schmidtt, B.D., Poole, S.R., Hegarty, T., & Kempe, A. (2005). An assessment of pediatric after-hours telephone care: A 1-year experience. *Archives of Pediatrics & Adolescent Medicine*, 159(2), 145-149.
- Belman, S., Murphy, J., Steiner, J.F., & Kempe, A. (2002). Consistency of triage decisions by call center nurses. *Ambulatory Pediatrics*, 2(5), 396-400.
- Brewer, B.B., Greenberg, M.E., McEwen, M.M., Doyle, M.D., Lamb, G., Effken, J.A., & Verran, J.A. (2002). *The Systems Research Organizing Model (SROM)*. The University of Arizona, Tucson, unpublished manuscript.
- Centers for Medicare & Medicaid Services. (2003). *Health care industry market update: Home health.* Retrieved from https://www.cms.gov/Research-Statistics-Data-and-Systems/Statistics-Trends-and-Reports/CapMarketUp dates/Downloads/hcimu92203.pdf
- Centers for Medicare and Medicaid Services. (2012). Stage 2 eligible professional meaningful use core measures:

 Measure 17 of 17. Retrieved from http://cms.gov/Regulations-and-Guidance/Legislation/EHR
 IncentivePrograms/downloads/Stage2_EPCore_17_UseSecureElectronicMessaging.pdf
- Centers for Medicare and Medicaid Services (2013). Medicare electronic health record (EHR) Incentive pilot: Quick-Reference guide. *Physician Quality Reporting System (PQRS), V1*(1), 1-6.
- Chen, C., Garrido, T., Chock, D., Okawa, G., & Liang, L. (2009). The Kaiser Permanente Electronic Health Record: Transforming and streamlining modalities of care. *Health Affairs*, *28*(2), 323-333.

- Choi, J., Bakken, S., Larson, E., Du, Y., & Stone, P.W. (2004). Perceived nursing work environment of critical care nurses. *Nursing Research*, *53*(6), 370-378.
- Clark, S.P., Sloane, D.M., & Aiken, L.H. (2002). Effects of hospital staffing and organizational climate on needle stick injuries to nurses. *American Journal of Public Health*, *92*(7), 1115-1119.
- Conwell, L.J., & Cohen, J.W. (2005). Characteristics of person with high medical expenditures in the U.S. civilian non-institutionalized population, 2002. Statistical Brief #73. Rockville, MD: Agency for Healthcare Research and Quality. Retrieved from http://www.meps.ahrq.gov/mepsweb/data_files/publications/st73/stat73.pdf
- Craig, C., Eby, D., & Whittington, J. (2011). Care coordination model: Better care at lower cost for people with multiple health and social needs. IHI Innovation Series white paper. Cambridge, MA: Institute for Healthcare Improvement. Retrieved from http://www.ihi.org/resources/pages/ihiwhitepapers/ihicarecoordination modelwhitepaper.aspx
- Cullen, J. (1998). The needle and the damage done: Research, action research, and the organizational and social construction of health in the "information society." *Human Relations*, *51*(12), 1543-1564.
- Cushman, M., Ellenbecker, C., Wilson, D., McNally, M., & Williams, K. (2001). Home healthcare nurses: Why they leave and why they stay. *Caring Magazine*, *20*(10), 62-67.
- Dansky, K., Milliron, M., & Gamm, L. (1996). Understanding hospital referrals to home health agencies. *Hospital and Health Services Administrations*, *41*(3), 331-342.
- Delaney, C., & Huber, D. (1996). A nursing management minimum data set (NMMDS): A report of an invitational conference. Chicago, IL: American Organization of Nurse Executives.
- Dickoff, J., & James, P. (1968). Symposium on theory development in nursing A theory of theories: A position paper. *Nursing Research*, *17*(3), 197-203.
- Dolansky, M.A., & Moore, S.M. (2013). Quality and safety education for nurses (QSEN): The key is systems thinking. Online Journal of Issues in Nursing, 18(3), Manuscript 1. Retrieved from http://www.nursingworld.org/Quality-and-Safety-Education-for-Nurses.html
- Edelbauer, A., Vlasses, F., & Rogers, J. (2013). *The value of proposition in nursing*. Paper presented at the International Administration Research Conference, Baltimore, MD.
- Ellenbecker, C.H. (2001). Home health care nurses' job satisfaction: A system indicator. *Home Health Care Management & Practice*, *13*(6), 462-467. doi:10.1177/1084882230101300608
- Estabrooks, C.A., Tourangeau, A.E., Humphrey, C.K., Hesketh, K.L., Giovannetti, P., Thomson, D., ... Shamian, J. (2002). Measuring the hospital practice environment: A Canadian context. *Research in Nursing and Health*, 25(4), 256-268. doi:10.1002/nur.10043
- Fazzi, R., Agoglia, R., Mazza, G., & Glading-DiLorenzo, J. (2006). *The Briggs quality improvement/Hospitalization reduction study.* West Des Moines, IA: The Briggs Corporation. Retrieved from http://www.nahc.org/assets/1/14/hospredstudy.pdf
- Flores, D., Hickenlooper, G., & Saxton, R. (2013). An academic practice partnership: Helping new registered nurses to advance quality and patient safety. *Online Journal of Issues in Nursing, 18*(3), Manuscript 3. doi:10.3912/OJIN.Vol18No03Man03
- Flynn, L. (2005). The importance of work environment: Evidence-based strategies for enhancing nurse retention. Home Healthcare Nurse, 23(6), 366-371.
- Flynn, L., Carryer, J., & Budge, C. (2005). Organizational attributes valued by hospital, home care, and district nurses in the United States and New Zealand. *Journal of Nursing Scholarship*, 37(1), 67-72. doi:10.1111/j.1547-5069.2005.00005.x

- Flynn, L., & Deatrick, J.A. (2003). Home care nurses description of important agency attributes. *Journal of Nursing Scholarship*, 35(4), 385-390. doi:10.1111/j.1547-5069.2003.00385.x
- Frazier, S.C. (2003). Magnet home care agencies: A professional way to impact quality and retention. *Home Healthcare Nurse*, *21*(9), 603-610.
- Freeman, H.P. (2006). Patient navigation: A community-based strategy to reduce cancer disparities. *Journal of Urban Health*, 83(2), 139-141.
- Glor, B.A., & Barko, W.F. (1982). Sociotechnical systems using an industrial tested technology to design quality assurance standards in health care systems. *Military Medicine*, *147*(4), 313-317.
- Greenberg, M.E. (2000). Telephone nursing: Evidence of client and organizational benefits. *Nursing Economic*\$, 18(3), 117-123.
- Greenberg, M., Rutenberg, C., & Scheidt, K. (2014). Telehealth nursing practice. In S.A. Haas, B.A. Swan, & T.S. Haynes (Eds.), *Care coordination and transition management core curriculum* (1st ed.). Pitman, NJ: American Academy of Ambulatory Care Nursing.
- Greene, J., & Hibbard, J.H. (2012). Why does patient activation matter? An examination of relationships between patient activation and health-related outcomes. *Journal General Internal Medicine*, 27(5), 520-526. doi: 10.1007/s11606-011-1931-2.
- Greene, J., Hibbard, J.H., Sacks, R., & Overton, V. (2013). When seeing the same physician, highly activated patients have better care experiences than less activated patients. *Health Affairs*, 32(7),1299-1305. doi:10.1377/hlhaff.2012.1409
- Gruman, J., Rovner, M.H., French, M.E., Jeffress, D., Sofaer, S., Shaller, D., & Prager, D. (2010). From patient education to patient engagement: Implications for the field of patient education. *Patient Education and Counseling*, 78(3), 350-356. doi: 10.1016/j.pec.2010.02.002.Epub 2010 Mar 3.
- Haas, S.A., & Hackbarth, D.P. (1995). Dimensions of the staff nurse role in ambulatory care: Part IV--Developing nursing intensity measures, standards, clinical ladders and QI programs. *Nursing Economic\$*, 13(5), 285-294.
- Haas, S.A., Hackbarth, D.P., Kavanagh, J.A., & Vlasses, F, (1995). Dimensions of the staff nurse role in ambulatory care: Part II--Comparison of role dimensions in four ambulatory care settings. *Nursing Economic\$*, *13*(3), 152-165.
- Haas, S., & Swan, B.A. (2014). Care coordination models for achieving quality and safety outcomes for patients and families. In G. Lamb (Ed.), *Care coordination: The game changer How nursing is revolutionizing quality care.* Silver Spring, MD: American Nurses Association.
- Haas, S., Swan, B.A., & Haynes, T. (2013). Developing ambulatory care registered nurse competencies for care coordination and transition management. *Nursing Economic*\$, 31(1), 44-49.
- Hackbarth, D.P., Haas, S.A., Kavanagh, J.A., & Vlasses, F. (1995). Dimensions of the staff nurse role in ambulatory care: Part I--Methodology and analysis of data on current staff nurse practice. *Nursing Economic\$*, *13*(2), 89-97.
- Halpern, R., & Boulter, P. (2000). *Population-based health care: Definitions and applications*. Boston, MA: Tufts Managed Care Institute.
- Havens, D.S., & Aiken, L.H. (1999). Shaping systems to promote desired outcomes: The Magnet hospital model. *Journal of Nursing Administration*, 29(2), 14-20, 97.
- Head, B.J., Aquilino, M.L., Johnson, M., Reed, D., Maas, M., & Moorhead, S. (2004). Content validity and nursing sensitivity of community-level outcomes from the Nursing Outcomes Classification (NOC). *Journal of Nursing Scholarship*, 36(3), 251-259.
- Hirsh, D.A., Simon, H.K., Massey, R., Thornton, L., & Simon, J.E. (2007). The host hospital 24-hour underreferral rate: An automated measure of call-center safety. *Pediatrics*, *119*(6), 1139-1144.

- Judkins, S., & Rind, R. (2005). Hardiness, job satisfaction and stress among home health nurses. *Home Health Care Management and Practice*, 17(2), 113-118. doi:10.1177/1084822304270020
- Keepnews, D., Capitman, J.A., & Rosati, R.J. (2004). Measuring patient-level clinical outcomes of home health care. *Journal of Nursing Scholarship*, 36(1), 79-85.
- Kempe, A., Bunik, M., Ellis, J., Magid, D., Hegarty, T., Dickinson, L.M., & Steiner, J.F. (2006). How safe is triage by an after-hours telephone call center? *Pediatrics*, *118*(2), 457-463.
- Kempe, A., Dempsey, C., Hegarty, T., Frei, N., Chandramouli, V., & Poole, S.R. (2000). Reducing after-hours referrals by an after-hours call center with second-level physician triage. *Pediatrics*, *106*(1 Pt 2), 226-230.
- Kennedy, R., Murphy, J., & Roberts, D.W. (2013). An overview of the national quality strategy: Where do nurses fit?

 Online Journal of Issues in Nursing, 18(3), Manuscript 5. Retrieved from http://www.nursingworld.org/National-Quality-Strategy.html
- Kozlowski, S.W.J., & Klein, K.J. (2000). A multilevel theory and research in organizations: Contextual, temporal and emergent processes. In K.H. Klein & W.W.J. Kozlowski (Eds), *Multilevel theory, research and methods in organizations: Foundations, extensions and new directions* (1st edition) (pp.3-90). San Francisco, CA: Jossey-Bass.
- Kramer, M., & Hafner, L.P. (1989). Shared values: Impact on staff nurse job satisfaction and perceived productivity. *Nursing Research*, 38(3), 172-177.
- Kramer, M., & Schmalenberg, C. (2004). Development and evaluation of essentials of magnetism tool. *Journal of Nursing Administration*, 34(7/8), 365-378.
- Kramer, M., & Schmalenberg, C. (2005). Revising the essentials of magnetism tool: There is more to adequate staffing than numbers. *Journal of Nursing Administration*, *35*(4), 188-198.
- Kramer, M., Schmalenberg, C., Maguire, P., Brewer, B.B., Burke, R., Chmielewski, L., ... Waldo, M. (2009). Walk the talk: Promoting control of nursing practice and a patient-centered culture. *Critical Care Nursing*, 29(3), 77-93. doi:10.4037/ccn2009586
- Lake, E.T. (2002). Development of the practice environment scale of the Nursing Work index. *Research in Nursing and Health*, 25(3), 176-188.
- Lamb, G.S., Jennings, B.M., Mitchell, P.H., & Lang, N.M. (2004). Quality agenda: Priorities for action recommendations of the American Academy of Nursing Conference on Health Care Quality. *Nursing Outlook*, 52(1), 60-65.
- Lawrence, P.R., & Lorsch, J.W. (1967). Organization and environment. Boston, MA: Harvard Business School Press.
- Longworth, D.L. (2011). Accountable care organizations, the patient-centered medical home, and health care reform: What does it all mean? *Cleveland Clinic Journal of Medicine*, 78(9), 571-582. doi: 10.3949/ccjm.78gr.11003.
- Lynn, M.R., & Layman, E. (1996). The nature of nursing administration research. Knowledge building or fire stomping? *Journal of Nursing Administration*, 26(5), 9-14.
- Mark, B.A., Sayler, J., & Smith, C.S. (1996). A theoretical model for nursing systems outcomes research. *Nursing Administration Quarterly*, 20(4), 12-27.
- Mark, B.A., Sayler, J., & Wan, T.T. (2000). Market, hospital, and nursing unit characteristics as predictors of nursing unit skill mix: A contextual analysis. *Journal of Nursing Administration*, *30*(11), 552-560.
- Mark, B.A., Sayler, J., & Wan, T.T. (2003). Professional nursing practice: Impact on organizational and patient outcomes. *Journal of Nursing Administration*, 33(4), 224-234.

- McClure, M.L., & Hinshaw, A. (2002). *Magnet hospitals revisited: Attraction and retention of professional nurses*. Washington, DC: American Nurses Publishing.
- Miller, M. (2005). *Pay for performance in Medicare*. Statement before Senate Committee on Finance, 109th Congress, July 27, 2005.
- Mitchell, P., Heinrich, J., Moritz, P., & Hinshaw, A.S. (1997). Outcome measures and care delivery systems: Introduction and purposes of conference. *Medical Care*, *35*(11 Suppl 1), NS1-NS5.
- Murtaugh, C.M., McCall, N., Moore, S., & Meadow, A. (2003). Trends in Medicare home health care use: 1997-2000. Health Affairs, 22(5), 146-156.
- Navaie-Waliser, M., Lincoln, P., Karuturi, M., & Reisch, K. (2004). Increasing job satisfaction, quality care, and coordination in home health. *Journal of Nursing Administration*, *34*(2), 88-92.
- Nelson, J.M., & Cook, P.F. (2008). Evaluation of a career ladder program in an ambulatory care environment. *Nursing Economic*\$, 26(6), 353-360.
- Newman, M.A. (1992). Prevailing paradigms in nursing. Nursing Outlook, 40(1), 10-13, 32.
- Olin, G.L., & Rhodes, J.A. (2005). The five most costly medical conditions, 1997 and 2002: Estimates for the U.S. civilian non-institutionalized population. Statistical Brief #80. Rockville, MD: Agency for Healthcare Research and Quality. Retrieved from http://www.meps.ahrq.gov/mepsweb/data_files/publications/st80/stat80.pdf
- Pasmore, W.A. (1988). Designing effective organizations: The sociotechnical systems perspective. New York, NY: John Wiley & Sons.
- Porter, M.E. (2010). What is value in health care? *New England Journal of Medicine*, 363(26), 2477-2481. doi:10.1056/NEJMp1011024.
- Reid, R.J., Coleman, K., Johnson, E.A., Fishman, P.A., Hsu, C., Soman, M.P., ... Larson, E.B. (2010). The group health medical home at year two: Cost savings, higher patient satisfaction, and less burnout for providers. *Health Affairs*, 29(5), 835-843.
- Royal College of Nursing. (2007). *Understanding benchmarking: RCN guidance for nursing staff working with children and young people.* London, England: Author. Retrieved from https://www2.rcn.org.uk/_data/assets/pdf_file/0005/586985/003_144_web.pdf
- Sangl, J., Saliba, D., Gifford, D.R., & Hittle, D.F. (2005). Challenges in measuring nursing home and home health quality: Lessons from the first National healthcare quality report. *Medical Care*, *43*(3 Suppl 1), 124-132.
- Seifert, K., (2010). We're not falling for that! Preventing falls in the ambulatory setting. *AAACN ViewPoint*, 32(2) 1, 8-11.
- Sochalski, J. (2004). Building a home healthcare workforce to meet the quality imperative. *Journal for Healthcare Quality*, 26(3), 19-23.
- Stamm, W.E., & Norrby, S.R. (2001). Urinary tract infections: Disease panorama and challenges. *The Journal of Infectious Diseases*, 183(Suppl 1), S1–4.
- Swan, B.A. (2007). Transitioning from acute care to ambulatory care. Nursing Economic\$, 25(2), 130-134.
- Swan, B.A., Haas, S.A., & Chow. M. (2010). Ambulatory care registered nurse performance measurement. *Nursing Economic*\$, 28(5), 337-339, 342.
- Swan, B.A., & Moye, J. (2009). Growing ambulatory care nurse leaders: Building talent from the primed pipeline. *Nursing Economic*\$, 27(4), 251-254.
- Taschdjian, E. (Ed.). (1972). Perspectives on general system theory by Ludwig von Bertalanffy. A collection of essays gathered together and published two years after his death in 1972. Retrieved from www.isss.org/laszlofw.htm

- Tosif, S., Baker, A., Oakley, E., Donath, S., & Babi, F.E., (2012). Contamination rates of different urine collection methods for the diagnosis of urinary tract infections in young children: An observational cohort study. *Journal of Pediatrics and Child Health*, 48(8), 659-664. doi: 10.1111/j.1440-1754.2012.02448.x.xEpub 2012 Apr 27.
- Tucker, S.J., Ytterberg, K.L., Lenoch, L.M., Schmit, T.L., Mucha, D.I., Wooten, J.A., ... Wahlen, K.J. (2013). Reducing pediatric overweight: Nurse-delivered motivational interviewing in primary care. *Journal of Pediatric Nursing*, 28(6), 536-547.
- Tullai-McGuinness, S., Madigan, E.A., & Anthony, M.K. (2005). Exercise of autonomous home care practice: The relationship with nurse characteristics. *Home Healthcare Nurse*, *23*(6), 378-384.
- Verran, J.A. (1997). The value of theory driven (rather than problem driven) research. *Seminars for Nurse Managers*, 5(4), 169-172.
- Wagner, E.H. (1998). Chronic disease management: What will it take to improve care for chronic illness? *Effective Clinical Practice*, 1(1), 2-4.
- Weston, M., & Roberts, D. (2013). The influence of quality improvement efforts on patient outcomes and nursing work: A perspective from chief nursing officers at three large health systems. *Online Journal of Issues in Nursing,* 18(3), Manuscript 2. Retrieved from http://www.nursingworld.org/Quality-Improvement-on-Patient-Outcomes.html
- Wetta-Hall, R., Berg-Copas, G.M., & Dismuke, S.E. (2005). Help on the line: Telephone-triage use, outcomes, and satisfaction within an uninsured population. *Evaluation & the Health Professions, 28*(4), 414-427.
- White, J. (1995). Patterns of knowing: Review, critique, and update. Advanced Nursing Science, 17(4), 73-86.
- Zhou, Y.Y., Garrido, T., Chin, H.L., Weisenthal, A.M., & Liang, L.L. (2007). Patient access to an electronic health record with secure messaging: Impact on primary care utilization. *The American Journal of Managed Care,* 13(7), 418-424.
- Zuber, R. (2005). Home health quality measure: Inside the National Quality Forum Steering Committee. *Home Healthcare Nurse*, *23*(7), 462-268.

C. Acronym List

Organization/Phrase	Acronym
Accountable Care Organization	ACO
After Visit Summary	AVS
Agency for Healthcare Research and Quality	AHRQ
Ambulatory Surgery Center	ASC
American Academy of Ambulatory Care Nursing	AAACN
American Academy of Nursing	AAN
American Academy of Orthopaedic Surgeons	AAOS
American Hospital Association	AHA
American Institutes for Research	AIR
American Nurses Association	ANA
American Nurses Credentialing Center	ANCC
Appropriate Disposition	AD
Appropriate Referral	AR
Care Coordination and Transition Management	CCTM
Center for Advancing Health	CFAH
Centers for Disease Control and Prevention	CDC
Centers for Medicare and Medicaid Services	CMS
Clinician and Group Consumer Assessment of Healthcare	CG-CAHPS
Providers and Systems	
Collaborative Alliance for Nursing Outcomes	CALNOC
Electronic Medical Record	EMR
Emergency Department	ED
Emergency Medical System	EMS
Emergency Severity Index	ESI
Extended Release	ER
Food and Drug Administration	FDA
Full Time Equivalent	FTE
Group Practice Reporting Option, Care Measure	GPRO CARE
Group Practice Reporting Option, Preventative Measure	GPRO PREV
Health Research and Educational Trust	HRET
Hospital Consumer Assessment of Healthcare Providers and	HCAHPS
Systems	
Institute of Medicine	IOM
National Association of Children's Hospitals and Related	NACHRI
Institutions	
National Association of Community Health Centers	NACHC
National Committee for Quality Assurance	NCQA
National Database of Nursing Quality Indicators	NDNQI
National High Blood Pressure Education Program	NHBPEP
National Priorities Partnership	NPP
National Quality Forum	NQF
National Quality Strategy	NQS
Nurse-Sensitive Indicator	NSI
Nursing Alliance for Quality Care	NAQC
Office of Disease Prevention and Health Promotion	ODPHP
Patient Activation Measure	PAM
Patient-Centered Outcomes Research Institute	PCORI

Organization/Phrase	Acronym
Patient Health Questionnaire #2	PHQ-2
Patient Health Questionnaire #9	PHQ-9
Patient Health Questionnaire #9 Adapted for Adolescents	PHQ-A
Patient Protection and Affordable Care Act	PPACA
Patient Satisfaction Questionnaire Short Form, 18 items	PSQ-18
Physician Quality Reporting System	PQRS
Primary Care Provider	PCP
Quality Assurance	QA
The Joint Commission	TJC
United States Preventative Services Task Force	USPSTF
Urgent Care	UC